University of South Carolina Scholar Commons

Theses and Dissertations

2016

The Structural Choice Of A Senior Marketing Executive In The Top Management Teams In MNCs In USA, UK and Germany: Antecedents And Impact On Firm Performance

Kurt John Gleichauf University of South Carolina

Follow this and additional works at: https://scholarcommons.sc.edu/etd Part of the <u>Business Administration, Management, and Operations Commons</u>

Recommended Citation

Gleichauf, K. J.(2016). The Structural Choice Of A Senior Marketing Executive In The Top Management Teams In MNCs In USA, UK and Germany: Antecedents And Impact On Firm Performance. (Doctoral dissertation). Retrieved from https://scholarcommons.sc.edu/etd/ 3823

This Open Access Dissertation is brought to you by Scholar Commons. It has been accepted for inclusion in Theses and Dissertations by an authorized administrator of Scholar Commons. For more information, please contact dillarda@mailbox.sc.edu.



THE STRUCTURAL CHOICE OF A SENIOR MARKETING EXECUTIVE IN THE TOP MANAGEMENT TEAMS IN MNCS IN USA, UK AND GERMANY: ANTECEDENTS AND IMPACT ON FIRM PERFORMANCE

by

Kurt John Gleichauf

Bachelor of Science, Colorado State University, 1989

Master of International Business University of South Carolina, 1991

Submitted in Partial Fulfillment of the Requirements

For the Degree of Doctor of Philosophy in

Business Administration

Darla Moore School of Business

University of South Carolina

2016

Accepted by:

Omrane Guedhami, Co-Chair, Examining Committee

Martin S. Roth, Co-Chair, Examining Committee

Satish Jayachandran, Committee Member

Sali Li, Committee Member

Cheryl L. Addy, Vice Provost and Dean of the Graduate School



© Copyright by Kurt John Gleichauf, 2016

All Rights Reserved.



ABSTRACT

The role of the Chief Marketing Officer (senior marketing executive) has received minimal attention in the literature. Only recently has academic research investigated the position of the senior marketing executive. This dissertation uses neo-institutional and contingency lenses to extend prior theory and add cross-national perspectives on marketing management structural choices. An analysis of secondary data sources is used to clarify the key antecedents involved in the organizational choice of a senior marketing executive as a structural response in both one-tier and two-tier board governance systems. Possible mechanisms for the hypothesized effects are presented. Further, gaps in the prior literature on the economic effects of the senior marketing executive in the multinational corporation (MNC) are addressed. The moderating effect of each antecedent on firm financial performance is tested. Possible mechanisms for their influence are explained using contingency and institutional theories.



TABLE OF CONTENTS

ABSTRACT	iii
LIST OF TABLES	V
LIST OF FIGURES	viii
CHAPTER ONE: INTRODUCTION	1
CHAPTER TWO: CONCEPTUAL FRAMEWORK AND HYPOTHESES	5
CHAPTER THREE: RESEARCH DESIGN AND METHODOLOGY	58
CHAPTER FOUR: ANALYSES AND RESULTS	76
CHAPTER FIVE: DISCUSSION	105
REFERENCES	122
APPENDIX A: DESCRIPTION OF DATA	142
APPENDIX B: T-TEST	143
APPENDIX C: MISSING DATA ANALYSIS	146
APPENDIX D: NATH & MAHAJAN DATA AND REPLICATE DATA SETS	150
APPENDIX E: COUNTRY LEVEL COMPARISONS	164



LIST OF TABLES

Table 3.1. Variables, measures, values and sources of data for antecedent model
Table 3.2. Variables, measures, values and data sources for firm performance model74
Table 4.1. Means, standard deviations and correlations 81
Table 4.1. (continued) Means, standard deviations and correlations
Table 4.2. GEE regression of antecedents to the presence of a SME in TMT 88
Table 4.3. Support for hypotheses using GEE regression (2000-2010)
Table 4.4. Logistic regression of antecedents to the presence of SME (2000-2010)92
Table 4.5. Support for hypotheses using GLS regression (2000-2010)
Table 4.6. Country comparison of antecedent model hypotheses from 2008-2010
Table 4.7. GLS random effects analysis of SME impact on firm performance in the USA (2000-2010)
Table 4.8. Summary of findings of SME impact on market based firm performancemetrics using the GLS random effects method
Table 4.9. Summary of findings of SME impact on accounting based firm performancemetrics using the GLS random effects method
Table A.1. Distribution of country subsamples 142
Table B.1. T-test of firm performance measures 144
Table B.2. Industry segment distribution of T-test data
Table C.1 Missing Data per Variable
Table C.2 Missing data correlations 149



Table D.1 Comparison of Nath & Mahajan (2008) data and replicate data across industry segments
Table D.2. Descriptive statistics and Pearson correlation coefficients for replicate data set 152
Table D.3. Descriptive statistics and Pearson correlation coefficients for Nath andMahajan (2008) data set table153
Table D.4. Explanation of the Nath and Mahajan (2008) variables used
Table D.5. Hypotheses from Nath and Mahajan (2008) with predicted and actual results compared to the actual results of the replicated data set
Table D.6. Logistic regression comparison with CMO as the dependent variable using replicated data from 2000-2004
Table D.7. Logistic regression comparison with CMO as the dependent variable using replicated data from 2000-2007
Table D.8. Logistic regression comparison with CMO as the dependent variable from 2008-2010
Table D.9. Logistic regression comparison with CMO as the dependent variable from 2000-2010
Table D.10. Summary of hypothesized direct and interaction effects of the presence of aCMO on firm performance and actual results
Table D.11. GLS regression results with firm performance as the dependent variable 2000-2004
Table E.1. GLS random effects analysis of SME impact on firm performance for the USA from 2000-2010
Table E.2 GLS random effects analysis of SME impact on firm performance for Germany from 2000-2010
Table E.3 GLS random effects analysis of SME impact on firm performance for the UKfrom 2000-2010
Table E.4. GEE random effects analysis of SME impact on firm performance for theUSA from 2000-2010
Table E.5. GEE random effects analysis of the SME impact on firm performance for Germany from 2000-2010 169



Table E.6. GEE random effects analysis of the SME impact on firm performance for the UK from 2000-2010
Table E.7. Summary of findings of SME impact on accounting based firm performance metrics using GEE 171
Table E.8. Summary of findings of SME impact on accounting based firm performance metrics using GEE 172
Table E.9. GLS random effects analysis of USA SME impact on firm performance 2008-2010
Table E.10. GEE random effects analysis of USA SME impact on firm performance 2008-2010



LIST OF FIGURES

Figure 2.1. Antecedents to SME presence in the TMT12
Figure 2.2. Institutional factors as antecedents to the presence of a SME in the TMT18
Figure 2.3. Structural factors as antecedents to the presence of the SME in the TMT24
Figure 2.4. Strategic factors as antecedents to the presence of the SME in the TMT31
Figure 2.5. The model of the direct relationship between the SME and firm performance with moderating institutional, structural and strategic factors
Figure 2.6. The direct effect of the presence of a SME on firm performance40
Figure 2.7. The moderating effects of institutional factors on the SME's impact on firm performance
Figure 2.8. The moderating effects of structural factors on the SME's impact on firm performance
Figure 2.9. The moderating effect of strategic factors on the SME's impact on firm performance
Figure 4.1. The percentage of firms in a given year and country that have a senior marketing executive as a member of their top management team
Figure D.1. The rate of the presence of a senior marketing executive in the top

Figure D.1. The rate of the presence of a senior marketing executive in the top management team (Nath and Mahajan use the CMO identification) comparison between the replicate data set and Nath and Mahajan's (2008) published results......154



CHAPTER ONE INTRODUCTION

Marketing academicians have voiced their concern over marketing's eroding presence in the top level of firm management since the 1980's. The resulting decrease in marketing's influence on the corporate strategy and planning process has caused a considerable amount of concern (Day, 1992; Varadarajan, 1992). It is, after all, the top management team (TMT) of a firm that is primarily responsible for establishing, planning, and overseeing the execution of the organization's strategy. This includes decisions impacting resource allocations, organizational structure, market presence, technology development and acquisitions (Cyert & March, 1963; Hambrick and Cannella, 2004; Zorn, 2004).

The presence of a Chief Marketing Officer (senior marketing executive) in the TMT has been identified by prior researchers as a strong indicator of the influence of marketing as a separate function in the strategic planning process, the status of marketing within the organization, and the level of acceptance throughout the organization of the marketing concept (Piercy, 1986; Webster, 1981). Further, the commitment to the marketing function is critical to the process of developing market orientation, servicing customer relationships, creating the right products, and driving the profitability of the firm (McGovern et al., 2004; Piercy, 1986).



The participation of a senior marketing executive in the top levels of corporate management can be traced back to the 1950's and early 1960's. During this period the position of a chief marketing officer emerged as companies began to move away from production and sales driven models to a marketing focus with centralized marketing staff and an orientation to strategic market planning and development (Keith, 1960). The rise of marketing was so rapid that Hopkins and Bailey (1971) estimated that over half of the largest manufacturing companies in the United States had a senior marketing executive (SME) by the 1970's. However, after reaching a peak in the 1970's, there was a steady decline in corporate level marketing functions and their influence in the corporate planning and strategy formulation process. Much of the blame for the slow devolution of marketing's influence during the 1980's and 1990's has been placed on several trends: 1) the emphasis on aggressive acquisition and leveraging; 2) the stage of internationalization of many large corporations which emphasized local markets and subsidiary autonomy which dispersed marketing responsibilities; 3) the profound changes in accounting and financial reporting rules; 4) the increasing emphasis placed on the equity markets and the rise of the investor stakeholder; and 5) the difficulty of establishing a link between marketing activities and financial accountability (Hopkins and Bailey, 1984; Kerin, Mahajan and Varadarajan, 1990; Kumar and Shah, 2009).

Nath and Maharajan (2008) reported that from 2000 through 2004 approximately 40% of the companies listed in the S&P 500 included a marketing executive as a member of the TMT¹. The level of representation of such an important functional area such as

¹ Nath and Mahajan (2008) identify this senior marketing executive as the Chief Marketing Officer (CMO), although the person may, in reality, not hold this title. The term SME (for senior marketing officer) is therefore preferred and used in this research.



marketing stands in stark contrast to the representation of the finance functional area in the top executive ranks. Zorn (2004) stated that approximately 97% of firms studied had a Chief Financial Officer. This decline in the presence of SMEs is an indication that not only is the importance of marketing in strategic management being called into question, but so is marketing's importance to the financial performance of the firm (Kumar and Shah, 2009). Recent scholarship on the antecedents to the presence of a senior marketing executive and the impact of the presence of a senior marketing executive on the firm's financial performance delivered inconsistent and mixed findings (Nath, 2006; Nath and Maharajan, 2008).

This research seeks to build on prior research on the antecedents to the presence of a senior marketing executive, and the SME's impact on the financial performance of the firm. It accomplishes this by addressing more comprehensively the internal and environmental contingencies that act as antecedents to the presence of a senior marketing executive and the impact of the SME on financial performance in greater breadth than has been previously done. The questions addressed in this research are the following: 1) what are the salient antecedents among institutional, structural and strategic factors that influence the presence of a senior marketing executive; 2) how does the presence of a senior marketing executive impact the financial performance of an organization; and 3) do the antecedents and firm performance effects vary across countries? Addressing these questions will add to the understanding of the role of marketing in responding to environmental contingencies and its impact on performance in multinational organizations. This will expand the scope of both theoretical and practical understanding of the marketing function at senior management levels.



PLAN OF THE DISSERTATION

Chapter Two introduces the senior marketing executive oriented literature as well as the relevant TMT and international management literatures. Using the extant research from these literature streams, a conceptual model and hypotheses are proposed. Chapter Three describes the methods used to conduct the study, including the data samples used, descriptions of the variables, and descriptions of the analytic methods employed. Chapter Four presents the findings of the analyses and hypotheses testing. Chapter Five presents a discussion of the study's results, conclusions, limitations, and suggestions for future research.



CHAPTER TWO

CONCEPTUAL FRAMEWORK AND HYPOTHESES

The objective of this dissertation is to extend previous research on the antecedents leading to the organizational choice of a senior marketing executive in the top management team (TMT) and to assess the impact of the presence of a senior marketing executive (SME) on the financial performance of the firm. In order to accomplish this, the organizational theory literature, as it applies to senior marketing executive and TMT literature, is drawn upon to propose a conceptual framework for two separate models. For the purposes of this research, the TMT is defined as the group of senior management executives identified by the company in their annual report or proxy material. The senior marketing executive is defined as a marketing executive who the company identifies as being a member of the TMT.

The first model and set of hypotheses address key antecedents leading to the presence of a senior marketing executive. The second model and set of hypotheses address the impact of the senior marketing executive on the economic performance of the firm. Both models are viewed through the lenses of institutional theory and contingency theory. Previous research has only addressed the contingency perspective in relation to explaining the presence of the SME in the TMT.



Model I: Antecedents to the Presence of a Senior Marketing Executive in the TMT

There is a long history of the application of contingency theory to address questions surrounding the choice of alternative organizational structures. The position of senior marketing executive is not a typical structural choice for an organization. It is a departure from the normal executive structure. And as such, it would be instructive to better understand under what specific contextual antecedents this structure is chosen. Through the lens of contingency theory, the senior marketing executive position can be seen as an alternative structural choice, a choice that is driven by management's rational assessment of an organization's context, and a choice that has been deemed to be the most instrumentally appropriate structural response for the specific environment being faced by the organization.

Contingency Theory

Organizational theory gave rise to the original structural contingency frameworks beginning in the 1960's. Initially, it was an explanation of general managerial responses to contingencies being faced by the organization. Chandler (1962) proposed a relationship between strategy choice and organizational structure, but Lawrence and Lorsch (1967) promoted the idea that the external environment was fundamental to the structural choices made by organizations. Firms were seen as open systems which react and assemble themselves as a response to the demands made upon them by their external environments. The assumption that the external environment is a very powerful contextual variable influencing firm structure has remained at the core of the contingency approach (Burns and Stalker, 1961; Lawrence and Lorsch, 1967).



A number of researchers have attempted to categorize the myriad influences which act upon organizations. Although the majority of the influences explicated in prior research emphasize contingencies external to the firm, contingency theory seeks a balance among external factors, such as changes in technology and market turbulence, and internal factors, such as strategic choice and size (Burns and Stalker, 1961; Lawrence and Lorsch, 1967; Jelinek, 1977).

As a mid-range theoretical approach, contingency theory focuses on how different structures, strategies and behavior processes perform in different settings. Although the open systems approach (Cyert and March, 1963) assumes organizational adaptation and equifinality, all organizations are subject to the assumption of rationality in response to contingency (external and internal) contexts. This means that firms will express various responses to their environments and these variations are not random. They are based on a matching between contingency factors and internal structural responses. The ability to identify the important contingency variables allows the firm to choose the most appropriate structural response (Zeithaml, Varadarajan, and Zeithaml, 1988).

The use of the contingency approach in marketing is less developed than its application in general management. However, it has been applied in marketing behavior studies to assess the impact of contextual effects in sales (Weitz, 1981), project tasks (Sujan, Weitz and Sujan, 1988) and personnel performance (Ramaswami, 1996). The contingency approach has also been used to assess strategic marketing processes such as planning (Hambrick, 1983; Piercy, 1981) and strategy formulation (Day, 1986; Wind & Mahajan, 1981). Nath and Mahajan (2008) addressed both the antecedents to the presence of a senior marketing executive and the impact of the senior marketing executive on firm



performance using a contingency framework. Their research indicated a stable positive association between the presence of a senior marketing executive and innovation, corporate branding, product differentiation and the recent installment of a CEO from outside the firm. However, their results were unstable or not in the expected direction for a TMT with marketing experience, TMT with general management experience, diversification, and market concentration. Classic contingency theory expectations were irregularly supported by Nath and Mahajan's (2008) research results. This may be because their sample was limited to a five-year time period, but contingency theory does not offer guidance on time frames for establishing an organizational response to a set of contingencies. This research seeks to establish stable associations between contingency variables derived at the institutional, structural, and strategic levels and organizational structural choice by using a longer time frame and a larger sample that includes both domestic (United States) and foreign multinationals (Zeithaml, Varadarajan, and Zeithaml, 1988; Donaldson, 2001).

In this study, each level of contextual contingency (institutional, structural, and strategic) has multiple variables associated with it. These variables assess the uncertainty and dependency of that contextual level. The contingency theory literature postulates that there is a relationship between the internal and external contingencies of an organization and organization's structure. This is a central tenet. Further, the proposition is made that changes in contingencies can result in changes in structure (it is a dynamic system), and these causal relationships can be either linear or curvilinear in nature. And finally, the contingency view proposes that an organization attempts to optimize the fit between its structure and the contextual contingencies that it is facing. A misfit between the



www.manaraa.com

organization's structure and its context is suboptimal and leads to lower organizational performance (Donaldson, 2001).

This research is not designed to test fit, but the assumption is that rational actors will attempt to adapt organizational structure in the direction of improved fit with environment contingency that then leads to increased organizational effectiveness (Donaldson, 2001; Ruekert, Walker, and Roering, 1985). The rational processes behind structural adaptations are, in fact, not optimizing, but rather lead to satisficing adaptations. Managers are boundedly rational (Simon, 1979), struggling with incomplete information. Uncertain environments are characterized by rapid change, high complexity, or limited information. They present a risk to boundedly rational actors and force organizations to find structural adaptations to address the risk they pose to the organization. Uncertain environments can be external or internal, both will initiate a structural response (Donaldson, 2001).

Each of the contextual contingencies presented (institutional, structural, and strategic) represent a level of risk to the organization because of the information complexity and uncertainty they represent. Classic structural contingency theory posits that these contingencies will be addressed through a structural adaptation to mitigate the informational complexity and uncertainty. Because these contingencies impact the organization wide coordination of marketing processes, the expected structural adaption would enhance the organizational capability to deal with information complexity and uncertainty in the marketing domain. The structural adaptation placing of a senior marketing executive in the top management team of an organization is one way in which a firm can help senior executives manage the complexities of marketing activities such as



www.manaraa.com

interpreting market data, competitor and product assessments, consumer development and relationships, particularly when the external market and environment is turbulent and fast changing (Hopkins and Bailey, 1984; McGovern and Quelch, 2004; Piercy, 1986). This structural adaptation toward fit is expected to positively address the contingency challenges facing the firm and enhance firm performance.

Institutional Theory

However, firms are not fully rational in their pursuit of optimal firm performance. They also satisfice or ceremonially engage in activities that offer perceived benefits to the organization. They, as do most social constructions, follow a meandering path that can, and does, deviate from optimal performance. The institutional perspective represents a relaxation of some of the optimization assumptions of instrumental action used by contingency theory to explain structural choice and organization change. The institutional perspective emphasizes the importance of the contextual environment in shaping the organizational structure. However, unlike the contingency perspective, institutional theory explores the complex relationships between organizations and their context with a relaxed economic efficiency requirement (Donaldson, 2001; Scott, 2001). This would allow for a more institutional argument as an explanation for senior marketing executive structures which might help to explain some of the difficulty in identifying structure to performance relationships.

The cultural, social and political environments in which firms are embedded produce the contextual contingencies to which firms respond. It is the societal level at



which the common social and normative understandings are created which guide, or pressure, organizations in their chosen forms and structures. These social and normative understandings inform firms as to the accepted means-ends structures to be used to achieve goals. This form of isomorphic pressure for specific organizational structures extends to management professionals and their functional position, responsibility and activities within the firm (Meyer and Rowan, 1977; Scott, 2001). The mechanism of mimetic pressure is also used in institutional theory to partially explain convergence toward a specific structural form. This type of isomorphic pressure is invoked here as a mechanism by which organizations that are facing uncertain or complex environments identify effective organizational structural forms. When successful organizations act as structure templates, this research proposes that isomorphic pressures based on mimetic mechanisms can represent an instrumental contingency fit process attempting to achieve performance improvements. However, in order for an organizational structural form to act as a source of isomorphic pressure, a mimetic template for the pursuit of fit, the structure must have social legitimacy before it will lead to convergence (DiMaggio and Powell, 1983).

Figure 2.1 presents the proposed antecedent model. It introduces the three levels of factors, institutional, structural and strategic, and the attendant constructs which are proposed to be associated with the presence of a SME. The following sections of this chapter describe the hypothetical relationships between institutional, structural and strategic factors and SME presence.





Figure 2.1 Antecedents to SME presence in the TMT



Hypotheses

Institutional Factors:

Isomorphic Pressure

Institutional theory suggests that organizations are compelled to justify their structures and actions so that their behaviors conform to prevailing societal norms and expectations (DiMaggio and Powell, 1983; Scott, 1987). Organizations are receptive to signaling which they receive from their environments about what is considered acceptable or legitimate behaviors and norms. In order to gain or maintain social legitimacy, organizations engage in mimicry of other more legitimate organizations. This ensures their survival and prosperity by allowing them access to resources and protecting them from social sanction.

Institutional constituents such as competitors, governmental bodies and trade associations provide signals of legitimacy, as do consumers. These groupings and organizations can act as a source for benchmarking by other organizations, if it is felt that the benchmarked organization or group is considered successful or otherwise recognized for its superior performance or capabilities (DiMaggio & Powell, 1983). The resulting mimicry of institutions of which it is felt have more social legitimacy leads to isomorphism in both structure and action (Scott, 2001).

The suggested mechanism behind isomorphism is risk reduction and survival. By imitating the actions and structures of organizations that are seen to perform exceptionally, a firm is more likely to be successful, be seen as successful, or be doing what is required to become successful. Particularly within an organizational field,



organizations tend to converge toward the same structures and processes driven through mimicry rather than any contingent need associated with their environment (Haunschild, 1993; Suchman, 1995).

This research focuses on the financial information communicated by public firms regarding their financial health and competitiveness, which once in the public domain, act as signaling devices to other firms within their industry. The assumption is made in this research that the construct of the organizational field within institutional theory is analogous in application to an industry, or an industry segment. In the business world, financial success is a very compelling signal communicated within and beyond an industry (organizational field) or industry segment through the financial reporting requirements of publicly held corporations. The financial metric comparison of a firm's performance to industry/industry segment peers is a fundamental driver for analysts' investment ratings, stock performance, and top executive compensation awards.

Secondary financial data signaling is an indirect method to assess legitimacy pressure exerted on other organizations within an industry. However, because of its importance, as stated previously, the required assumption that peer firms in an industry segment are aware of the signals given by other members of the same segment is a reasonable one. Further, the strategy and structural choices of organizations which lead their industry have legitimacy by virtue of their economic success. Laggards will attempt to emulate, to the extent they can, the same strategic and structural choices of the successful firms in order to solve their competitive problems (DiMaggio and Powell, 1983; Fligstein, 1990). Therefore, this research assesses the isomorphic pressure (legitimacy) of the structural choice of a senior marketing executive within an industry



segment by identifying the leading firms within an industry segment. The prominent financial metric of "sales revenue" is used, since it is probably the most significant signaling metric (aside from "market share") available to, and recognized by, other relevant competitors within the industry segment and by external financial stakeholders.

It is assumed that the largest and most successful firms in an industry segment would be the most watched by their competitors in order to glean insights into their success. Haveman (1993) and Haveman and Rao (1997) have identified trait-based organizational imitation, whereby the organizational practices of subgroups with high status were imitated by the general population. Further support for this finding is supplied by studies done by both Haunschild and Miner (1997) and Greve (2000) in which it was shown that as successful firms became larger and more profitable, the awareness and sensitivity of their peer firms to the processes and structural choices these firms made increased. The isomorphic pressure of successful firms on their peers within an industry segment would cause the adoption by peer firms of the managerial structures and processes chosen by the segment leading firms, even when the adoption of their adoption is not a rational managerial choice for the organization.

The organizational choice of whether to have a senior marketing executive, or not, will be influenced by whether the largest and most successful firms in an industry segment also choose to have a senior marketing executive, or not (Scott, 1987). Thus:

Hypothesis 1: Firms operating in industry segments in which the leading firms have a SME in the TMT are more likely to have a SME in the TMT.



Societal Context

The country location of a firm's headquarters and its country of incorporation is an important definer of the societal (institutional) context in which an organization is embedded. Cultural and historical considerations at the country level help to define and shape the organizational logics that are legitimate in that environment (DiMaggio and Powell, 1983). There are two aspects of the societal context that are of interest to this research, 1) the general cultural orientation to Marketing and its status as a functional area of business and its strategic importance to the firm, and 2) the governance system, or choice of either unitary or dual board approach, as it is reflected in statute on the country level. Both of these aspects of business culture are considered because of the influence that they could exert over the presence of a senior marketing executive in the TMT.

Marketing Acceptance: Attitudes and beliefs about the status, roles and legitimacy of functional areas within organizations (e.g. marketing) are institutionalized at the societal (country) level. These generalized expectations exert normative pressure on organizations embedded in the respective societal context to conform.

Homburg, Workman, and Krohmer (1999) have found evidence that marketing departments suffer from lower levels of influence within German organizations, as compared to marketing departments in American organizations. They argue that marketing, as a functional area of business, does not have the same legitimacy, and therefore influence, in Germany as it does in the Anglo-American societal context for historical reasons. Many of the foundational theories and concepts of marketing were developed in the United States and England and diffused slowly to other countries much



www.manaraa.com

later, including Germany. As Homburg, Workman and Krohmer (1999) pointed out, "…, the first German-language marketing textbook was not published and the first marketing professorship at a German university was not established until the early 1970s."

It has also been argued that the lack of deregulation and continuing government regulatory control over advertising content and media access are institutional evidence of a negative social attitude toward the marketing profession (Homburg, Workman, and Krohmer, 1999). Other researchers have found evidence that German business culture is less supportive of market and customer orientation and instead represents a technology oriented business culture (Froeschl, 1997; Ulijn, Nagel and Tan, 2001), and have even anchored their claims for a negative orientation toward marketing in national culture frameworks (Brettel, Engelen, Heineman and Vadhansindhu, 2008). The more validity and importance marketing has in a societal context, the more likely marketing will be influential and take on a strategic role within the firm. Resources will be allocated to support marketing activities and organizational structures supporting marketing based decision making will be evident (Betektine and Haak, 2015; Engelen, Brettel, 2011; Hitt et al, 1982; Homburg, van der Wurff, Bakker and Picard, 2008; Workman and Krohmer, 1999). Therefore, it is expected that in the societal context of Germany, marketing will be seen as less valid and there will be less support toward marketing as a functional area. This should express itself in lower levels of marketing activity (lower allocation of resources to marketing related activities) and a lower incidence of senior marketing executives in the top management team (lack of willingness to support structural organizational changes associated with marketing) of firms headquartered in Germany, as



www.manaraa.com

compared to those based in the United States or the UK (Bitektin and Haack, 2015; Homburg, Workman and Krohmer, 1999).

Therefore, because of "cultural-professional" differences in the marketing legitimation, it is believed that the presence of an explicitly identified senior marketing executive in the top management team is more likely in Anglo-American business cultures. Thus:

Hypothesis 2: Firms headquartered and incorporated in a country in which there is greater social legitimation of the marketing profession and its activities are more likely to have a SME in the TMT.

Figure 2.2 shows the hypothesized relationship between SME isomorphic pressure and marketing legitimacy and the presence of a SME in the TMT.



Figure 2.2: Institutional factors as antecedents to the presence of a SME in the TMT



Structural Factors:

Industry Turbulence

Industry turbulence is defined for the purposes of this research as a lack of industry segment stability. Turbulence in an industry segment is indicated by relatively more competition (Drucker, 1986), relatively higher rates of dynamic change in the market environment (Eisenhardt and Bourgeois, 1988) as compared to other industry segments, a relatively higher segment growth rate, and a relatively higher level of technology within the segment (Homburg, Workman, and Krohmer, 1999).

Market concentration is posited to affect the environment of a firm primarily through market competition and power relationships with consumers. As market concentration increases within a segment, the power of the seller increases and that of the consumer decreases. As the seller's power increases, the seller's perception of environment risk associated with having to be sensitive and responsive to consumer needs, wants, and aspirations is reduced. The quality and number of the product offerings and innovations become less relevant to firm success and ultimately lead to an oligopolistic approach to a market (Porter, 1985).

An oligopolistic situation reduces the number of environmental events which occur within any given period of time and reduces the amount, and turnover, of market information (Eisenhardt and Bourgeois, 1988, Glazer and Weiss, 1993). Both the number of events in the environment and the amount of information turnover has direct implications for management.



A low level of market turbulence is indicated by a high level of market concentration. As organizations acquire market power and consolidate market share, a market will experience lower rates of change, which in turn supports the consolidation of the market and allows fewer opportunities for new competitors to enter the market. Under a market concentration scenario, the logic of efficiencies of scale and scope will cause marketing to pursue coherent and standardized approaches to the markets. The presence of a senior marketing executive lends itself to the development, application and execution of firm-wide marketing policies and strategies and is an indicator of an organization's emphasis on standardization and efficiency.

When an industry segment is in an expansion phase with rapid sales growth (higher information turnover and a greater number of market change events) there is a greater need for market responsiveness. There is a rush of competitors and constant pressure to establish products or standards within the segment. Marketing capabilities are highly valuable as firms attempt to not only stay ahead of the competition, but also to understand, create and manage connections with customers better than the competition (Moorman and Rust, 1999). The same informational demands occur as a market segment starts to erode and collapse. The capacity of marketing management to react to frequent time sensitive changes in the environment will necessitate a managerial structure with decision making autonomy closer to the market.

Technology based industry segments experience high rates of growth and technical change. Technological standards can, and are, quickly eclipsed. Marketing function activities must be very close to the market and highly integrated with product development internally. Industry segments with a high technology component require



www.manaraa.com

marketing inputs not only on a tactical level, but also on a strategic level (Eisenhardt and Bourgeois, 1988).

In a turbulent scenario the organizational emphasis will not be on standardization and efficiency, but on attempting to address the informational ambiguity caused by turbulent environments and supporting quick market oriented organization responses (Eisenhardt and Bourgeois, 1988). Thus:

Hypothesis 3: Firms in industry segments that are more turbulent are more likely to have a SME in the TMT.

Market Internationalization

Large multinational corporations may be present in a multitude of markets simultaneously with individual subsidiaries supporting their activities in each of these markets. Within the strategic management framework, it is assumed that a firm establishes a presence (subisidiary) in a foreign market in order to access the market or to access resources which are located in the market (Buckley and Casson, 1976; Hymer, 1966).

Foreign subsidiaries are embedded in environments which differ along many dimensions from those of the headquarters. Each subsidiary will have its own development history which leads to subsidiary differences in size, age, market position, and resource complements. The actual demands of the consumers in the various markets make it increasingly difficult to oversee and control processes as the organization



increases in the number of subsidiaries. As individual foreign markets develop and grow in real and relative importance to a firm's home market, the firm's top executive team needs to be able to synthesize the complexities of the information and knowledge of these markets in order to be successful (Ambos and Schlegelmilch, Bartlett and Ghoshal, 1989; Xu and Shenkar, 1994).

Because environmental differences create competing agendas within an organization and between subsidiaries and their headquarters, it is a strategic imperative for a MNC to be able to coordinate and control across all of its units, ensuring that there is a convergence towards a common goal (Bartlett and Ghoshal, 1989; March and Simon, 1993; Simon, 1979). For the top executive team, coordination and control requirements increase as the organization internationalizes and grows the number of subsidiaries within its structure. As the sales from international markets become more important to the financial success of the firm, the importance of the complexities of the marketing activities become more relevant to strategic planning processes. The organizational choice of which managerial structure is to be used for coordination and control becomes more critical (Birkinshaw and Moore, 1998; Malone and Crowston, 1994).

Bartlett and Ghoshal (1989) addressed many of the complexities and difficulties resulting from establishing subsidiaries in foreign markets. They built on earlier international strategy descriptions (Bartlett, 1986; Doz and Prahalad, 1987) based on a continuum of integration versus coordination and differentiation versus responsiveness. This approach led them to propose a typology of international strategies which an organization can pursue. These typologies have their corollary within the marketing function and are typified by their degree of centralized control or local market



responsiveness. Of these, two typologies are applicable to this research: 1) a global marketing strategy in which there is a high degree of integration (centralized control) and low responsiveness to local market requirements, and 2) a multi-domestic strategy in which there is a low degree of integration (decentralized control) and high responsiveness to local market requirements (Porter, 1986).

As both the number of international subsidiaries and their relative importance to the total revenue of the firm increase, the amount of environmental uncertainty and risk within the area of marketing responsibility increases. This research proposes that the organization will respond to increasing market uncertainty and risk with marketing management structures that are designed to reduce uncertainty by increasing the overall level of coordination and control within the marketing function.

The senior marketing executive, as the most senior marketing manager and member of the TMT, is responsible for ensuring that the organization's marketing strategy is accurately carried out. Whether a MNC chooses to pursue a global strategy, a multi-domestic strategy, or some variation on that continuum is not a deciding factor in whether or not an organization will choose to have a senior marketing executive. Rather, the degree of uncertainty in the coherent execution of an organization's chosen marketing strategy, as the number of subsidiaries increases, is the more likely deciding factor in whether or not an organization chooses to have a senior marketing executive. The increased level of coordination and control necessary to reduce the risk to the organization, as the number of subsidiary markets increase, will drive the choice of a marketing executive within the TMT. The presence of a senior marketing executive will



allow the TMT to more easily align corporate oversight with complex market scenarios. Thus:

Hypothesis 4: Firms with higher levels of internationalization will be more likely to have a SME in the TMT.

Figure 2.3 shows the predicted relationship between industry turbulence and market internationalization and the presence of a SME in the TMT.





Strategic Factors:

Branding Strategy

Chandler (1962) argued that organizational structure follows strategy. The assumption that a strategy choice precedes the organization's structure, in addition to its intuitive appeal, has received considerable empirical support (Child, 1972, 1997). Donaldson (2001) and Lawrence and Lorsch (1967) have also argued that both environmental and strategic contingencies influence the organizational structure.



There are a number of taxonomic approaches used in the literature to classify branding strategies. Some of the most prominent typologies are the four category system proposed by Murphy (1987, 1989), the three category system proposed by Olins (1989), and the three category system of Laforet and Saunders (1994).

Olin's (1989) approach has a number of weaknesses. He only uses corporate identities, and the identification of the corporate subsidiaries with their respective corporate entities and corporate brands. This is not the focus of this research and would therefore engender misspecification of the construct of interest. Murphy's (1987, 1989) four classifications include corporate, product brand, balanced, and mixed strategies. However, there is not enough measurement clarity between balanced and mixed strategies for the design of this research. This research will use the approach based on that of Laforet and Saunders (1994) which classifies marketing strategies into three groups: corporate branding strategy, house of brands strategies). The Laforet and Saunders (1994) typology avoids some of the misspecification and construct clarity issues of other approaches. Laforet and Saunder's (1994) typology descriptions are as follows:

Corporate branding: The corporate name is dominant in promoting all of the firm's products and services. There is a standardization of the corporate brand and an emphasis is placed on a global marketing strategy that is coherent across markets. This is true throughout the corporation, its markets and subsidiaries. Examples of firms which primarily use a corporate branding strategy include USAA and BMW.



House of Brands: The corporate name is not used in branding the firm's products or services. Instead, the products and services are marketed using individual brand names and the marketing of the individual brands may be limited to individual and regional markets or may be managed uniformly across markets. The individual brands are often managed as a portfolio of brands. Examples of a firm that primarily uses a house of brands strategy is Procter & Gamble and its brands Pampers, Crest, and Iams.

Mixed branding: The corporate name is used along with a portfolio of house or family brands. Non-corporate name brands are typically strong and significant in value. Examples include the 3M corporate brand name and 3M Post-it and 3M Scotchgard brand names.

A corporate branding strategy is predicated on communicating, promoting, and maintaining a single corporate brand. The single brand is the vehicle and embodiment of the organization's market identity and value. When this is the case it is advantageous to centralize the decision making processes which maintain, protect, and control the brand. The centralization of brand control also makes it easier to align the management of the brand with the strategic planning process. In the case of a corporate brand, brand management and the strategic planning process are inextricable from one another. Organizations that have chosen a corporate brand strategy will also choose a structural configuration that effectively coordinates marketing operations synergistically across all their markets (Aaker and Joachimsthaler, 2000; Keller, 2003; Rao, Agarwal, and Dahlhoff, 2004).



www.manaraa.com

In contrast, the house of brands strategy is a brand portfolio approach which may, on a brand level, deviate from corporate strategic goals. In such a situation, the product level brand development planning process and the corporate level strategic planning process do not have to be in complete alignment. The marketing management responsible for the individual brands are located at divisional or regional levels, not at the senior executive level. The organization's marketing focus is on specific market or brand development requirements.

Therefore, the presence of a senior marketing executive is expected to be more prevalent in an organization when brand related information inputs are important to the organization's strategic planning process, not when informational requirements market planning activities are oriented toward individual or regional markets. Thus:

Hypothesis 5: Firms which pursue a corporate branding strategy will be more likely to have a SME in the TMT than those pursuing other brand strategies.

Customer Type

The product market (business-to-business [B2B] or the business to consumer [B2C] marketplace) in which a firm competes is considered an important firm strategy contingeny. For the purposes of conceptualization, the choice of customer type is relevant in that it represents a firm level contingency that impacts marketing strategy and execution at all levels (Zeithaml, Varadarajan, and Zeithaml, 1988).


There are a number of dimensions on which B2B and B2C markets are differentiated: first, the level of customer concentration is higher in B2B segments, often reflecting a Pareto distribution, than in B2C segments; two, the choice of a supplier reflects a more rational and technical needs dimension in B2B segments than in the B2C segments; and third, there is a lower level of impulse or affect driven decision making involved in the B2B segments as compared to B2C segment purchase decisions (Segal, 1989).

In addition, there is a relational element to the B2B segment which requires higher levels of trust than is typical in a B2C transaction. This is due to the smaller number of transactions, the higher transaction value, and the potential for higher switching costs in B2B segments, as compared to B2C segments (Segal, 1989). Switching costs can lead to risk aversion and reluctance of consumers to switch suppliers, products, or services. The consequences of switching component suppliers for manufacturing processes and product characteristics can be significant. These potential switching costs tend to express themselves in B2B relationships that are closer and longer term than B2C relationships. The result is a lower overall transaction rate in B2B relationships, but a higher rate of transaction repetition (Segal, 1989).

The consistency of a firm's corporate image or reputation becomes more important in a B2C segment than in a B2B segment (Srinivasan, Lilien and Sridhar, 2011). Market specific product, stability, marketing message communication, and customer service consistency are more important in B2C segments. B2C sectors have much higher numbers of contact points with consumers and higher levels of innovation and introduction in the products and services being offered to consumers. The constant



change in products and contact points allows for a process of continual redemption of loyalty and innovation in presentation of brands (Dwyer and Tanner, 2008; Srinivasan, Lilien, and Sridhar, 2011).

The impact of individual product branding strategies is less important in the more technical and rational decision making B2B markets than in B2C markets. For example, bid or tendering processes in which the technical requirements of the product or service are explicitly stated and competition among multiple vendors is based primarily on price alone is common in B2B markets, but rare in B2C markets (Rangan, Moriarty, and Swartz, 1992). This reduces the importance of many marketing activities in the purchase process in B2B markets. In contrast, firms in the B2C markets rely heavily on affect centered mass communication based on large numbers of individual customers and invest heavily in brand development and product positioning that is projected through third party channels to the end consumer (Reed, Story, and Saker, 2004). It is expected that firms which are primarily, or solely, involved in B2C markets are likely to have senior marketing executives as members of their top management team. Thus:

Hypothesis 6: Firms whose business is primarily business to customer (B2C) will be more likely to have a SME in the TMT than firms whose business is not primarily B2C.

Product Type

In addition, this research makes the distinction between physical products and services in both B2B and B2C markets. Services, as compared to product goods, are less tangible, the production and consumption are temporally very compressed or take place



at the same time, are highly variable in their characteristics (less standardized), and are more perishable (Zeithaml, Parasumaran, and Berry, 1997). In addition, services have shorter life cycles and provide easier competitive entry. In short, services have a very different set of competitive concerns associated with them that create market uncertainty and dynamic fluctuation (Srinivasan, Lilien, and Sridhar, 2011). Dynamic and uncertain market conditions require a greater dedication of firm resources to assess market conditions, acquire and keep customers, and to identify new customer needs in order to stay ahead of the competition (Srinivasan, Lilien, and Sridhar, 2011). It is expected that firms which are primarily engaged in a service industry will be more likely to have a senior marketing executive in the top management team. Thus:

Hypothesis 7: Firms that primarily compete in a service goods market will be more likely to have a SME in the TMT than firms that compete in a product goods market.

Figure 2.4 shows the relationship between corporate brand strategy, business to customer strategy and service product strategy to the presence of a SME in the TMT.





Figure 2.4: Strategic factors as antecedents to the presence of the SME in the TMT Model II: The Impact of Choosing to have a Senior Marketing Executive on Firm Performance

Marketing Activity and Firm Performance

Traditionally, the influence of marketing activities on firm performance has been characterized as product-market oriented and directly related to sales performance. The four P's of product, pricing, placement, and promotion have described the tactical areas of business activity in which marketing traditionally holds sway. It is through these activity domains that marketing initiates, develops, and maintains the firm's relationship with customers (Kotler, 1984; Day, 1994). However, the corporate strategic level influence of the marketing domain on planning, communicating and delivering of the firm's value proposition to customers, managing the customer and stakeholder relationships is essential to financial and market based firm performance (Webster, 2005).

Firm revenue is driven by building and managing a positive and solid relationship with customers for the firm's products and services. Since the marketing domain is the primary business function responsible for creating and maintaining profitable customer



relationships, it directly impacts the revenue performance as well as the equity based valuations of the firm such as market capitalization (Kumar and Shah, 2009; Villanueva and Hanssens, 2007). Prior empirical research supports the notion that effective marketing activities are linked to sales growth, return on assets and measures of firm value (Boyd, Chandy and Cunha, 2010; Krasnikov and Jayachandaran, 2009). Additionally, positive relationships have been shown between brand strength and firm value (Mizik and Jacobson, 2008), advertisement and firm value (Joshi and Hanssens, 2009), and customer satisfaction and firm value (Anderson, Fornell, and Mazvancheryl, 2004).

A meta-analysis of the impact of marketing, R&D, and operations on firm financial outcomes done by Krasnikov and Jayachandran (2008) showed a consistent positive impact of marketing activities on revenue growth, market share, and profitability. These results support other research findings which indicate that a strong marketing capability can positively impact shareholder value (Srivastava, Shervani, and Fahey, 1998, 1999) and that effective, well executed marketing strategies support top line sales growth (Zuckerman and Hudson, 2007).

The Role of the Senior Marketing Executive

The presence of the senior marketing executive as a member of the TMT indicates a relatively greater marketing influence on decision making and planning in the development and implementation of strategic marketing activities of the firm. The role of a senior marketing executive in the TMT is to provide strategic leadership in all issues relating to the marketing domain. Boyd et al. (2010) identified three roles that a senior



marketing executive plays: first, an informational role in which market relevant information is communicated to the strategic management level of the firm; second, a decisional role in which marketing relevant issues are included in key strategic decisions of the firm; and third, a relational role in which marketing relevant relationships with external stakeholders are considered at the strategic management level of the firm. Because of the market relevant inputs that a senior marketing executive can bring to the strategic table, the presence of a senior marketing executive in a firm's TMT is expected to enable a firm to develop and implement marketing policies and strategies that are more market sensitive and effective in generating revenue growth and firm value, than if a senior marketing executive to consistently bring marketing resources to bear on strategic decision making through the TMT creates both dynamism and consistency in the creation and implementation of marketing strategy.

Although there is limited empirical research on senior marketing executive roles in the firm, it is possible to augment the present state of the literature with further research on 1) when the presence of senior marketing executives in the TMT are indicated by environmental contingencies, and 2) what impact a senior marketing executive in the TMT will have, given the environmental contingencies present.

The members of the TMT address management issues of importance to the firm that tend to be in domains that are characterized by complexity and uncertainty. Within the TMT context, the senior marketing executive is responsible for processing information from the marketing domain, communicating it to other members of the TMT, and actively supporting or making complex and uncertain decisions with respect to the



www.manaraa.com

marketing domain that have significant potential impacts on firm resources and performance. McGovern et al. (2004) state that these activities include market monitoring, consumer assessment, brand strategy development, advertising strategy development, market planning, market research, and inter unit marketing coordination in a senior marketing executive's responsibilities. Other researchers have included new product development, sales forecasting, market entry, staff selection, and corporate structure to a senior marketing executive's areas of influence (Hopkins and Bailey, 1984; Piercy, 1986; Varadarajan, Jayachandran, and White, 2001). Although the possible areas of senior marketing executive is integral to the process of developing, executing, and assessing the firm's product-market strategies, both near and long term.

This research looks at the effect of the presence of a senior marketing executive on two categories of firm performance metrics: accounting based (sales growth, return on assets and return on sales) and market based (Tobin's q, market to book and market to equity). This helps to directly relate the impact of the senior marketing executive to two distinct time horizons, one short (accounting based) and the other long (market based), or put another way, to a tactical impact and a strategic impact. The short term impact evaluation is directly tied to market-product performance and is defined as revenue based, the longer term impact is tied to the valuation of the firm through its stock performance which is defined as an equity (stock) based metric.

Previous research has shown very weak, conflicting, or no evidence of the presence of a senior marketing executive influencing firm financial performance (Boyd, Chandy, and Cunha, 2010; Nath and Mahajan, 2010). Weinzimmer et al (2003) found a



www.manaraa.com

small positive impact on sales performance, but Nath and Mahajan (2008) reported no effect of the presence of a SME on sales performance. In later studies, Nath and Mahajan (2010) did find a small positive effect of the presence of a SME on sales performance, but they ascribed the effect to the amount of influence which a SME wields within the firm's TMT.

Prior research findings on the impact of a SME on market based performance metrics has been equally uneven. Nath and Mahajan (2008) found that the presence of a senior marketing executive did not impact firm value as measured by Tobin's q. However, recently, German et al. (2015) presented evidence for a positive impact on Tobin's q, and possibly excess stock returns (using Jensen's alpha), when a SME is present in the TMT. Boyd, Chandy, and Cunha (2010) were also able to tease out some of the complex interactions between senior marketing executive power and its impact on firm equity value in their event research on SME appointments. They were able to show small effects of SME appointment announcements on stock movements. All in all, there is a very mixed picture of when and how the presence of a senior marketing executive impacts firm financial performance and equity valuation.

In this research it is proposed that the contingency factors which are predicted to influence the presence of a SME in the TMT also act to moderate the positive impact of the SME presence on firm performance. Figure 2.5 shows the expected main and contingency effects of the SME on firm performance.





Figure 2.5: The model of the direct relationship between the SME and firm performance with moderating institutional, structural and strategic factors

Hypotheses

Revenue Performance

Sales revenue growth and profit margins are the most common objectives mentioned by senior managers, and they are direct and highly visible outcomes of product-market effectiveness. Sales, and the cash flow that it represents, are crucial to the firm's health and competitiveness. The responsibility for generating sales and achieving margin targets that generate profit is primarily the responsibility of marketing management (Brush, Bromiley, and Hencrickx, 2000; Homburg, Workman, and Krohmer, 1999).



Hambrick, Cho, and Chen (1996) found evidence that functional domain heterogeneity in the TMT, and the specialized knowledge that it supports, improves the decision making quality and performance of the TMT. Having a senior marketing executive to act as a conduit for marketing information and knowledge into the highest executive ranks of the firm will positively impact the firm's sales and profitability performance by supplying knowledge and expertise linking the customer to various processes within the firm (Day, 1994). In assessing the marketing function within the firm, Moorman and Rust (1999) state that the principal responsibilities of the marketing domain in an organization are making sure that the customer is connected with 1) the product, 2) service delivery, and 3) financial accountability. Although the assessment of the customer's connection with product and service delivery are outside the scope of this research, this research does use the measure of sales growth as an assessment metric for the impact of marketing on firm performance (Finkelstein and Boyd, 1998).

Firm Profitability Performance

The profitability performance of the firm is defined as the return on sales (ROS), and return on assets (ROA). There are several reasons why ROA is included. First, pricing strategy directly affects profitability, and therefore firm performance, and second, marketing strategy drives overall pricing strategy decisions, which assumes a strategic consensus at the highest executive levels on an issue of marketing strategy that directly affects firm performance outcomes (Homburg, Jensen, and Hahn, 2012; Kotler, 1984; Moorman and Rust, 1999). For example, if a market penetration strategy is chosen as a



firm strategy, then even though sales revenue may grow, profitability may fall due to pricing discounts. Both ROA and ROS are well documented in the TMT literature as metrics for measuring the impact of the top executives on firm profitability and performance (Hambrick and Cannella, 2004; Zhang, 2004).

Firm Valuation Performance

A key role of marketing, and the senior marketing executive, is to establish a relationship with the customer and link customers to sales outcomes (Moorman and Rust, 1999). How well, or how poorly, a firm is able to do this is reflected in its revenue growth and profitability. The market's expectations of the future revenue and profit trend is encompassed in the firm's share price (Krasnikov and Jayachandran, 2008; Srinivasan and Hanssens, 2009).

Marketing capability is based on the knowledge of customer needs, the ability to respond to those needs, and the ability to forecast the future status and character of those needs. Much of the knowledge and capability that is embedded in the marketing domain is difficult to codify. It is socially complex, constantly changing and tacit in its nature (Simonin, 1999). Therefore, the direct participation of marketing, in the form of the presence of a senior marketing executive in the TMT, is critical to the successful development and implementation of marketing strategy. Marketing resources and capability have been shown to positively impact shareholder value (Srivastava, Shervani, and Fahey, 1998, 1999). Although prior research outcomes have been mixed, it is expected that consistent participation of a senior marketing executive in the TMT will



www.manaraa.com

result in firm performance signals which will positively influence share valuation of the firm.

The presence of a senior marketing executive in the TMT should improve the cross-functional integration of knowledge, communication, consensus building, and commitment to marketing issues within senior executive ranks. These activities are seen as necessary for the formulation, coordination, and execution of optimal marketing strategy (Menon, Bharadwaj, Adidam, and Edison, 1999). The improvement in the execution of these important internal processes should lead to revenue growth, profitability, and superior stock performance of a firm.

In general, it is expected that the presence of a SME in the TMT will positively impact both financial and market metrics of firm performance. Thus,

Hypothesis 8a: The presence of a SME in the TMT will positively impact revenue/profitability performance metrics of the firm.

Hypothesis 8b: The presence of a SME in the TMT will positively impact market value performance metrics of the firm.

Figure 2.6 shows the proposed positive relationship between the presence of a SME on the TMT and firm performance.





Figure 2.6: The main effect of the presence of a SME on firm performance

Institutional Factors:

Isomorphic Pressure

Institutional and neo-institutional perspectives on organizational structure soften the rational or functional requirements of the chosen structures. Instead, they emphasize the isomorphic pressures exerted on organizational action and structure which are accepted, legitimated, or simply taken for granted within an organizational field, without a strict requirement of rational utility (DiMaggio and Powell, 1983; Scott, 1987). Within these "fields" organizations choose their structure and behavior, in part, to conform to norms that they believe will increase their resources and legitimacy in the eyes of important stakeholders. Firms may also practice mimetic isomorphism as a short-cut to dealing with highly uncertain or turbulent environments.

The largest and most important firms within an organizational field will exert considerable isomorphic pressure on the other firms within the field. This pressure may be in favor of having a senior marketing executive, or not. It is expected that if the weight of isomorphic pressure is in favor of the presence of a senior marketing executive, then there is a rational advantage to having a senior marketing executive that will express



itself in higher short and long term performance in sales, profitability and market value. Thus:

Hypothesis 9a: The effect of the presence of a SME on revenue/profitability performance metrics of a firm will be positively moderated the greater the level of isomorphic pressure for a SME in the TMT within the industry segment.

Hypothesis 9b: The effect of the presence of a SME on the market value performance metrics of a firm will be positively moderated the greater the level of isomorphic pressure for a SME in the TMT within the industry segment.

Societal Context:

This research has proposed that two aspects of the societal context, 1) the general acceptance, or cultural orientation, toward marketing and its status as a functional business area, and 2) the type of governance structure used, affect the prevalence of having a senior marketing executive in the top management team. This research also proposes that these same aspects, when coupled with the presence of a senior marketing executive in the top management team.

Marketing Acceptance

The level of societal acceptance of marketing and marketing activities can also be interpreted as the importance of marketing within a society. Status and influence of marketing within society and within business organizations are a result of the societal level understandings surrounding marketing and marketing activities. The level and



direction of both have been debated for some time (Day, 1992; Verhoef et al, 2011). There is evidence that societal levels of the acceptance of marketing activities and the perception of it as a professional field varies across countries, including the USA, UK, and Germany (Homburg, Workman, and Krohmer, 1999; Verhoef et al, 2011).

The link between the acceptance of marketing and marketing activities on a firm level and firm performance is not consistent. Recent scholarship has indicated a positive relationship between marketing capabilities and firm performance (Krasnikov and Jayachandaran, 2008). This research proposes that societal level marketing acceptance will support the presence of marketing capabilities (senior marketing representation in the top management team), which will be further leveraged by the level of societal marketing acceptance to positively influence market and revenue based firm performance outcomes (Krasnikov and Jahachandaran, 2008; Verhoef et al, 2011). Thus:

Hypothesis 10a: The effect of the presence of a SME on the revenue/profitability performance metrics of a firm will be positively moderated the greater the level of societal marketing acceptance.

Hypothesis 10b: The effect of the presence of a SME on the market value performance metrics a firm will be positively moderated the greater the level of societal marketing acceptance.

Governance System

The regulatory environment in which a firm operates is part of its societal context. The dimension of the societal context that is of interest is the set of governance regulations



which apply to public stock corporations. These regulations are generally established at the country level and are in effect for all public stock corporations incorporated in the country, although most of the governance structure requirements in the United States are legislated at the state level.

In this study, the interest in the type of governance structure of a firm is related to the differential ability of a member of the TMT to positively influence and impact firm performance dependent, in part, on whether they are embedded in a unitary or dual governance structure. The rationale for a firm performance effect in conjunction with the presence of a senior marketing executive in governance is explained more fully.

Corporate governance statutes are viewed within institutional theory as part of Scott's (1987) coercive pillar. Normative and cultural pressures of the correct way to do things are expressed in statute. So, pressures from all three of Scott's pillars are supporting compliance to expected governance practices, legitimizing and empowering both the firms and their senior managers participating in the governance structure.

When the marketing function, in the form of the senior marketing executive, is included in the corporate governance level of management (board member), it can be expected that this signals high access of the marketing function to resources under the control of the firm. It also signals that the marketing function has the legitimacy and power to wield these resources in fulfillment of objectives and goals within the marketing domain. The type of governance system in which a senior marketing executive is a member will influence the impact of the senior marketing executive.



Although this research is not focused on corporate governance issues per se, it is concerned with the possible ramifications of governance structures on the ability of a senior marketing executive to impact the performance of the firm. I will first describe the basic structural and procedural requirements of each system and then I will explain how each system may bring its own sets of pressures to bear on the senior marketing management.

The One Tier System

Although there are a number of variants of the one tier (unitary) board system, this research is restricted to the U. S. and U. K. variants. The one tier board system used in the U. S. and U. K. is reflective of the Anglo-American governance culture which emphasizes shareholder interests. There is a single board and it acts as both a management body and a supervisory body at the same time. It is comprised of both senior company executives and external directors who are elected by the shareholders. The CEO of the company plays a very prominent role in the one tier board and may even be the Chairman of the Board. Because members of the firm's executive management are members of the board, the non-executive board members have direct exposure to senior firm management and their information when assessing strategic plans or supervising business activities. As a result, unitary boards are active in both management and supervision of operational activities and the development of strategic plans.

For the U.S. headquartered and incorporated companies, corporations are directly governed by state laws of incorporation, not federal law. Most states have adopted the Model Business Corporation Act (MBCA), a set of laws designed to harmonize



www.manaraa.com

incorporation and governance law across states. As a practical matter, many large public corporations are incorporated in the state of Delaware and, although Delaware is not an adherent to the MBCA, Delaware corporate governance law does not deviate from the general description of the one tier system described here.

For the U. K. headquartered and incorporated companies, the U.K. Corporate Governance Code comprise the set of good governance recommendations that public corporations are expected to follow, and have been a part of the statutory listing requirements for companies on the London Stock Exchange. Although compliance is not required, almost all public corporations follow the unitary board governance structure.²

The Two Tier System

The two tiered (dual) board system is used in Germany and most other Continental European countries. However, some European countries allow public stock corporations to choose which board form they would like to follow. Therefore, I have chosen to limit the present research to companies that are both headquartered and incorporated in Germany. In Germany, the dual board system is ensconced in several regulatory statutes, the German Stock Corporation Act (*Aktiengesetz*), the German Codertermination Act (*Mitbestimmungsgesetz*) and the German Corporate Governance Code.

The German regulatory statutes require that German stock corporations be managed by an executive management board (Vorstand) that is comprised only of

² There are a number of good overviews of the governance regimes in the European Union including Davies, Paul L.,"Board Structure in the UK and Germany: Convergence or Continuing Divergence?" Available at SSRN: http://ssrn.com/abstract=262959 or http://dx.doi.org/10.2139/ssrn.262959



executive directors and is responsible for the management of the corporation. They also require a second board, a supervisory board (Aufsichtsrat), that is comprised only of non-executive directors, including employee representatives, whose purpose is to supervise the activities of the management board. ³

The management board's mandate and activities are both operative and strategic and are comparable to those of senior executive managers in a U.S. corporation. However, members of the German management board, including the Chairman, are considered co-equals and manage the corporation's activities collectively. The formal wording of this charge in German law is "primus inter pares" and exhorts that the members of the management board act as equals in a collective management of the corporation's activities. This places collective responsibility and liability on the members of the management board for all activities and decisions, and importantly for this research, assumes a high degree of competence of each member in all business matters. However, individual management board members may be given greater responsibilities in various functional areas of business activity, for example, marketing. If this is the case, this is indicated officially in proxy materials, but does not absolve the other management board members of co-responsibility as indicated under the German Stock Corporation Act and the German Corporate Governance Code. A member of the management board that has been given specific responsibilities (*Ressortverantwortung*) for Marketing would be the equivalent to a Chief Marketing Officer in a U.S. corporation.

³ The Aktziengestz is part of the larger Bundesgesetzbuch (BGB) codex. It guides actions related to stock corporation legal entities. It was passed into law in 1966 and comprises 410 separate paragraphs, though some have been repealed. The German Corporate Governance Code is part of the Aktiengesetz.



The two tier governance system centralizes the responsibility for the management of the firm within the management board. The members of the management board are collectively responsible for the firm's operative and strategic management decisions. Responsibility for specific business domains, including marketing, may be expressly assigned to one or more individuals, but this is not always the case. It is likely that in such a collective scenario, unless the marketing responsibility is explicitly stated, decisions affecting marketing activities may be swayed by members with power, but little marketing competence, or that marketing issues, because they do not have a dedicated advocate with market specific knowledge, will have little influence in the strategic planning and decision making processes.

Further, in the two tier system, the supervisory board (Aufsichtsrat) may be more distant and isolated from managerial concerns of functional areas such as marketing. This is due to several characteristics of the two tier system in Germany: one, the board size tends to be very large (on average 20 members) which reduces the opportunity for marketing relevant information to be communicated to supervisory board members (Milne, 2007); two, half of all supervisory board members are labor representatives due to the codetermination requirement (*Mitbestimmungsrecht*) which dilutes the equity performance orientation of the board (Dammann, 2003); three, the supervisory board includes representatives of creditors, who, as debt holders, are typically reticent to invest corporate resources into future oriented strategic marketing projects, but would rather marshal resources to the satisfaction of obligations (Kraus and Britzelmaier, 2011); four, the board has only one inside director, the chairman (Karus and Britzelmaier, 2011). This leads to a lower level of marketing relevant information and orientation among the



supervisory board and thus less support for pro-marketing strategies and resource allocations. The result is lower integration of marketing relevant information into plans and controls in two tier governance arrangements than would be expected in the case of one tier governance.

When one or more members of the management board explicitly carry responsibility for the marketing domain, there is a clear indication that marketing activities are recognized as being very important and are being represented at the highest levels of management. Explicit marketing responsibility of management board members in the two tier governance system will affect marketing activities in a similar fashion as having a senior marketing executive in a one tier board.

The presence of a senior marketing executive on a one tier-board provides them with direct contact with executive and non-executive directors. They, therefore, have the opportunity to fully communicate and advocate marketing imperatives and prerogatives, to build close relationships and better manage conflicting agendas directly with both managerial and supervisory stakeholders on all strategic and operative decisions.

The type of governance system used in a country in which a firm is headquartered and incorporated is a cultural expression of socially constructed understandings of the best way to supervise organizations (Scott, 1987, 2001). The governance system will exert more than just a ceremonial influence. It will also exert a functional influence by giving a senior marketing member both soft and hard influence with which to support decisions critical to the firm's success in the marketplace.



In general, it is expected that the presence of a senior marketing executive in a one tier board system will be more effective in representing the marketing domain than a senior marketing executive in a two tier board system. Thus:

Hypothesis 11a: The effect of the presence of a SME on the revenue/profitability performance metrics of the firm will be positively moderated when the SME is a member of a unitary board, and greater than if they are member of a dual tier board.

Hypothesis 11b: The effect of the presence of a SME on the market value performance metrics of the firm will be positively moderated when the SME is a member of a unitary board, and greater than if they are member of a dual tier board.

Figure 2.7 shows the proposed moderating relationship of institutional level factors. The societal context variables, Marketing Acceptance and SME Board Membership, and the institutional variable, SME Isomorphic Pressure, act as contingency moderators on the direct relationship between the presence of a SME on the TMT and firm performance.

Structural Factors:

Industry Turbulence

Some industry segments are highly uncertain because of segment growth. When consumer demand grows rapidly there is a tremendous amount of uncertainty. There is an ever increasing flow of information about the external and internal environments that needs to be assessed quickly and accurately. The marketing information processing





Figure 2.7: The moderating effects of institutional factors on the SME's impact on firm performance

requirements include information about customer preferences, the capabilities of competitors, the progression of technological developments, and the appropriate structure for distribution and service networks, optimal pricing, and sales forecasting (Porter, 1985).

Top management teams need to make long-term strategic plans and investment decisions regarding capital investments, market entry, and product development, to name a few. The interpretation of market information requires the input of experienced and capable marketing managers. It is expected that the presence of a senior marketing executive in firms which operate in industries that are growing quickly are able to take advantage of market conditions, nimbly adjust to any changes, and more accurately forecast future market conditions. The more dynamic the environment is, the more important it becomes for management to be able to scan and interpret the environment (Cyert and March, 1963; Daft and Weick, 1984; Finkelstein and Hambrick, 1996;



Hambrick, Finkelstein and Mooney, 2005). Therefore, it is expected that firms in turbulent industries that have a senior marketing executive in their TMT will be better able to take advantage of their market's turbulence (Homburg, Workman, and Krohmer, 1999). Thus:

Hypothesis 12a: The effect of the presence of a SME on the revenue/profitability performance metrics of the firm will be positively moderated the greater the level of turbulence in the industry segment.

Hypothesis 12b: The effect of the presence of a SME on the market value performance metrics of the firm will be positively moderated the greater the level of turbulence in the industry segment.

Internationalization

Firms which have subsidiaries in many different country markets operate in a complex environment. Similar to turbulent markets, the information processing demands of senior management are high for organizations with many subsidiaries embedded in different national contexts, as are the coordination and control requirements (Ambos and Schlegelmilch, 2007; Powell, 1986; Martinez and Jarillo, 1991; Xu and Shenkar, 1994). Multinational firms have to have managerial structures and capabilities that allow them to process market information and coordinate and control marketing activities across many different markets simultaneously and effectively (Bartlett and Ghoshal, 1989; Kotler, 1984; Moorman and Rust, 1999).



It is important that marketing capabilities be part of the senior executive decision making team in order for a firm to accurately and effectively assess market conditions (Cyert and March, 1963; Finkelstein and Hambrick, 1996; Moorman and Rust, 1999). In addition, the competing control and coordination demands of activity integration and localization increase as the number of multinational subsidiaries increases (Martinez and Jarillo, 1991). This adds to the cognitive complexity to which executives are exposed and increases the decision resource requirements of the marketing domain in order to effectively manage this complexity (Day, 1994; Simon, 1979).

The type of marketing strategy that a firm chooses will also influence how much demand will be made on senior management resources for information processing, control and coordination. A multi-domestic corporate strategy delegates most marketing decision responsibilities to the subsidiary level, whereas a global marketing strategy may require considerably more central marketing resources to monitor and coordinate the marketing functions at all levels of the organization (Porter, 1986).

Senior marketing executive roles and responsibilities are often linked with brand management and the development and maintenance of a consistent corporate brand image across all units and markets of the firm. The more national subsidiaries a firm has the more complex and difficult the task of coordinating, controlling, and managing marketing activities and messages across these units (Nath and Mahajan, 2008).

The results of prior empirical research on the impact of geographic diversification on firm performance are inconsistent, but tend to fall on the side of increased operating performance (Grant, 1987). This is in keeping with the resource based view of the firm



and the ability to arbitrage resources across national borders to enhance firm performance (Barney, 1991; Kogut, 1992). The combined effects of geographic diversification and product diversification have a quadratic relationship to firm size. This indicates that as the firm increases in geographic diversity, marketing resources become less impactful and less able to overcome control and coordination issues. (Tallman and Li, 1996). Thus:

Hypothesis 13a: The effect of the presence of a SME on the revenue/profitability performance metrics of the firm will be positively moderated the greater the level of internationalization of the firm.

Hypothesis 13b: The effect of the presence of a SME on the market value performance metrics of the firm will be positively moderated the greater level of internationalization of the firm.

Figure 2.8 shows the proposed moderating relationship of structural level factors. The variables of industry turbulence (industry segment concentration, industry segment growth) and internationalization act as contingency variables on the direct relationship between the presence of a SME on the TMT and firm performance.

Strategic Factors:

Brand Strategy

Firms which pursue a corporate branding strategy have a tremendous amount of firm value dependent on the effective management of the brand. They use the brand name for all, or most, of the firm's products and services. It is highly likely that a corporate





Figure 2.8: The moderating effects of structural factors on the SME's impact on firm performance

branding strategy will be more successful if the firm dedicates senior management resources to the effective management of the brand. Brand management is within the domain of marketing and the executive responsible should be the senior marketing executive, not a mid-level manager, which is often the level of brand managers (Keller, 2003; Rao, Agarwal and Dahlhoff, 2004; Webster, Malter, and Ganesan, 2005). Thus:

Hypothesis 14a: The effect of the presence of a SME on the revenue/profitability performance metrics of the firm will be positively moderated with the greatest positive effect seen with a corporate branding strategy.

Hypothesis 14b: The effect of the presence of a SME on the market value performance metrics of the firm will be positively moderated with the greatest positive effect seen with a corporate branding strategy.



Customer Type

The product market in which a firm is primarily active defines the customer type which is important to its success and the marketing activities in which it engages. A common typology of a firm's customers and activities is that of being either business-to-business (B2B) or business-to-consumer (B2C). The B2B organization is typified as having repeat transactions with customers over a long period of time, direct relationship development, sales based on technical issues, custom product development, fewer customer numbers, and demand being driven by product performance characteristics (Ford, et al. 2003; Hakansson, Johanson, and Wootz, 1976, Harmon, Conrad, and Brown, 1997). The burden of satisfying customer needs and managing customer relationships lies with sales and product development domains. The classic consumer marketing outreach is less relevant than in consumer product B2C markets.

Since there are often fewer customers in a B2B business than B2C, it is expected that the customers will have relatively more power than in a B2C setting. As Boyd, Chandy, and Cunha (2010) have shown, lower power of the senior marketing executive relative to powerful external customers relates to lower firm performance. It can be expected that B2B firms will perform lower than B2C firms and that senior marketing executives will have relatively less impact in B2B firms than in B2C firms. Thus:

Hypothesis 15a: The effect of the presence of a SME on the revenue/profitability performance metrics of the firm will be positively moderated with the greatest positive effect seen when the firm is primarily B2C oriented.



Hypothesis 15b: The effect of the presence of a SME on the market value performance metrics of the firm will be positively moderated with the greatest positive effect seen when the firm is primarily B2C oriented.

Product Type

As has been previously described, this research applies a further distinction in the type of product-market in which a firm is active, whether the firm is primarily active in the service goods market or the product goods market. This is important because service goods and product goods differ on many dimensions which are sensitive to the presence of a senior marketing executive.

Service goods are less tangible, their production and consumption take place at, or very nearly, the same time, they are highly variable in their characteristics (less standardized), they are more perishable, they have shorter lifecycles, and their markets usually have lower barriers to entry. In addition, the marketing environments of service goods are more uncertain and dynamic than product goods marketing environments (Srinivasan, Lilien, and Sridhar, 2011; Zeithaml, Parasumaran, and Berry, 1985). The dynamic and uncertain market conditions of service products require a greater dedication of firm resources to assess and process market information, acquire, keep, and identify new customer needs in order to stay ahead of the competition (Srinivasan, Lilien, and Sridhar, 2011). It is expected that the performance of firms which are primarily engaged



www.manaraa.com

in a service industry and have a senior marketing executive in their TMT will perform better. Thus:

Hypothesis 16a: The effect of the presence of a SME on the revenue/profitability performance metrics of the firm will be positively moderated with the greatest effect seen when the firm is primarily service goods oriented.

Hypothesis 16b: The effect of the presence of a SME on the market value performance metrics of the firm will be positively moderated with the greatest effect seen when the firm is primarily service goods oriented.

Figure 2.9 shows the proposed moderating relationship of strategic level factors. The strategic variables of brand strategy (corporate brand, house brand or mixed brand), product type (financial or physical) and customer type (business to business, business to customer or a mixture of both) act as contingency variables on the direct relationship between the presence of a SME on the TMT and firm performance.



Figure 2.9: The moderating effect of strategic factors on the SME's impact on firm performance



CHAPTER THREE RESEARCH DESIGN AND METHODOLOGY

Research on the structure and effectiveness of senior executives has typically made use of survey data. However, this research uses secondary data sources instead in order to enable a longitudinal approach to data capture and analysis across factor level groupings (institutional, structural, and strategic) for longer periods of time. It is felt that secondary longitudinal data would be more effective when addressing issues associated with environmental forces, as in the case of contingency and institutional influences on organization structure, and organization structure on firm performance because of the likelihood of a temporal lag between cause and effect (Nath and Mahajan, 2008; Zorn, 2004).

Sample and Sources of Data

The sample used in this research is comprised of firms selected using Standard & Poor's COMPUSTAT North America and Global databases over a span of eleven years (2000-2010). The firms were studied over a time period in which a worldwide economic downturn took place that is commonly viewed to have begun in the spring of 2008. This is intentional, to further extract information about the relationship between the environment and the firm. The period of 2008 through 2010 is addressed separately in the results section. Only USA firms with annual sales greater than \$250 million in year 2002



are included.⁴ Because 2002 acts as an "anchor" year and the data sets are not balanced, the largest number of firms appear in the data set in 2002. There is attrition in both prior and later years. The firm size restrictions are relaxed in the samples from Great Britain and Germany. In both the UK and Germany, the firms with greater than \$100 million in sales in year 2002 were included. If this had not been done, it would have been difficult to establish sample sizes in the industry segments of interest large enough to make meaningful inferences.

The data are unbalanced panel data collected over a timeframe of 11 years. The data set includes 7,112 firm years of observations which cover 724 individual firms from three countries. Within the aggregated data set, the USA data includes 524 firms and 4,936 firm years of data, the German data includes 124 firms and 1,339 firm years and the UK data includes 76 firms and 837 firm years of observations. Tables A.1 in Appendix A describe the distribution of the firms across industry segments in the overall sample and in each of the country subsamples. In keeping with the prior industry segment selection of Nath and Mahajan (2008), the observations were collected in the two digit SIC industry segments of 25, 26, 28, 30, 33, 34, 35, 36, 38, and 73.⁵ Because the data set is unbalanced, the number of observations can vary depending on the variables being assessed.

⁴ Since this research builds on prior studies (Hambrick and Cannella, 2004; Nath and Mahajan, 2006, 2008, 2011), the same general dataset building guidelines established in these studies were followed for the USA sample. The major restriction is that all firms without R&D expense data are dropped from the data set. ⁵ The sample of firms used in the study are limited to the following two-digit Standard Industrial Classifications (SIC): Business Services (73), Industrial Machinery & Equipment (35), Electrical & Electronic Equipment (36), Instruments & Related Products (38), Chemicals & Allied Products (28), Rubber & Miscellaneous Plastic Products (30), Fabricated Metal Products (34), Furniture & Fixtures (25), Paper & Allied Products (26), and Primary Metal Industries (33).



In addition to the main sample data set just described, a second data set was also created using the selection criteria published by Nath and Mahajan (2008, p. 70). This data set was used to recreate, as closely as possible, their results, and to test their model on an extended time frame of eleven years (increasing it from the original five years). Despite careful attention to their directions, it was not possible to exactly replicate a data set with the same descriptive statistics for the same five-year time period from 2000-2004. The replicated data contained 166 firms and 757 firm years, rather than the 167 firms and 668 firm years found in the original data. A more detailed comparison of the descriptive statistics and distribution of the firms across industry segments is given in Appendix D. Although not identical, the two data sets are very similar. As a result, the replicate set was used for comparative analyses.

Variables and Measures

Model 1: The Antecedent Model

Table 3.1 summarizes the variables and measures used in testing the proposed antecedents to the presence of a SME. Only secondary sources were used for the data collection. These sources included the databases offered through Standard & Poor's COMPUSTAT, the PASSPORT MONITOR database, annual reports (Form 10-K, 20-F, or similar annual reporting), proxy reports and corporate websites.

The operationalization of the SME was done by identifying the most senior executive identified by "marketing" in their title, or with explicit functional responsibility for marketing, in a firm's top management team (TMT), as reported to the responsible



national financial authorities (SEC, FSA or BaFin). They may carry the Chief Marketing Officer, Executive Vice President, Senior Vice President, Vice President, Director, or other titles. Following Hambrick and Cannella (2004), this analysis assumes that the decision to have a senior marketing executive in the TMT is revisited every year. The TMT has been operationalized in a variety of ways by previous researchers (Gordon et al., 2000). This research follows the definition employed by Hambrick and Cannella (2004) and recognizes the TMT as any executive team manager named in the 10-K or proxy filings with the Securities and Exchange Commission (SEC) for U. S. listed public stock corporations; the same is true for those executives named to the executive council in the annual report or other filings with the Financial Services Authority (FSA), Financial Conduct Authority, or precursor entities for listed public stock corporations in Great Britain; the same is true for those executives named to the management committee in the annual report or other filings with the *Bundesanstalt fuer Finanzdiensleistungsaufsicht* (BaFin) for listed public stock corporations in Germany.

Dependent Variable

Using the definitions for the TMT and senior marketing executive presented here, the presence or absence of a senior marketing executive in the TMT of a firm is established. The presence or absence of the senior marketing executive is then coded as 1, for the presence of a senior marketing executive in the TMT, or 0 for the absence of a senior marketing executive in the TMT.

61



www.manaraa.com

Independent Variables

Institutional Factors

Isomorphic Pressure: The isomorphic pressure within an industry segment (organizational field) regarding the most appropriate executive structure is captured by ranking the largest (and presumably most successful and legitimate) firms (in revenue) for each year within a segment to create a rank of the top four (CR4) firms using a Herfindahl-Hirschman index approach (McAlister, Srinivasan, and Kim, 2007). The ratio of firms within the CR4 with a SME compared to those without a SME is calculated for the CR4 group within each two-digit SIC industry segment for each fiscal year. This creates a variable which will vary in value from 0 to 1. The rationale for this approach is the belief that the chosen organizational structure of the largest (and historically most economically successful) firms within a segment carry more legitimacy in signaling superior structural organization to the rest of the industry segment members.

Societal Context: The societal context of the country in which an organization is embedded is represented by the country in which it is headquartered. The location of the headquarters is the environment which exerts the most influence on the organization's choices of organizational structure and management composition. In addition, it is the environment from which the TMT members most often originate, and it is the environment that is the most influential in informing the opinions and cognitive processes of the TMT members.

Countries differ on the degree to which their societies embrace marketing activities (Wurf, Bakker and Picard, 2008). The degree to which marketing and



www.manaraa.com

marketing activities are considered legitimate is, in part, a function of the societal context in which these activities take place. The degree of collective legitimacy of marketing at the societal level is expressed through the mechanism of the activities and structures of marketing considered valid. In order to assess the degree to which marketing as a discipline and its activities are legitimized within a particular societal context, the level of media in advertising activity in the focal countries is used as a proxy for the level of consensus. Prior research supports this approach as an accurate measure of the marketing orientation and also the degree of acceptance of marketing as an organizational function in a country (Bilektine and Haack, 2015; Tost, 2011; Wurff, Bakker and Picard, 2008). To capture the societal orientation (legitimation) of a country toward marketing, marketing acceptance, two measures are used. The first measure, *Marketing Context* is a composite measure comprised of, 1) the total expenditures on media and advertising divided by the national gross domestic product and, 2) the total expenditures on media and advertising divided by the population (per capita value). These measures are then standardized and summed for an index score by country and year, giving the *Marketing Context* variable.⁶ In this way, country level measures, indicating the total allocation of resources to activities strongly associated with marketing, both relative to the size and mix of the economy and the population, are representative of the level of professionalization of media and advertising activities and

⁶ Media and advertising expenditures are historically the largest single marketing expense category (40% or greater of the total marketing outlays) in the USA, UK and Germany (Barwise and Styler, 2002). A variety of sources indicate that although media and advertising expenditures are influenced by many factors such as the economic cycle, the mix of industrial and consumer segments, the level of economic development, etc., they represent, in aggregate, a relatively stable percentage of total economic activity at the country level (Wurff, Bakker and Picard, 2008).


the legitimization of marketing (Pan, Zinkhan and Sheng, 2007; Srinivasan and Hannsens, 2009).

Structural Factors

Industry Turbulence: The degree of industry segment turbulence is assessed by two different measures. These measures are, 1) the industry segment growth rate, which is measured as a running average of the segment's total sales over two successive years at the country and two-digit SIC level. The degree of change in sales growth, either positive or negative, is an indication of the amount of turbulence in the segment (Eisenhardt and Bourgeois, 1988), and 2) the level of market concentration of an industry segment at the two-digit SIC level. The literature has identified that a lower market concentration index indicates a higher level of competition and therefore a more turbulent environment (Eisenhardt and Bourgeois, 1999). The Hirschman-Herfindahl industry concentration index is used as a measure of the level of competition in an industry segment (Hou and Robinson, 2006). The formula for calculating the industry concentration (IC) level in each segment was as follows:

IC
$$_{jt} = \sum_i s^2_{ij}$$

where s_{ij} is the market share of the firm i in industry j. The concentration index was calculated for each country and fiscal year. Each measure was standardized and then the two standardized values were summed to create a composite measure of industry segment turbulence. The concentration being a negative value and the growth being a positive value.



www.manaraa.com

Туре	Variable	Measure	Value	Source
	Senior Marketing	Indicator of the presence of a SME in the TMT (1), or		Annual reports (10-K, 20-F), proxy
DV	Executive	not (0).	0 or 1	statements or corporate website
	Learner his December	The incidence ratio of SME (presence/absence) within	0 4= 1	Annual reports (10-K, 20-F), proxy
	Isomorphic Pressure	C4 by year and industry segment	0 10 1	statements or corporate website
		Marketing acceptance: Index of z-transformed country		
Institutional		level measures		
Factors	Markating Assentance	1) advertising and media spend divided by country GDP	Desitive	Passport Monitor database
	Marketing Acceptance	(in millions of US\$)	FOSILIVE	rassport Monitor database
		2) advertising and media spend divided by a country's		
		population (in millions)		
		Turbulence: Composite of z-transformed industry level		
		measures.		
	Industry Turbulance	1) Indicator of the industry segment concentration at the	Positivo	COMPUSTAT
	industry rurbulence	two-digit SIC code level in each year	1 Osnive	COMI USTAT
		2) Indicator of the average growth rate of an industry		
		segment at the two digit SIC code level by year in	Positive	COMPUSTAT
Structural		millions of US\$		
Factors		Internationalization: Index of z-transformed firm level		
1 detoils	Internationalization	measures		
		1) Number of countries in which the firm has	Positive	Corporate Affiliations (LexisNexis) or
		subsidiaries		corporate website
		2) The percentage of a firm's total revenue derived from	Positive	Annual reports (10-K, 20-F), proxy
		foreign sales		statements or corporate website
		Indicates whether the company puruses a corporate		Annual reports (10-K, 20-F), proxy
	Brand Strategy	brand, house of brands or mixed brand strategy (mixed	0, 1, 2	statements or corporate website
a		is base condition)		*
Strategic	G	Indicates whether the company is primarily business to		Annual reports (10-K, 20-F), proxy
Factors	Customer Strategy	customer, business to business or mixed strategy (mixed	0, 1, 2	statements or corporate website
	C D L	is base condition)		
	Service Product	Indicates whether the company product is primarily a	0 or 1	Annual reports (10-K, 20-F) or corporate
	Strategy	service product (1), or physical product (0)	D ::	website
	Firm Size	The natural log of number of employees (in thousands)	Positive	COMPUSTAT
	Year	The fiscal year	Positive	COMPUSTAT
	R&D Intensity	The amount of R&D spend divided by revenue (in	Positive	COMPUSTAT, annual reports (10-K, 20-F)
		millions of local currency)		or corporate website
Controls	CEO Tenure	The natural log of the years the CEO has held office	Positive	Annual reports (10-K, 20-F), proxy
		Indicates a shange in the shief executive officer (1) or		Approximate statements or corporate website
	CEO Change	nucleus a change in the chief executive officer (1), or $p_{0}(0)$	0 or 1	Annuar reports (10-K, 20-F), proxy
		Indicates the presence of a chief operating officer (1) or		Annual reports (10-K, 20-F), provy
	COO Presence	not (0)	0 or 1	statements or cornorate website
L				statements of corporate website

Table 3.1: Variables, measures, values and sources of data for antecedent model

Market Internationalization: Firms that have a large number of subsidiaries embedded in different markets face greater marketing complexity in comparison to firms that do not. Similarly, firms with a significant percentage of their total revenues derived from international markets face an increased risk in association with the complexity of their operating environment (Bartlett and Ghoshal, 1989). Two measures are used to assess a firm's level of market internationalization. The first, *Subsidiaries*, uses the



natural log of the number of countries in which a firm has subsidiaries. The LexisNexis Corporate Affiliations database, filings and annual reports served as the sources for identifying the country locations of subsidiaries.

The second, *Foreign Revenue*, uses the percentage of total firm revenue which is derived from international markets. Annual reports, official filings and corporate websites formed the basis for calculating the percentage of revenue which was derived from foreign markets (Lee and Park, 2006; Sullivan, 1994).

Each measure was standardized and then the two standardized values were summed to create a composite measure of internationalization. Each measure being a positive value.

Strategic Factors

Brand Strategy: The branding strategy is coded using the approach of Laforet and Saunders (1994) which uses three categories: corporate branding, house of brands, and mixed strategies. The type of strategy used by a firm is assigned to one of the three categories based on information provided in their annual reports and websites. The firms' brand architecture is analyzed and, when possible, the revenue associated with the brands identified in order to establish a firm's brand strategy (Rao, Argawal, and Dahlhoff, 2004). The brand strategies are dummy coded as 0, for a corporate brand strategy; as 1, for a house of brands strategy; and as 2, for a mixed strategy.



Customer Strategy: A basic distinction in the marketing literature identifies the customer of a firm's products and services as a business customer (business to business, or B2B) or the general public as a customer (business to customer, or B2C). The annual reports and corporate websites are used to identify whether or not a firm followed a B2B strategy, or a B2C strategy. The customer strategies are dummy coded as 0, for B2B, and 1 for B2C.

Product Strategy: The marketing literature has identified fundamental differences in the marketing requirements, activities and strategies between financial services and physical goods. Prior research by Zeithaml, Parasuraman, and Berry (1985) identified only one of the industry segments used in this research sample as a service goods industry (SIC two-digit segment code 73). However, the firms which were not included in the service product segment were reassessed using annual reports (10-K and 20-F) and corporate websites to confirm that they were following a physical goods product strategy. If evidence of substantial marketing of service goods was identified, then the firm was reclassified as following a mixed service/physical goods product strategy. The product strategies are dummy coded as 0, for physical goods and 1, for services.

Control Variables

A review of the relevant literature indicates that there are a number of variables that may be expected to influence resources dedicated to marketing activities and may thus impact the structural choice of a senior marketing executive.

Size: The increasing scale and complexity of the firm will impact the resources dedicated to marketing processes and personnel (Grant, 1996; Day, 1994; Vorhies,



www.manaraa.com

Morgan, and Autry, 2009). The natural log of the number of employees as reported in COMPUSTAT was used as the proxy for size of the organization.

R&D intensity: The ratio of the amount of investment in R&D, as gross expenditure, to total sales, as reported in COMPUSTAT, is used to describe R&D intensity of the firm. Although this measure has been used in the literature to represent other concepts, such as absorptive capacity (Cohen and Levinthal, 1990), it has also been used in marketing literature to control for the dedication of firm resources to marketing activities and personnel (Vorhies, Morgan, and Autry, 2009; Weigelt and Sarkar, 2009).

Advertising Intensity: The ratio of the amount of advertising spending made by the firm to total sales, as reported in COMPUSTAT, is used to describe the advertising intensity of the firm. This measure has been used in prior research as an indicator of the dedication of firm resources to marketing (Nath and Mahajan, 2008; Vorhies, Morgan, and Autry, 2009). The data for advertising expenses is only available for US companies only in the COMPUSTAT database. So, this variable appears in analyses that use US data only.

Year: The variable is used to control for trends in the incidence of the senior marketing executive as a member of the TMT over the time period of interest. Prior research (Nath and Mahajan, 2008) claim that there is a substantial negative trend in the incidence of senior marketing executives in the TMT. This variable assigns the value of 0 for the first year and increases sequentially through the final year of the time period of interest. Therefore, the time period from 2000 through 2010 would be coded 0 through 11.



www.manaraa.com

Industry Segment: The firms used in the sample are drawn from ten different industry segments. The firm industry effects are controlled by using dummy variables for the industry segments and the two digit SIC level. It is expected that some industry segments are more marketing oriented and may have a higher incidence of the senior marketing executives (Nath and Mahajan, 2008).

CEO tenure: The tenure of a firm's CEO is captured as the natural log of the years in that position. Prior research has indicated that longer tenure may indicate more knowledge, power and influence, reducing the need for a senior marketing executive (Finkelstein and Hambrick, 1990; Hambrick and Cannella, 2004; Nath and Mahajan, 2010).

CEO change: The change of the CEO can influence the subsequent removal or inclusion of a senior marketing executive in TMT (Nath and Mahajan, 2008, 2010).

COO presence: The COO is recognized as being the second in command (below only the CEO), and more influential or powerful than the other senior executives in the TMT. The presence of the COO may impact the operational need for a senior marketing executive (Hambrick and Cannella, 2004). The presence of the COO is coded as 1, and the absence of the COO is coded as 0.

Prior performance: Prior performance was included in the firm performance models and consisted of the one year lagged value of the dependent value being measured.



Model 2. The Firm Performance Model

The dependent variables used to assess firm performance fall into two basic categories: accounting based measures and market (equity) based measures. The COMPUSTAT database was used to create both the revenue and market based measures.

The same variables which acted as independent variables in the preceding antecedent model are now acting, in accordance with contingency theory (Donaldson, 2001) as moderating variables with the presence of a SME in the TMT in the firm performance model. The measures and their sources are listed in Table 3.2.

Dependent Variables

Firm Performance (accounting based): The relationship between marketing resources and capabilities has shown a strong link to revenue and profit based performance measures (Finkelstein and Boyd, 1998; Krasnikov and Jayachandran, 2008). The COMPUSTAT database is used to obtain the sales revenue performance of the firms over the period of interest. COMPUSTAT reports revenue for U. S. based firms in U.S. dollars, British Pounds for the U.K. based firms, and in Euros for firms based in Germany. Revenue growth is calculated as year on year revenue growth to assess the impact of executive structure on revenue (Hambrick and Cannella, 2004). The variable, *Sales Growth,* assesses revenue growth at the firm level. The variable is calculated using the equation: (revenue_t – revenue_{t-t}).

Two measures of profitability, ROS and ROA, are also used. Both measures are well established in the literature as metrics of profitability, particularly the TMT literature



(Crossland and Hambrick, 2007; Homburg, Jensen, and Hahn, 2012; Finkelstein and Boyd, 1998).

Return on Sales: The return on sales measure is calculated by the ratio: Net income / total revenue.

Return on Assets: The return on assets measure is calculated by the ratio: Net income / book value of total assets.

Firm Performance (market based): The relationship between firm market value metrics and marketing has become increasingly popular topic within the marketing literature.

Srinivasan and Hanssens (2009) and Crossland and Hambrick (2007) recently provided an excellent overview of links between marketing activity and firm value. This study uses Market to Book (equity) calculation as proposed by Crossland and Hambrick (2007) to measure stock value effects and the Market to Book Assets ratio represented by the version of Tobin's Q suggested by Pruitt and Chung (1994).

The variable, *Market-to-Book (equity)*, is calculated by the ratio: (common shares outstanding X year end closing price)/book value of common equity.

The variable, *Tobin's q*, is calculated using the ratio: ((common shares X year end closing price) + (long term debt + short term debt)) / book value of total assets.

Moderating variables

Because this research is assessing the impact of organizational executive structures through the lens of the structural contingency theory, informed by institutional



theory, the independent variables from the antecedent model are now characterized as moderating variables in the firm performance model (Donaldson, 2001, p.7).

As such, it is expected that the moderating variable will, at higher levels, enhance the base effect of the senior marketing executive on firm performance. The listing of the moderating variables are as follows: *SME x Isomorphic pressure, SME x Market acceptance, SME x Unitary governance, SME x Industry turbulence, SME x Internationalization, SME x Corporate branding, SME x House branding, SME x Business to business, SME x Business to customer, SME x Service product.*

An additional measure is introduced here, *SME x Unitary Board Member*. This variable assesses the effect on firm performance when the SME is also a member of the board in a unitary governance system.⁷

Control Variables

A review of the relevant literature indicates that several variables have been shown to have direct relationships to measures used in the model to assess the impact of a senior marketing executive on firm performance. These variables are included in the model as control variables.

⁷ Both the United Kingdom and Germany allow companies to choose the *Societas Europaea* (SE) form of incorporation. The SE incorporation allows a firm to choose between a unitary or dual board governance form, irrespective of the governance form required by national law. In these cases, the coding of their governance system follows the documentation submitted to the responsible national authority (FSA or BaFin). Firms that chose the SE incorporation form during the time period studied were dropped from the sample in the year they became an SE corporation. Seven (7) firms chose SE incorporation in the sample during the time period studied, and therefore their chosen governance form was anomalous to their home country's prevailing societal context.



Size: The size of the firm has been shown to have a curvilinear relationship with sales performance (Lee, 2009) and equity performance (Joshi and Hanssens, 2010). Firm size is measured by the natural log of the total number of employees.

R&D intensity: The amount of resources dedicated to R&D has been shown to positively impact both revenue and equity performance of the firm (Chan, Martin, and Kensinger, 2000; Krasnikov and Jayachandran, 2008), although this impact can be influenced by the firm's industry segment (Rao, Agarwal, and Dahlhoff, 2004). It is calculated by dividing the R&D expenditure by total revenue.

Year: A dummy variable is used to control for trends in economic cycles which will impact both financial and equity based firm performance metrics (Hambrick and Cannella, 2004). The year 2000 is used as the base year.

Advertising Intensity: This measure has been shown in prior literature to be positively related to firm sales performance (Lee, 2009) and equity performance (Joshi and Hanssen, 2010; Wang, Zhang and Ouyang, 2009). Because the data for advertising expenses is only available for US companies in the COMPUSTAT database, this variable is used only for the US data comparison with Nath and Mahajan's (2008) results.

CEO tenure: Prior research has indicated that longer tenure may indicate more power and influence, thusly reducing the impact of other senior executives, including senior marketing executives, and having greater impact on the performance of the firm (Finkelstein and Hambrick, 1990; Hambrick and Cannella, 2004; Nath and Mahajan, 2010). It is the natural log of the total years in office.



Туре	Variable	Measure	Value	Source
		Sales growth: the year on year sales growth at the firm	a .:	COMPUSTAT, annual reports (10-K, 20-F)
		level.	Continuous	or corporate website
	A accumting Matrice	Patum on Salaci Natingome/tatal myanya	Continuous	COMPUSTAT, annual reports (10-K, 20-F)
	Accounting Metrics	Return on Sales. Net income/total revenue	Continuous	or corporate website
		Return on Assets: Net income/book value of total assets	Continuous	COMPUSTAT, annual reports (10-K, 20-F)
DV		Retuin on Assets. Iver income book value of total assets	Continuous	or corporate website
DV		Market to Book (equity): Market value/book value	Continuous	COMPUSTAT, annual reports (10-K, 20-F)
	Market Metrics	common equity	Committod	or corporate website
		Tobin's q: Market value of equity + debt/Book value of	Continuous	COMPUSTAT, annual reports (10-K, 20-F)
		assets		or corporate website
Main Effect	Senior Marketing	The presence of senior marketing executive	0 or 1	Annual reports (10-K, 20-F), proxy statement
Enect	Executive	The incidence ratio of SME (presence/absonce) within		or corporate website A_{proved} reports (10 K 20 E) proves statement
	Isomorphic Pressure	C4 by year and industry segment	0 to 1	Annuar reports (10-K, 20-F), proxy statement
		Marketing acceptance: Index of z-transformed country		
		level measures		
		1) advertising and media spend divided by country GDP		
	Marketing Acceptance	(in millions of US\$)	0 to 1	Passport Monitor database
		2) advertising and media spend divided by a country's	a i	
		population (in millions)	0 to 1	Passport Monitor database
	Unitary board	Indicator of whether the SME is also a member of the	0 == 1	Annual reports (10-K, 20-F), proxy statement
	membership	board within a unitary board structure	0 01 1	or corporate website
		Turbulence: Composite of z-transformed industry level		
		measures.		
	Industry Turbulence	1) Indicator of the industry segment concentration at the	0 to 1	COMPUSTAT
	industry furbacite	two-digit SIC code level in each year	0101	
		2) Indicator of the average growth rate of an industry		
Moderators	5	segment at the two digit SIC code level by year in	0 to 1	COMPUSTAT
		millions of US\$		
		Internationalization: Index of z-transformed firm level		
		1) Number of countries in which the firm has		Componente Affiliatione (LevieNevie) or
	Internationalization	(1) Number of countries in which the firm has	0 to 1	corporate Amiliations (LexisNexis) or
		2) The perceptage of a firm's total revenue derived from		Appual reports (10-K, 20-E), provy
		foreign sales	0 to 1	statements or corporate website
		Indicates whether the company puruses a corporate		succession of corporate website
	Brand Strategy	brand, house of brands or mixed brand strategy (mixed	0.1.2	Annual reports (10-K, 20-F) or corporate
		is base condition)	•, -, -	website
		Indicates whether the company is primarily business to		
	Customer Strategy	customer, business to business or mixed strategy (mixed	0, 1, 2	Annual reports (10-K, 20-F) or corporate
		is base condition)		wedsite
	Service Product	Indicates whether the company product is primarily a	0 or 1	Annual reports (10-K, 20-F) or corporate
	Strategy	service product (1), or physical product (0)	0.01.1	website
	Firm Size	The natural log of number of employees (in thousands)	Positive	COMPUSTAT
	Year	The fiscal year	Positive	COMPUSTAT
	Prior Performance	The dependent variable lagged 1 year	Continuous	COMPUSTAT, annual reports (10-K, 20-F)
				or corporate website
	R&D Intensity	The amount of R&D spend divided by revenue (in	Positive	COMPUSTAT, annual reports (10-K, 20-F)
Controls		millions of local currency)		or corporate website
	CEO Tenure	The natural log of the years the CEO has held office	Positive	Annual reports (10-K, 20-F), proxy
		Indiantes of a change in the state of a software (1)		statements or corporate website
	CEO Change	indicates of a change in the chief executive officer (1) ,	0 or 1	Annual reports (10-K, 20-F), proxy
		$\frac{1}{1}$		Appual reports (10 K 20 E) provide
	COO Presence	indicates of the presence of a chief operating officer (1), or pot (0)	0 or 1	etatements or corporate website
				statements or corporate website

Table 3.2: Variables, measures, values and data sources for firm performance model



CEO change: The possible impact of the change in CEO on senior executive personnel (Hambrick and Cannella, 2004), as evident in the influence on the subsequent inclusion of senior marketing personnel, change in the personnel, or removal of the position from the TMT. Also, prior research has identified possible effects of CEO change on firm performance (Hambrick and Cannella, 2004). A change in the CEO is coded as 1 in the year it takes place, otherwise it is 0.

COO presence: The COO is recognized as being the second in command (below only the CEO), and more influential or powerful than the other senior executives in the TMT (Hambrick and Cannella, 2004). The presence of a COO is coded as 1, otherwise it is 0.

Industry Segment: Industry effects on firm performance are controlled for by using the median values at the two digit SIC code level to center variables of interest (Hambrick and Cannella, 2004). As a check, industry effects were also controlled for by using dummy variables at the two digit SIC code level.



CHAPTER FOUR

ANALYSES AND RESULTS

The replicate data set built following Nath and Mahajan (2008) are presented in Appendix D along with comparison to their published results and attendant extensions. Since their work acts as the basis for the following analyses, it is suggested that the reader acquaint themselves with the information in Appendix D.

The analyses and results of the antecedent and firm performance models proposed in this study are presented here.

Data set for Antecedent Model and Firm Performance Model

Table 4.1 reports the means, standard deviations, and the correlations of variables pooled across all years (2000-2010). Multicollinearity is not thought to be a problem within the data set. The paired correlations among the independent and control variables are all below 0.6, with the highest being the correlation between firm size (natural log of the number of employees) and the total number of foreign subsidiaries at 0.48. The intervariable correlations among some of the dependent variables were quite high, but they are retained because they are tested separately and they have a theoretical relevance to the research questions being asked. The potential for multicollinearity in variables used in both the antecedent and firm performance models were assessed by calculating the variance inflation factors (VIFs). The resulting VIF values were much lower than the



recommended cut-off value of 10 recommended by Kutner et al. (2004). The mean VIF for the variables used in the antecedent model was 1.43, and the mean values for variables used in the firm performance models was 1.42. These VIF values indicate that multicollinearity is not an important concern for the models presented.

Missing data can pose difficulties for a variety of estimating procedures, including GEE and GLS procedures (Rubin, 1988) used in this research. An assessment was done for the data collected and analyzed for the models presented in this research using the Little's MCAR method as presented by Li (2013). A discussion of the missing data patterns and their analysis is presented in Appendix C. Using the techniques suggested by both Hair, (1998) and Li (2013) it was established that the data used conforms to missing at random (MAR) and therefore the GEE and GLS procedures are efficient estimators when applied to this data set.⁸

The presence of a senior marketing executive in the top management team varied across the data sets of the individual countries and years. Figure 4.1 presents the relative frequencies in the data set. It can be seen that the frequency for the structural choice of a senior marketing executive in the top levels of corporate management have, on the whole, a slight downward trend in both the USA and UK, whereas there seems to have been a slight increase in the Germany firm data since 2006. Whether this increase is a short-term trend, or not, is not clear, but the literature tends to argue that the position of a senior marketing executive on top executive level has been declining in relevance and presence for several decades. It would seem that this trend is indicated, at least in the USA and

⁸ Missing data patterns were not assessed for the replicated data set used in the comparative analyses with published Nath and Mahajan (2008) results.



UK, although all three country environments vary considerably in their incidence rate and trend lines. This is an indication that the respective country environments differ in ways that truly matter when it comes to choices concerning executive structure.



Figure 4.1: The percentage of firms in a given year and country that have a senior marketing executive as a member of their top management team

Because the governance form of the firm plays a role in the firm performance models presented in this study, it should be noted that the SE (Societas Europeae) incorporation option, which allows for an essentially unitary governance form within the dual governance regime countries of the European Union. In the data analyzed, this was only relevant for German firms. The SE form was rarely chosen and represented a very small sub-population within this research data (see footnote 7). This form was not included in the analyses and those firms which chose the SE form were categorized as having a unitary governance form. Three of the German firms which chose the SE option had a SME in their top management team as a member of the Vorstand, which is 43% incidence rate, far above the 26% average of the German sample. It might be interesting to also note that the companies which chose the SE governance form deviated from the



non-SE German companies on many attributes. On average, the SE firms were much larger in size (mean of 38,400 employees versus 15,000), had subsidiaries in more international markets (mean of 25 versus 16), and received a larger percentage of their revenue from international markets (mean of 64% versus 58%).

The analyses and results will be presented in the following manner: first, the antecedent model and the firm performance model are tested using the data set (fiscal years 2000-2010) for the USA, then Germany and then the UK; second, the results of the antecedent and firm performance models are applied to the three-year period of the recession and recovery period (2008-2010).

Two analytic methods are used to evaluate both the antecedent and firm performance models; a random effects panel regression (GLS) method and a random effects generalized estimating equation (GEE) method. Recent research on the executive structure of top management teams has used both GLS and GEE methods of analysis (Hambrick and Cannella (2004); Nath and Mahajan (2006); Zhang (2006). They are used together in this research in order to make useful comparisons with prior research and to give greater robustness to the analyses. The GLS method delivers a random effects estimator which is a matrix weighted average of both the within and between estimators. The GEE method, in comparison, delivers a population averaged estimate by using the within and between estimators and weighting them depending on an assumed covariance structure (Fitzmaurice, et al., 2009).

The GEE method is a population averaging method for estimating that fits generalized linear models to non-independent observations and allows for unknown



www.manaraa.com

correlation among the variables without specifying the origins of the dependence. It can be applied to both logistic and linear models using a link function specification. A further advantage of the GEE method is that it allows a user specified within-group correlation structure (working correlation) in order to efficiently estimate the model coefficients (Pan, 2001). Because of the longitudinal nature of the data and natural grouping at the firm level, it was decided that an autoregressive (AR1) within-group correlation structure would be most appropriate. The appropriateness of this assumption was assessed by analyzing the within group correlations among variables using the quasi-Aikake information criterion (QIC) which confirmed that autoregressive correlation structure fit the data for both of the models (Hardin and Hilbe, 2003).

The data set covers a longer time period (2000-2010) than has been used in prior research on the SME in the TMT and has been extended to three countries (USA, UK and Germany) in order to test the generalizability of findings from the smaller and more limited earlier research. The analyses will be presented first, as a pooled data set, and then each country separately, based on the relevance of the research issue being discussed. The data available for the countries of interest differed in terms of the form and availability in small but important ways, which will be addressed later. However, because of this, the data subsets are analyzed separately, except for those situations in which the data form and availability were comparable. In addition, each of the models (Model 1: antecedents to the presence of an SME and Model 2: impact of the presence of an SME, and moderating variables, on firm performance) will be presented separately (Reminder: the term SME is the more accurate term and is used in this research, rather



www.manaraa.com

	Mean	S.D.	1	2	3	4	5	6	7	8	9	10	11	12
1 Presence of SME	0.29	0.46	1.00											
2 Market-to-book (assets)	1.35	18.33	-0.03 **	1.00										
3 Market-to-book (equity)	1.16	44.54	-0.01	0.03 ***	1.00									
4 Tobin's q	0.46	2.85	0.03 **	0.96 ***	0.24 ***	1.00								
5 Sales growth	0.43	15.18	-0.01	0.00	0.00	-0.01	1.00							
6 ROA	0.04	1.47	-0.02 **	0.72 ***	0.00	0.03	0.00	1.00						
7 ROS	0.05	2.79	0.01	0.00	0.00	0.01	0.00	0.02	1.00					
8 Isomorphic pressure	0.44	0.26	0.10 ***	0.02	0.00	-0.01	0.00	-0.04 ***	-0.02	1.00				
9 Market acceptance	0.00	1.92	0.06 ***	-0.07 ***	-0.01	-0.04 ***	0.01	-0.05 ***	0.01	-0.03	1.00			
10 Unitary governance	0.81	0.39	0.00	-0.09 ***	-0.01	0.01	0.01	-0.07 ***	0.01	-0.06 ***	0.76 ***	1.00		
11 Industry turbulence	-0.01	1.45	0.03 ***	-0.01	-0.03 **	-0.02	0.04 ***	-0.01	0.00	0.09 ***	0.42 ***	0.30 ***	1.00	
12 Number of subsidiaries	16.09	13.96	0.08 ***	0.02	-0.01	0.01	-0.01	0.08 ***	0.04 ***	0.01	-0.02	-0.04 ***	0.00	1.00
13 International sales ratio	0.01	0.05	0.02	0.00	-0.01	-0.02 *	-0.02	0.00	0.00	0.01	0.01	0.01	0.02	0.10 ***
14 Corporate brand strategy	0.55	0.50	0.04 ***	0.04 ***	-0.02 *	0.02 *	0.01	0.03 **	-0.01	0.04 ***	0.01	0.03 **	-0.01	-0.01
15 House of brands strategy	0.07	0.25	-0.01	-0.01	0.00	0.01	-0.01	-0.01	0.00	-0.02	-0.14 ***	0.02	-0.07 ***	-0.03 **
16 Business-to-business	0.59	0.49	-0.08 ***	0.01	-0.02 **	-0.06 ***	-0.02 *	-0.04 ***	-0.02 *	0.08 ***	0.04 ***	-0.05 ***	0.07 ***	-0.12 ***
17 Business-to-customer	0.08	0.27	0.06 ***	0.00	0.02 *	0.10 ***	0.00	-0.01	0.00	-0.07 ***	-0.16 ***	0.05 ***	-0.10 ***	0.07 ***
18 Product type	0.22	0.42	0.07 ***	-0.02	0.00	0.04 ***	0.01	-0.03 *	0.00	0.16 ***	-0.03 **	0.05 ***	0.24 ***	0.02 **
19 Size ^a	1.63	1.39	0.01	-0.02 *	-0.01	-0.02	0.02 *	0.08 ***	0.05 ***	-0.11 ***	0.11 ***	0.16 ***	0.00	0.48 ***
20 R&D intensity	0.18	2.05	0.02	0.00	0.00	0.00	0.00	0.00	0.37 ***	0.00	0.03 **	0.03 **	0.01	0.01
21 CEO tenure ^a	6.26	5.61	0.06 ***	0.01	-0.01	0.06 ***	-0.01	0.01	0.00	-0.02 **	0.02 *	0.01	0.03 **	-0.08 ***
22 Presence of COO	0.29	0.46	0.03 *	-0.03 *	0.00	0.02 **	0.02	-0.02 *	0.00	0.05 ***	0.12 ***	0.10 ***	0.06 ***	-0.04 ***
23 Change of CEO	0.12	0.33	-0.01	-0.01	-0.01	-0.03 ***	-0.01	-0.02	0.00	-0.01	0.00	-0.01	0.02	0.00

Table 4.1: Means, standard	l deviations and	correlations
----------------------------	------------------	--------------

*p<.1,**p<.05, ***p<.01

a= log transformed variable

Notes: Data set includes USA, UK and Germany; Independent and control variables are lagged one fiscal year (t-1); All dependent variables and structural variables are centered b

المنسارات المستشارات

	Mean	S.D.	13	14	15	16	17	18	19	20	21	22	23
13 International sales ratio	0.01	0.05	1.00										
14 Corporate brand strategy	0.55	0.50	0.01	1.00									
15 House of brands strategy	0.07	0.25	0.04 ***	-0.29 ***	1.00								
16 Business-to-business	0.59	0.49	0.00	0.26 ***	-0.11 ***	1.00							
17 Business-to-customer	0.08	0.27	-0.02	-0.02	0.21 ***	-0.34 ***	1.00						
18 Product type	0.22	0.42	0.03 ***	0.11 ***	-0.02 *	0.02 *	0.00	1.00					
19 Size ^a	1.63	1.39	0.05 ***	-0.09 ***	0.03 ***	-0.15 ***	0.03 **	-0.04 ***	1.00				
20 R&D intensity	0.18	2.05	0.01	0.01	-0.02	-0.01	-0.01	-0.01	0.02	1.00			
21 CEO tenure ^a	6.26	5.61	-0.03 ***	-0.03 **	-0.03 ***	0.00	0.01	-0.02 *	-0.06 ***	0.02	1.00		
22 Presence of COO	0.29	0.46	-0.03 **	0.01	-0.03 ***	-0.01	0.02	0.03 ***	0.01	-0.01	0.09 ***	1.00	
23 Change of CEO	0.12	0.33	0.04 ***	0.00	0.00	-0.01	-0.01	0.01	-0.01	-0.01	-0.34	-0.02 **	1.00

Table 4.1: (continued) Means, standard deviations and correlations

*p<.1,**p<.05, ***p<.01

a= log transformed variable

82

Notes: Data set includes USA, UK and Germany; Independent and control variables are lagged one fiscal year (t-1); All dependent variables and structural variables are centered by the two digit SIC segment for the country and year.

than CMO as used by Nath and Mahajan (2008), though this can be slightly confusing. However, the definition is the same.).

The GEE method is the analytic method that has previously been used to test antecedent models with binary response variables and is used here for the antecedent model. The Stata statistical package is used with the panel data format command "*xtgee*." Both the GEE and the random effects panel regression are used to assess the firm performance models. The random effects method uses the generalized least squares (GLS) approach to estimate the coefficients and is applied in order to estimate the important time invariant components of the models. The Hausman test showed no significant difference between the fixed effects estimator when compared to the random effects estimator. In both the logistic and linear models, robust standard errors are specified for the GLS method using the "*vce(cl firm id)*" variance estimator option which allows for intragroup correlation, and "*vce(robust)*" for the GEE method which allows for valid estimation of the standard errors even when the specified correlation structure is inaccurate. The panel regression method used applies the "*xtlogit*" (antecedent model) and the "*xtreg*" (firm performance model) commands.⁹

Serial correlation and endogeneity within the data structure of longitudinal panels is problematic for making accurate estimates of both standard errors and coefficients. Serial correlation is primarily addressed by lagging the predictor and control variables from the dependent variables by one year. This temporal separation is considered

⁹ Stata 13.0 statistical package. XT commands are used for longitudinal/panel data analyses. The three XT commands used here are the *xtlogit* (logistic), *xtreg* (regression), and *xtgee* (generalized estimating equation).



sufficient and is standard practice in the literature (Nath and Mahajan, 2008; Zhang, 2006). However, lags of up to three years were also tested and the results did not depart from the results presented here. Also, as mentioned elsewhere, the GEE approach includes a robust variance estimator which can correct for non-independence in clustered data. Temporal separation between predictor and control variables in relation to the dependent variable helps ameliorate potential endogeneity effects by addressing reverse causality. Augmented regression was used to test for endogeneity in the firm performance models (Nath and Mahajan, 2008; Woolridge, 2013)). This method did not indicate that endogeneity was a problem in the models as specified.

Model 1: The Antecedent Model Analyses

We are interested in assessing the contingent conditions which act as antecedents to the presence or absence of a senior marketing executive (SME) in the top management team. Due to the binary nature of the dependent variable, a logistic model is used for the hypotheses 1-7. A random effects approach is used to take into account the presence of both binary and continuous independent variables in the model. GEE has been applied to similar analytic scenarios (Nath and Mahajan, 2008; Zhang, 2006). However, the logistic regression approach has a longer history of use when assessing executive structural choices (Fligstein, 1987; Hambrick and Cannella, 2004).

The hypothesized antecedent conditions were tested using the following logistic regression model:

logit
$$Pr(Y_{it} | X_{it-1}, \mu_{it}) = \beta_0 + \beta 1 X_{1it-1} + \beta 2 X_{2it-1} + \beta 3 X_{3it-1} + \beta_n C_{it-1} + \mu_{it}$$



Where:

 Y_{it} = probability of the presence of SME in top management team is 1, otherwise 0

 β_0 = the intercept of Pr(SME=1)

 β_l = the direct effect of X1_i on Pr(SME=1)

 $X1_{it-1}$ = institutional factors: isomorphic pressure and societal context factors of marketing acceptance and governance at firm *i*, for year *t*-1,

 $\beta 2$ = the direct effect of X2_{*it-1*} on Pr(SME=1)

 $X2_{it-1}$ = structural factors: industry segment turbulence and level of internationalization at firm *i*, for year *t*-1,

 β 3 = the direct effect of X3_{*it*-1} on Pr(SME=1),

 $X3_{it-1}$ = strategic factors: brand strategy, customer type and service product at firm *i*, for year *t*-1,

 βn = the direct effects of the C_{*it-1*} on Pr(SME=1),

Ci = vector of control variables for firm *i* year *t*-1,

 μ_{it} = the randomly varying unique error terms $\alpha_i + \varepsilon_i$ contributed by firm *i*, for year *t*,

(where ε_i : N(0, σ^2)).



The following generalized estimating equation (GEE) approach described by

Liang and Zeger (1986), an extension of the general linear model, was also used to test the model. The GEE model gives the marginal, population averaged response of Y_{it} as:

$$\mu_{ii} = E(Y_{ij})$$
 has a link function to the covariates g (μ_{ij}) = X_{it} β

Where:

 Y_{ij} = the population averaged probability of presence of SME=1, or =0 for firm *i*, in year *t*,

 X_{it} = corresponding to 1 x p vector of covariates,

 β = corresponding to p x 1 vector of parameters,

g(.) =logistic link function: $\log(\frac{\mu i j}{1 - \mu i j})$

Logistic regression is sensitive to some aspects of the data sample, so the steps were taken to assess, specification error, goodness-of-fit, multicollinearity and influential data points. The possibility of misspecification was tested using a linktest (Stata linktest) which indicated that the variables in the model were reasonable and that the model is not misspecified. A general goodness-of-fit assessment was made using McFadden's R² (0.06) which indicated a moderate fit. In addition, Tjur's R² (Tjur, 2009), a relatively recent approach to calculating R² for logistic regressions, was 0.07, indicating a moderate fit. Further testing of the model fit using (Stata *estat gof, group (10)*) the Hosmer-Lemeshow Chi-square test (p>Chi2=0.02) indicated that the model fits the multinational data set well.



Tests of Hypotheses

Table 4.2 presents the results of the logistic GEE random effects regression estimates for the antecedent model. Robust standard errors were specified. There are significant positive antecedent relationships between isomorphic pressure, corporate branding strategy, business to customer strategy and service product strategy with the presence of a senior marketing executive in the top management team. The relationships, however, vary across countries.

Hypothesis 1 predicts that firms operating in an industry segment will be subjected to isomorphic pressures to follow organizational structure practices of the most successful firms. If the leading firms have chosen to have a senior marketing executive as a member of the top management team, then it is likely that the other firms in the segment will also have a management structure that includes a senior marketing executive in the top management team. The coefficient for the USA is positive (in the predicted direction) and significant (p<.05). Thus, there is support for Hypothesis 1 in the USA data. The effects for both the UK and Germany are positive, but not significant. This is partial support for the hypothesis. Table 4.6 gives a summary of the support for the tested hypotheses across the countries analyses.

It is interesting to note here that the predictive power of the presence or absence of a SME in the TMT in the past is highly predictive of that same firm having the same TMT at a future time. This association was tested using logistic regression a subset of the data in which only firms which have observations for SMEs in all eleven years were



	Pred.	USA		UK		German	y
	Sign	coef	se	coef	se	coef	se
Constant		-0.54 **	0.26	-3.03 ***	1.04	-0.90	0.67
Institutional Factors							
Isomorphic pressure	Positive	0.29 **	0.14	0.29	0.63	0.04	0.22
Marketing acceptance	Positive	-0.06	0.05	-0.07	0.12	-0.03	0.15
Structural Factors							
Industry turbulence	Positive	0.02	0.02	0.05	0.05	-0.05	0.03
Internationalization	Positive	0.01	0.07	-0.01	0.28	0.00	0.01
Strategic Factors							
Corporate brand	Positive	0.34 *	0.18	0.01	0.61	-0.41	0.30
House of brands		0.71 *	0.41	0.75	0.73	-0.71	0.77
Business to business		-0.45 **	0.19	0.59	0.72	-0.41	0.45
Business to customer	Positive	0.27	0.43	2.05 ***	0.74	0.23	0.68
Service product	Positive	0.56 **	0.22	0.42	0.62	1.78 ***	0.64
Controls							
Firm size		-0.01	0.06	0.18	0.18	-0.01	0.05
R&D intensity		0.01 *	0.01	4.04	2.57	-1.15	2.02
CEO tenure		0.01	0.01	-0.03	0.04	0.03 **	0.01
COO presence		-0.23 ***	0.07	0.29	0.33	0.38 **	0.19
CEO change		0.04	0.06	0.00	0.23	0.24 ***	0.09
Year		-0.04 **	0.02	-0.01	0.07	0.01	0.03
Wald X^2		45.21 ***		16.21		22.65 *	
Observations		3232		378		790	

Table 4.2: GEE regression of antecedents to the presence of a SME in TMT (2000-2010)

*p<.1, **p<.05, ***p<.01

Notes: GEE regression with robust standard errors, 2000-2010. All variables are lagged one year; Industry effects are controlled for by centering the continuous variables using the two digit SIC mean value. Governance form variable is dropped for country level data.

retained. When the same logistic regressions were run using lagged SME predictors (from one year, through ten year lags), every lagged SME variable was highly significant (p<.01) and positively associated with the SME dependent variable. This was also confirmed using the Chi-square test which indicated that there are highly significant relationships between the lagged SME variables (L1-L10) and the non-lagged SME



variable (chi-square with one degree of freedom between 118.17 (L10) to 3700.0 (L1) and a significance of p<.0001). Although the relationship attenuates as the time between the two variables increases, the significance of association between "before" and "after" states is very high. This would suggest that the driving isomorphic pressure resides at the firm level, rather than at the industry segment level which is tested by the *Isomorphic pressure* variable.

Hypothesis 2 predicts that the presence of a senior marketing executive in the TMT will be positively associated with firms headquartered and incorporated in a country in which marketing related activities receive larger allocation of resources at the aggregate country level. The coefficients for the marketing context was not significant across the countries tested. Thus, Hypothesis 2 is not supported.

Hypothesis 3 predicts that there will be a positive antecedent association between a firm that is facing turbulent environments (fast growing segment, low market concentration in segment) and the presence of a senior marketing executive on the TMT. In this model the coefficients are not statistically significant. Thus, Hypothesis 3 is not supported.

Hypothesis 4 predicts that there will be a positive antecedent association between higher a firm that is facing high levels of internationalization (has subsidiaries in many different country markets and a high percentage of its revenue is derived from foreign markets). In this model the coefficients are not statistically significant. Thus, Hypothesis 4 is not supported.



Hypothesis 5 predicts that there will be a positive (and stronger) antecedent association between the presence of a senior marketing executive in the TMT if the firm is pursuing a corporate branding strategy, than if a house of brands branding strategy is pursued. The coefficient was positive and significant for corporate branding strategy in the USA sample. However, the effect was smaller than the coefficient for a house of brands strategy. The effect was not significant in the UK and Germany, and was actually negative in Germany. Thus, Hypothesis 5 is not supported.

Hypothesis 6 predicts that there will be a positive antecedent association between a firm that is primarily competing in a business to customer (B2C) industry and the presence of a senior marketing executive in the TMT, rather than one which competes primarily in a business to business (B2B) industry. The association between the presence of a senior marketing executive in the TMT and a firm being in a B2C industry was positive, but significant only in the UK. However, it was always more positive and greater than the association with B2B strategy. Thus, Hypothesis 6 is only supported in the UK.

Hypothesis 7 predicts that there will be a positive (and greater) antecedent association between a firm competing primarily in a service goods market and having a senior marketing executive in the TMT, than a firm that competes in a product goods market. Hypothesis 7 is supported in the USA and Germany.



		S	Support for Hypotheses					
Hypothesis	Predicted	USA	UK	Germany				
Hypothesis 1	Positive	Yes (**)	No	No				
Hypothesis 2	Positive	No	No	No				
Hypothesis 3	Positive	No	No	No				
Hypothesis 4	Positive	No	No	No				
Hypothesis 5	Positive	No	No	No				
Hypothesis 6	Positive	No	Yes (***)	No				
Hypothesis 7	Positive	Yes (**)	No	Yes (***)				
		N= 3232	N= 378	N=790				
		385 firms	46 firms	92 firms				

 Table 4.3: Support for hypotheses using GEE regression (2000-2010)

*p<.1; **p<.05; ***p<.01

Note: all independent variables testing hypotheses were lagged one year. Variables for hypotheses 3 & 4 were centered by two digit SIC code. The year was controlled by using a dummy variable with 2000 as the reference year.

The results of the logistic regression analysis of the antecedent model for the time period of 2000-2010 are presented in Table 4.4. The results were consistent with those found using the GEE method. The directions of the coefficients are consistent and all the variables that were previously found to be significant in the GEE analysis are also significant here. The summary of support for the hypothesized results using logistic regression across the countries analyzed is presented in Table 4.5. The results are consistent with those already presented for the GEE analyses.

The recessionary and recovery time period from 2008-2010 was assessed separately in order to identify possible changes in the importance of the role that antecedents might play. Table 4.6 presents the summary of support found for the same set of hypotheses over this particularly tumultuous economic period using GEE analyses. The support for isomorphic pressure and B2C strategy as antecedent conditions for a



	Pred.	USA		UK		Germa	Germany		
	Sign	coef	se	coef	se	coef	se		
Constant		-2.31 **	1.12	-13.42 **	5.38	-5.33 *	2.85		
Institutional Factors									
Isomorphic pressure	Positive	1.44 **	0.67	-1.12	1.35	-0.99	1.38		
Marketing acceptance	Positive	-0.16	0.19	-0.38	0.23	-0.81	0.65		
Structural Factors									
Industry turbulence	Positive	0.11	0.07	0.07	0.14	-0.03	0.20		
Internationalization	Positive	0.01	0.40	0.62	0.61	0.86	0.54		
Strategic Factors									
Corporate brand	Positive	1.60 ***	0.56	-0.06	1.77	-3.21	2.01		
House of brands		1.89	1.25	0.70	1.95	-4.57	4.50		
Business to business		-1.86 ***	0.73	2.98	2.34	-2.55	2.13		
Business to customer	Positive	1.79	1.68	10.66 ***	3.92	2.14	2.67		
Service product	Positive	1.94 *	1.11	2.32	2.01	8.88 **	3.75		
Controls									
Firm size		-0.17	0.29	1.14	0.71	-0.36	0.47		
R&D intensity		0.01	0.03	21.28 *	11.22	-9.19	11.86		
CEO tenure		-0.01	0.03	0.01	0.09	-0.03	0.06		
COO presence		-0.63 **	0.30	0.83	1.00	2.71 *	1.45		
CEO change		-0.25	0.24	0.44	0.36	0.15	0.35		
Year		-0.11 **	0.05	-0.15	0.14	0.00	0.10		
Wald X^2		36.85 ***		23.72 *		9.32			
Observations		3296		502		864			

Table 4.4: Logistic regression of antecedents to the presence of a SME in TMT(2000-2010)

*p<.1, **p<.05, ***p<.01

Notes: Logistic regression with robust standard errors, 2000-2010. All variables are lagged one year; Industry effects are controlled for by centering the continuous variables using the two digit SIC mean value. Governance form variable is dropped for country level data.

SME in the TMT are no longer present in the USA data. Rather, support has shifted to corporate branding and service product strategies. Both the UK and Germany data continued to identify service strategy as an antecedent, though the UK data showed significant support for the B2C strategy. These changes seem to indicate that the



economic conditions influenced the importance associated with the structure contingencies of the model.

		S	Support for Hypotheses					
Hypothesis	Predicted	USA	UK	Germany				
Hypothesis 1	Positive	Yes (**)	No	No				
Hypothesis 2	Positive	No	No	No				
Hypothesis 3	Positive	No	No	No				
Hypothesis 4	Positive	No	No	No				
Hypothesis 5	Positive	No	No	No				
Hypothesis 6	Positive	No	Yes (***)	No				
Hypothesis 7	Positive	Yes (*)	No	Yes (**)				
		N= 3296	N= 502	N=864				
		399 firms	60 firms	107 firms				

*p<.1; **p<.05; ***p<.01

Note: all independent variables testing hypotheses were lagged one year. Variables for hypotheses 3 & 4 were centered by two digit SIC code.

The year was controlled by using a dummy variable with 2000 as the reference year.

Table 4.6: Country	v comparison	of antecedent	t model hype	otheses from	2008-2010
Tuble Hor Country	, comparison	or anteccuent	i mouei nyp	Juneses mon	2000 2010

		Support for Hypotheses					
Hypothesis	Predicted	USA	Germany	UK			
Hypothesis 1	Positive	No	No	No			
Hypothesis 2	Positive	No	No	No			
Hypothesis 3	Positive	No	No	No			
Hypothesis 4	Positive	Yes (**)	No	No			
Hypothesis 5	Positive	No	No	No			
Hypothesis 6	Positive	No	No	Yes (**)			
Hypothesis 7	Positive	Yes (**)	Yes (**)	Yes (**)			
		N=816	N=255	N=159			
		277 firms	86 firms	54 firms			

*p<.1; **p<.05; ***p<.01

Note: all independent variables testing hypotheses were lagged one year.

Variables for hypotheses 3 & 4 were centered by two digit SIC code.

The year was controlled by using a dummy variable with 2000 as the reference year.



Model 2: The Firm Performance Model Analyses

In the firm performance model we are interested in the moderating effects of the institutional, structural, and strategic contingency variables in the presence of a SME on both market based and accounting based performance metrics. The direct effect of the presence of a senior marketing executive in the top management team on the performance metrics was also assessed. Again, as in the assessment of the antecedent model, both the GEE and GLS methods of analysis are used. It is assumed that variance across firms influences the firm performance variables and that the non-time variant variables are important to the models, so a random effects method is used for both analytic approaches. The potential impact of a SME on firm performance is addressed by hypotheses 8-16 and modeled using the following linear equation:

 $Y_{it} = \beta_0 + \beta 1 X 1_{it-1} + \beta 2 X 2_{it-1} + \beta 3 X 3_{it-1} + \beta 4 X 4_{it-1} + \beta 5 X 1_{it-1} * X 2_{it-1} ... * X 4_{it-1} + \beta_n C_{it-1} + \mu_{it}$ Where:

 Y_{it} = the predicted firm performance for firm *i* in year *t*,

 β_0 = the intercept of Y_{it} ,

 β_1 = the direct effect of X1_{*i*} on Y_{*i*t},

 $X1_{it-1}$ = the presence of a SME at firm *i*, for year *t*-1,

 $\beta 2$ = the direct effect of X2_i on Y_{it},

 $X2_{it-1}$ = institutional factors: isomorphic pressure and societal context factors of marketing acceptance and governance at firm *i*, for year *t*-1,



 $\beta 3$ = the direct effect of X3_{*it-1*} on Y_{*it*},

 $X3_{it-1}$ = structural factors: industry segment turbulence and level of internationalization at firm *i*, for year *t*-1,

 $\beta 4$ = the direct effect of X4_{*it-1*} on Y_{*it*},

 $X4_{it-1}$ = strategic factors: brand strategy, customer type and service product at firm *i*, for year *t*-1,

 $\beta 5$ = the interaction effects of X_{1it-1} and X_{2it-1}, X_{3it-1} and X_{4it-1} on Y_{it},

 βn = the direct effects of the C_{*it-1*} on Y_{*it*},

Ci = vector of control variables for firm *i* year *t*-1,

 μ_{it} = the randomly varying unique error terms α_i + ε_i contributed by firm *i*, for year *t*, (where ε_i : N(0, σ_i 2)).

Both the GLS (linear random effects) and GEE methods using random effects have been used in prior research on firm performance and executive positions in the TMT (Hambrick and Cannella, 2004; Zhang, 2006). The GEE method is considered to be of particular benefit in research using panel data because GEE estimates the regression coefficients and the standard errors using a weighting procedure to compensate for the serial correlations which occur in panel data. This increases the efficiency of the estimates compared to other analytic procedures (Zhang, 2006; Nath and Mahajan, 2008).



The GEE model for the marginal response (population averaged firm performance) of Y_{ij} is given as:

 $\mu_{ii} = E(Y_{ij})$ has a link function to the covariates g (μ_{ij}) = X_{it} β

Where:

 Y_{ij} = firm performance

 X_{it} = corresponding to 1 x p vector of covariates

 β = corresponding to p x 1 vector of parameters

g(.) = identity link function

Revenue-based and market value-based performance metrics were used to assess the impact of a SME on firm performance. All continuous variables were centered (by country, fiscal year, and industry segment). Prior to centering, the dependent variables were winsorized at the 1% level. To address reverse causality, the independent variables were lagged one fiscal year.

As a precursor to the main analyses, the empirical evidence for an association between the presence of a SME on the TMT and firm performance was assessed using a T-test. In order to perform the exploratory test, two groups of firms were identified within the data, an SME positive group and an SME negative group. Of the firms which appeared in each of the eleven (11) years of the time period of interest, one group comprised those firms which had a SME in every year, the other group was comprised of



those firms which did not have a SME in any year during those eleven years. This created for the USA data set a SME (+) group with 51 firms and a SME (-) group with 160 firms, for the UK data set a SME (+) group with 6 firms and a SME (-) group with 37 firms, and for the Germany data set a SME (+) group with 20 firms and a SME (-) group with 58 firms. The SME (+) and (-) groups were then compared using a T-test on each of the firm performance metrics for each country separately. The Satterthwaite approximation was used to account for unequal variances in the response variables. The results are presented in Appendix B. Table B.1 presents the results and indicates that the presence of a SME over the full eleven-year time period is associated with a greater level of Tobin's q and market to book asset value in the USA data and greater levels of sales growth, return on assets and market to book asset value in Germany than companies that did not have a SME for the same time period. The results of the GEE and GLS test methods used to elucidate the hypothesized relationships between the presence of a SME and firm performance is presented next.

Tests of Hypotheses

Table 4.7 presents the estimation results for the USA data using the GLS random effects regression for the time period 2000-2010. It is presented here for representation purposes. The estimation tables for the Germany (Table E.2) and UK (Table E.3) results are in Appendix E. A summary of the support for the proposed hypotheses using the GLS method are presented here for both the accounting based metrics (Table 4.8) and the market based metrics (Table 4.9). The GEE method was also used to analyze possible effects of a SME on firm performance. The GEE estimation results are presented in



Appendix E in tables E.4-E.6 for all three countries and the summary of hypothesis support in tables E.7 and E.8.

Hypothesis 8 predicts that the presence of a SME on the TMT will exert a positive and significant effect on firm performance. Using the GLS random effects method, no significant main effect of the presence of a SME on the accounting based firm performance metrics was found in the USA or Germany data. The UK delivered a positive and significant association (Beta=0.16) with return on sales at the 5% level. Using the GEE approach, the main effect was positive and significant (Beta=0.27) with Tobin's q at the 10% level in the USA data. No main effects were found with the accounting based metrics in any of the countries using either the GLS or GEE methods. Thus, this hypothesis was partially supported in the USA and UK data only. As mentioned previously, Weinzimmer, et al (2003) did find some evidence for an impact on sales growth in their USA data, but Nath and Mahajan (2008) did not (see Appendix D). So far, evidence of a direct main effect on market related performance metrics has also been lacking. However, German et al. (2015) did find evidence for the presence of a positive and significant effect on Tobin's q (Nath & Mahajan, (2008) reported no effect). The results presented here further underline the lack of direct support for the argument that a SME directly influences firm performance.

Hypothesis 9 predicts that a SME presence on the TMT, when this is consonant with the prevailing isomorphic pressure in the industry segment, will lead to higher firm performance. No support was found for any of the firm performance measures in any of the countries analyzed using the GLS method. The GEE method did present a significant



association (Beta= 0.73) with Tobin's q in the Germany data. No other associations were found. Thus, this hypothesis was minimally supported.

Hypothesis 10 predicts that a SME presence on the TMT, when the societal level of marketing activity is high, will lead to higher firm performance. No support was found for any of the firm performance measures in any of the countries analyzed using either GLS or GEE methods. The level of advertising and spending within a country does not seem to moderate the impact of the presence of a SME on firm performance measures used in this study. Thus, this hypothesis was not supported.

Hypothesis 11predicts that a SME presence on the TMT and on a unitary board will lead to higher firm performance. The interaction term *SME x Board membership* was associated with a positive (Beta=0.12) and significant impact at the 1% level on return on assets using the GLS method, and using the GEE method (Beta=0.02) at the 5% level in the USA data. No support was found in the Germany data and the UK data did not have any observations where the SME was also a board member. The membership of a SME on a board does not positively impact the market based performance measures used in this study. Thus, this hypothesis was partially supported in the USA data only.

Hypothesis 12 predicts that a SME presence on the TMT, in the presence of high industry turbulence, will lead to higher firm performance. The interaction term *SME x Turbulence* was associated with Tobin's q positively (Beta=0.08) and statistically at 10% in the Germany data using the GLS method. This was the only statistically significant relationship seen between the interaction term and the performance measures. Thus, there is minimal support for this hypothesis.


Hypothesis 13 predicts that a SME presence on the TMT, when the level of internationalization of the firm is higher than the industry norm, will lead to higher firm performance. The interaction term *SME x Internationalization* is positively (Beta= 2.52) and significantly associated with equity market to book values at the 5% level in the USA data, and with sales growth at the 10% level (Beta=0.02) in the Germany data. These results were present in the GEE analyses as well, indicating that there is substantial support for a moderating effect of internationalization on these two performance metrics. Thus, there is partial support for this hypothesis.

Hypothesis 14 predicts that a SME presence on the TMT will lead to higher firm performance when the firm pursues a corporate brand strategy, rather than a house of brands or a mixed strategy. The interaction term *SME x Corporate branding strategy* has a positive and statistically significant association with Tobin's q (Beta=0.08) and asset market to book valuation (Beta= 0.86) using both GLS and GEE methods in the UK data. The corporate branding coefficient is greater than the house of brands strategy. The USA and Germany data sets do not deliver any significant associations. These results indicate a relationship between the interaction term and market based performance metrics in the UK data. Thus, there is partial support for this hypothesis in the UK data only.

Hypothesis 15 predicts that a SME presence on the TMT will lead to higher firm performance when the firm pursues a business to customer strategy, rather than a business to business or mixed strategy. No support was found for a positive and statistically significant relationship between the interaction term *SME x Business to Customer* and any of the firm performance measures across the countries analyzed, using either GLS or GEE methods. Although the effect on the firm performance measures was,



in almost all cases, positive or less negative than the effect of the business to business interaction, none of the effects were significant. Thus, this hypothesis was not supported.

Hypothesis 16 predicts that a SME presence on the TMT will lead to higher firm performance when the firm pursues a service business strategy, rather than a physical product strategy. The interaction term *SME x Service product* was positively (Beta= 0.07) and significantly associated at the 5% level with return on sales in the Germany data set, using both the GLS and GEE methods. This is solid evidence of support in this particular data subsample. However, there were no other significant relationships. The support is partial for this hypothesis in the Germany data only.

Approximately 69% of the data were USA data observations, 19% Germany and 12% UK. The GEE analytic procedure dropped observations when the time increment (fiscal year) for the panel was inconsistent (non-sequential year measurements). This reduced the observations available for analysis and at times rendered the UK and Germany data sets rather small. However, the smallest sets of observations retained for any single analysis never had less than 40 firms (cluster unit). The number of clusters required for accurate estimation is a matter of discussion, but recommendations in the literature indicate that the number of clusters should be greater than 40 for consistent standard errors and efficient coefficient estimates using GEE (Teerenstra, et al., 2010). Because of the sample was at times reduced to threshold levels in the GEE method, bootstrapping was used to improve the reliability of the standard errors of the coefficients in the UK analyses. Fifty (50) repetitions was found to deliver consistent standard errors for the UK data sample. Bootstrapping was not done for the GLS random effects analyses because, as a subject specific method, rather than a population averaging method,



GLS is more efficient than GEE under the same small sample conditions (Hu, et al.,

1998).10

As a robustness test, rather than centering using median values at the two digit SIC level, dummy variables were used for the industry segment. Using this alternate method to control for industry effects did not influence the outcomes of interest.

 Table 4.7: GLS random effects analysis of SME impact on firm performance in the USA (2000-2010)

	Industry Adjusted						Industry Adjusted					
	Market Based Performance						Revenue Based Performance					
	Tobin's q		MTB(assets) ^a		MTB(equity) ^a		Sales Growth ^a		ROA ^a		ROS ^a	
	coef	se	coef	se	coef	se	coef	se	coef	se	coef	se
Constant	0.25 **	0.12	0.18	0.16	2.68	5.02	-2.64	3.4	-0.05 *	0.03	-0.25	0.19
SME presence	0.04	0.06	0.04	0.11	2.83	2.22	-0.00	2.06	-0.05	0.04	-0.21	0.48
Institutional Interactions									1		1	
Isomorphic pressure x SME	0.19	0.15	0.13	0.14	-7.56 *	4.46	0.33	1.09	0.06	0.04	0.60	0.83
Marketing acceptance x SME	-0.03	0.04	-0.06	0.04	-2.47	1.52	-0.04	1.03	0.01	0.01	-0.16	0.27
Board membership x SME	-0.33 **	0.20	-0.37 **	0.14	0.26	1.53	-1.22	0.96	0.12 ***	0.04	0.22	0.28
Structural Interactions									1		1	
Industry turbulence x SME	-0.01	0.02	-0.02	0.02	0.71	0.49	-4.87	4.07	0.01	0.01	0.04	0.12
Internationalization x SME	-0.02	0.04	-0.02	0.04	2.52 **	1.20	0.47	0.85	-0.00	0.01	0.02	0.08
Strategic Interactions									1		1	
Corporate brand x SME	-0.10	0.11	0.13	0.11	3.36	3.21	-2.82	2.18	-0.02	0.02	0.07	0.28
House of brands x SME	-0.19	0.15	-0.04	0.16	4.98	3.47	-0.19	0.92	-0.02	0.04	0.06	0.24
Business to customer x SME	0.05	0.11	0.11	0.21	6.38	4.35	2.30	3.06	0.07	0.04	0.08	0.10
Business to business x SME	0.04	0.12	-0.06	0.11	1.01	2.85	3.81	2.69	0.04	0.02	0.19	0.10
Service product x SME	-0.15	0.14	-0.06	0.14	-7.52	5.63	1.49	1.32	0.01	0.03	-0.08	0.23
Year ^c	-0.02 **	0.01	-0.01	0.01	-0.18	0.23	0.36	0.24	0.01 **	0.00	0.02	0.02
Wald X^2	239.63 ***		225.10 ***		93.46 ***		94.52 ***		64.60 ***		225.8 ***	
R sq	0.44		0.51		0.01		0.028		0.01		0.162	
Observations	3296		3142		3140		2917		3292		3291	

*p<.1, **p<.05, ***p<.01

a= Tobin's q, MTB(assets), MTB(equity), Sales Growth, ROA and ROS, R&D intensity, and Prior performance are centered at the country

and two digit SIC level to control for industry segment effects.

b=Prior performance is the lagged form of the dependent variable.

c= Year is controlled for as a dummy variable.

¹⁰ The bootstrap procedure was applied using STATA 13.0 command *vce(bootstrap)*. A random number seed (10) was set and subsequent bootstraps of 50, 100, 200, 500, 1000, 1500 and 2000 repetitions were run for each performance model for the UK data, then the Germany data. The smallest repetitions required to achieve stable standard error values for the UK data was 1000. The bootstrap results for the Germany data indicated that stable standard errors where achieved with 200 repetitions.



The GEE method, in general, delivered less conservative results (smaller standard errors) than the GLS random effects approach. The GLS method is the more common approach that is applied in the TMT literature when using continuous dependent variables, and is therefore emphasized here. However, both methods did deliver similar results in the direction of the effects, or when results were significant. In this study the GEE results are looked at as a source of additional support when consonant with the GLS results because of the method's robustness in the face of possible model misspecification and correlated data structures.

Table 4.8: Summary of findings of SME impact on market based firm perf	ormance
metrics using the GLS random effects method	

		Tobin's q				MTB(asset	ts)	MTB(equity)			
Hypothesis	Predicted	USA	Germany	UK	USA	Germany	UK	USA	Germany	UK	
Hypothesis 8	Positive	No	No	No	No	No	No	No	No	Yes (**)	
Hypothesis 9	Positive	No	No	No	No	No	No	No	No	No	
Hypothesis 10	Positive	No	No	No	No	No	No	No	No	No	
Hypothesis 11	Positive	No	No	No	No	No	No	No	No	No	
Hypothesis 12	Positive	No	Yes (*)	No	No	No	No	No	No	No	
Hypothesis 13	Positive	No	No	No	No	No	No	Yes (**)	No	No	
Hypothesis 14	Positive	No	No	Yes (**)	No	No	Yes (*)	No	No	No	
Hypothesis 15	Positive	No	No	No	No	No	No	No	No	No	
Hypothesis 16	Positive	No	No	No	No	No	No	No	No	No	
Observations		3296	806	388	3142	848	477	3140	848	392	
Firms		399	102	52	380	106	60	380	106	53	

*p<.1; **p<.05; ***p<.01

Note: all independent variables testing hypotheses were lagged one year.

Variables for hypotheses 3 & 4 were centered by two digit SIC code.

The year was controlled by using a dummy variable with 2000 as the reference year.

In order to assess the possible impact on the results of the global recessionary

downturn which took place during the time period analyzed, the same analyses were

made using the USA data set for the time periods 2000-2007 and 2008-2010.¹¹ This was

¹¹ The National Bureau of Economic Research (NBER) identified the beginning of the recession in December of 2007 and lasting through June of 2009. For simplicity, this research identifies the recession time frame as starting with 1.01.2008 and continuing through 1.31.2009 in the USA. Analyses using the exact recession start and end dates identified by the NBER for the USA, by Office for National Statistics



not done for the UK and Germany data because the shorter three-year time frame was not stable with the models being tested, even when bootstrapping was applied. This time period did not deliver any significant main or interaction effects for the USA data set. The estimation tables for both GLS and GEE analyses are presented in tables E.9 and E10.

		Sales Growth				ROA		ROS		
Hypothesis	Predicted	USA	Germany	UK	USA	Germany	UK	USA	Germany	UK
Hypothesis 8	Positive	No	No	No	No	No	No	No	No	No
Hypothesis 9	Positive	No	No	No	No	No	No	No	No	No
Hypothesis 10	Positive	No	No	No	No	No	No	No	No	No
Hypothesis 11	Positive	No	No	No	Yes (***)	No	No	No	No	No
Hypothesis 12	Positive	No	No	No	No	No	No	No	No	No
Hypothesis 13	Positive	No	Yes (*)	No	No	No	No	No	No	No
Hypothesis 14	Positive	No	No	No	No	No	Yes (*)	No	No	No
Hypothesis 15	Positive	No	No	No	No	No	No	No	No	No
Hypothesis 16	Positive	No	No	No	No	No	No	No	Yes (**)	No
Observations		2917	782	436	3292	840	477	3292	840	477
Firms		395	107	60	399	106	60	399	106	60

 Table 4.9: Summary of findings of SME impact on accounting based firm

 performance metrics using the GLS random effects method

*p<.1; **p<.05; ***p<.01

Note: all independent variables testing hypotheses were lagged one year.

Variables for hypotheses 3 & 4 were centered by two digit SIC code.

The year was controlled by using a dummy variable with 2000 as the reference year.

for the UK, and by the Statistisches Bundesamt for Germany were also used. The results did not differ from the 1.01.2008-1.31.2009 time frame, so that is the time period used in all models.



CHAPTER FIVE DISCUSSION

This study explores new contingency variables that are theorized to influence the structural choice of a SME as member of the TMT and the resulting impact on firm performance. The analyses drew upon contingency theory, which has been used as an instrumental argument for structural form choice in prior research, and extended the instrumental emphasis by leveraging institutional theory arguments to add richness to the discussion of the motivations for both structural form choice and the resulting impact on firm performance. In the past, research on organizational structural forms as a response to environmental contingencies has delivered tepid or, at times, conflicting results. By identifying new internal and external contingency factors, this study sought to clarify the relationships between the contingencies that motivate an organization to fit its environment and the subsequent realized performance enhancement.

In order to extend the application of contingency and institutional analysis of structural choice, it was important to create a strong link to the prior research in this area. First, by using Nath and Mahajan's (2008) previously published findings, a best effort attempt was made to recreate their data set and analyses. The data set that was developed had very similar summary statistic characteristics to the original. However, only five of their eight variables were constructed and used in the analyses presented here. Four of them



(Innovation, Differentiation, Corporate branding and Outsider CEO) were used because Nath and Mahajan had identified them as significant predictors of a SME on the TMT, and therefore putative contingency variables. A further variable, Market concentration, that had not been identified as a significant predictor in the prior research, but was included because it has possible theoretical relevance for mimetic processes in the form of SME isomorphic pressure.

Comparison to Nath and Mahajan (2008)

According to the contingency theory view, organizations attempt to create an optimal fit between their structural form choices and the conditions they face. Nath and Mahajan (2008) found support for the positive antecedent role of innovation, differentiation, corporate branding, and the presence of an outsider CEO to the choice of a senior marketing executive in the top management team as a structural choice. They were not able to find support for a main effect, or interaction effects, on firm performance. This research, using the replicated data, found support for an association of innovation as an antecedent condition to the presence of a SME. Again, no support was found for a main effect, or an interaction effect, of a SME on firm performance (Tobin's q and sales growth). So, Nath and Mahajan's (2008) research findings on the antecedent contingencies for a SME in the TMT were only partially supported. The significant contingency factor association in the antecedent model (positive for innovation) was reasonable and compelling contingency effect. However, the association with innovation falls away once the industry segment is controlled for. So, even this effect is not really a contingency effect, but could be largely an industry segment effect, although none of the segments were significantly associated with the presence of an SME.



When the time frame for the replicated Nath and Mahajan (2008) data was extended an additional five years, the contingency variables of interest produced different results. The previous significant contingency associations disappeared and a new one, corporate branding strategy, became significant as an antecedent contingency variable. The presence of a SME still did not impact firm performance, either directly or indirectly. When the UK and Germany data for the same time frame were added to the data, all significant antecedent associations disappeared. It seems that the explanatory usefulness of Nath and Mahajan's (2008) model, as presented in prior research, degrades as the time frame is lengthened, or non-USA country data is added. The theoretical structural contingency arguments for the variables used in the model are reasonable, but clearly, the robust prior findings are only partially replicable, and not generalizable. This may, of course, be due to the inability to exactly recreate all the variables used in Nath and Mahajan's (2008) models, or possibly to the small differences between the original and replicate data sets. However, it was expected that the previously published robust effects of the selected contingency factors would also show significant associations with organizational structures when the time frames were extended, or when non-USA data was included. Early deterministic descriptions of contingency theory (Bourgeois, 1984) and even later relaxed "strategic choice" contingency explications of effects on managerial structure (Child, 1972) do not seem to be supported. The results do give indications that there are rationales for having more marketing influence (and less) under different contingency conditions, but these indications are not as robust as implied by earlier published results.



In order to develop more stable and generalizable models than those proposed by Nath and Mahajan (2008), this research proposed updated models which used both institutional and compound contingency variables. The aim was to capture theory coherent effects using variables which tapped more broadly into constructs. These models proposed three categories of contingency variables (institutional, structural and strategic) argued to be salient in the decision to include a marketing executive in the top management team, and which, following contingency theory, would act as moderators. These models, although a variation on previously tested models, had not been proposed before.

Antecedents to presence of a SME in the TMT

The antecedent model is the most important model to the contingency theory orthodoxy that context variables will predict the presence of particular structural response of an organization, in this case a SME in the top management team. The institutional factor contingencies that were included in the antecedent model explored the contingency importance of isomorphic pressures within the relevant organizational fields (industry segments) and societal contexts (country) on the choice of an organization's form of executive structure. Following Meyer and Rowan (1977) and DiMaggio and Powell (1983), both mimetic and normative forces are used to inform and influence firms as to appropriate managerial structures. The institutional contingency factor, isomorphic pressure, was only predictive of the presence of a SME in the TMT in the USA. It was not predictive in the UK or Germany. In addition, *Marketing acceptance*, as an indicator of varying legitimacy for marketing as a functional form had been alluded to in prior research (Homburg, Workman and Krohmer, 1999), was also not predictive of a SME in



the top managerial echelon. In the data, the highest resource dedication was in the USA, followed by the UK and then Germany. It was theorized that higher levels of legitimacy and status may enhance opportunities for marketing managers to reach the highest executive levels. Since prior research had failed to find a relationship between the choice of a SME in the TMT and organizational performance, it was hypothesized that this structural decision might lie with institutional variables that exert contextual conformance pressure when legitimacy is anchored in rationales that only indirectly benefit the organization. The metrics used in this study were not able to identify these influences.

The structural factors measuring industry turbulence and internationalization did not have predictive saliency for a SME in the TMT. Findings by Nath and Mahajan (2008) hinted that structural contingencies might be salient (market concentration).¹² In this study industry turbulence was conceptualized slightly differently from that of Nath and Mahajan (2008). Turbulence was conceptualized more broadly as a compound variable composed of two components, market concentration and industry segment revenue growth. Contrary to Nath and Mahajan's (2008) findings, market concentration was not associated with the presence of a SME in the TMT in any of the countries analyzed. The strength of the result within the USA data (both GEE and GLS methods) might mean that the perception of informational complexity and market uncertainty, and thus the perception of the need marketing resources in the top management team is culturally influenced (i.e. organizations in the UK and Germany might not perceive business activities in numerous foreign markets as being as complex or uncertain as their

¹² Nath and Mahajan (2008) used only the Herfindahl Hirschman index (HHI) as a measure of turbulence. They did find a sizable negative association, though not significant, between HHI and the presence of a SME.



American counterparts). The structural factor contingencies proved to be strong predictors of a SME presence in the USA and fit well with contingency arguments of structural forms responding to organizational requirements to address informational complexity and uncertainty. They were not generalizable as contingency factors to the UK or Germany contexts.

Strategic factor contingencies played a variable role in predicting the presence of a SME in the TMT. A corporate branding strategy was a robust predictor in the USA data, but not in the UK or Germany data. Interestingly, the effect was smaller than for a house of brands strategy, which does not support to the proposed hypothesis and is contrary to prior theorizing in the literature. This result might be an indication that managing high brand complexity supports SME presence because of the informational and cognitive demands it would place on executive level decision processes. Support for the importance of a business to customer strategy as a contingency was evident in both the USA and the UK, but not Germany. This can be interpreted as support for the notion that a customer orientation, which is higher in the US and UK (Homburg, Workman and Krohmer, 1999), is associated with a SME being involved in executive level decision making. Evidence for the hypothesized relationship between a service product strategy and a SME in both the USA and Germany supports the theorized need for a SME when an organization is faced with the nuanced and continuously changing marketing programs needed to address the uncertainty and short cycle time of a service product offering.

The patterns of the antecedent model analyses are unique to each country data set. This makes it impossible to identify a generalizable relationship between the tested contingencies and the presence of a SME in the TMT. There is, however, some



corroboration of prior published findings and very promising identification of associations with isomorphic pressure, business to customer strategies and service markets in the USA data.

SME impact on firm performance

The marketing literature takes the view that marketing is primarily responsible for generating and maintaining demand for a firm's products and services, that the activities and capabilities of the marketing domain create both intangible and tangible assets (Kotler, 1984; Srinivasan and Hanssens, 2009; Webster, 1989). Prior research has had a difficult time confirming this assumption (Nath and Mahajan, 2008, 2010; Weinzimmer, 2003). Germann et al. (2015) did claim to find support for a positive effect on Tobin's q, but not on sales growth.

Although contingency theory does not specify how a specific structural choice might improve firm performance (other than "fit"), it does claim that a specific structural configuration is a rational attempt to improve performance in the face of certain contingencies. A SME in the TMT is expected to positively influence strategic and operational activities which then impact a firm's demand generating capabilities (market sensing and customer linking) to create value for the firm. The presence of the SME, then, through direct and indirectly action, increases the organizational fit with the contingency context (Donaldson, 2001). This study assessed the impact of a SME on firm performance, and by unpacking these performance measures into external market and



internal accounting based measures attempted to identify the impact of a SME to a more fine-grained extent than had been attempted previously.

No significant main effects were found between the presence of a SME in the TMT in the USA or Germany data. The UK data did indicate a main effect on the market to book equity valuations. Only Weinzimmer et al (2003) have previously reported a direct effect on firm performance. In general, the presence of a SME in the TMT does not seem to impact market or accounting based firm performance metrics in any consistent manner. Since the main effect could not be identified using either the population averaging analytic method (GEE), or the subject specific method (GLS), it is likely that the association found in the UK data is an artifact of the sample.

The interaction contingency variables gave mixed results across the performance metric categories, as well as across the countries. In the USA data there was a large positive and significant association with the market to book value of equity performance by the interaction of SME presence with the level of complexity and importance of international market to the firm (*SME x Internationalization*). This could be interpreted as an indication that a SME structure is beneficial to firm performance when the firm's level of international market complexity and dependence is higher than the industry average. From a contingency theory perspective, this is an odd result because internationalization was not significant as an antecedent condition to a SME. Under this theoretical view, the interaction would also be insignificant. This would seem to argue for a less deterministic, but rather nuanced, view of contingency effects than what is offered in theory.



The USA, UK and Germany data gave inconsistent (positive and negative weightings) revenue based performance results. The effects were often small, so it is reasonable to assume that the results oscillated around a nil effect. This study did not deliver the expected association between the modeled contingency variables and firm performance.

Overall, the evidence for an impact of the SME on either market or accounting based firm performance metrics is inconclusive.

Implications for research

This study addresses the findings of prior research into marketing's role at a strategic corporate level and expands the scope of this inquiry to include not only strategic and structural factors, but institutional factors as well.

The findings of this study contribute to the contingency view by explicating the structural contingency factors of the presence of a SME in top management teams by, 1) specifically applying institutional variables and lengthening the time frame over which data was collected in order to effectively assess institutional effects, 2) applying compound structural contingency variables which measure informational complexity and market uncertainty in order to improve construct assessment over prior research, 3) assessing the generalizability of structural contingency effects by extending the analyses to include non-USA data sets, and 4) increasing the sample size in order to address the possibility of small structural contingency effects sizes.

The research results suggest that there are strong context specific antecedent conditions driving the choice to include a SME in the top managerial level of firms, and



www.manaraa.com

that these antecedent conditions vary across countries. In particular, the findings associating a SME structural choice with the need to address information complexity (internationalization and service markets) in the top executive team extends and supports similar findings in prior research (Nath and Mahajan, 2008). The results also suggest that, since the effect is not seen across all countries, that the managerial perception of what constitutes complexity and uncertainty may also vary across countries. Further unpacking of the organizational perception of complexity by using surveys to identify how a SME in the TMT addresses challenges posed to the organization operating in service industries and/or complex international markets would be informative.

The impact of the SME on firm performance in the USA data was not apparent. This was in keeping with prior research findings. And again, factors which identified themselves as important antecedent conditions to a SME in the TMT did not moderate this structure's impact on firm performance. This study did contribute to the discussion of the decisional influence of senior marketing executives by assessing both institutional factors (isomorphic pressure and market acceptance) and structural factors (market turbulence and internationalization) of informational complexity on firm performance. Since these factors did not seem to be related to firm performance, it begs further investigation into the influence of marketing on performance and the rationales for it.

The use of both GEE and GLS random effects methods to assess the antecedent conditions and impact of a SME in the TMT delivered similar results. The generalized estimating equations (GEE), logistic random effects and ordinary least squares methods have been used in similar prior research (Hambrick and Cannella,2004; Nath and Mahajan, 2008; Zhang, 2006). However, prior research has not specified the rationales



behind the methodological choices, or addressed possible implications for results. This study specifically included the GEE and random effects regression methods for both logistic and linear regression models. The results did vary slightly depending on the method used. Generally, the mean response and the impact of the covariates on the mean response (GEE method) delivers more statistically robust inferences (significance) for the relationships of interest than the random effects regression approach. It would seem that the theory of contingency factor effects on firm structure and performance supports the application of a population means analytic approach more than the subject specific inference approach of a random effects regression method. This suggests that means oriented analytic techniques applied to longitudinal data of more than just a few years is probably most appropriate when using the contingency lens.

Implications for practice

Decisions about the functional composition of executive structures can be informed based on the results of this research. The contingency view implies that there are optimal structural choices (Donaldson, 2001; Zeithaml, Varadarajan, Zeithaml, 1988). Although this research indicates that country context differences are important, within the USA context the optimal executive team structure implied by contingency theory includes a SME when the firm faces high levels of industry turbulence, pursues an internationally oriented business strategy (robust across methods) or pursues a corporate branding strategy. Results also seem to indicate that firms pursuing a business to business strategy are not optimizing if they choose to include a SME in their top executive circle.



It has already been noted that these findings indicate a context dependency since they were not consistent across countries. In the UK, optimality of executive structure would indicate that having a SME in the top executive circle would be advisable in situations of industry turbulence, a business to customer strategy or when leading competitors have chosen the SME structure (isomorphic pressure). In Germany, there was a strong indication against the SME structure if a firm pursued a house of brands strategy.

n keeping with prior research, the results of this study delivers inconclusive results regarding the direct impact of the SME on firm performance. In addition, most of the interaction effects on firm performance were not significant. For the USA context there was a strong positive interaction effect on market to book value (a similar metric to tobin's q used by Nath and Mahajan (2008)) in conjunction with isomorphic pressure to have a SME. There was also a positive, though not significant, effect on the market to book (equity) metric as well. This might be reasonably explained by the expectations of the (strong) investment community in the USA rewarding an executive structure that includes a SME when the leading firms in the industry segment include a SME in the top executive team. Otherwise, there were negative effects on performance in the USA, UK and Germany contexts that were associated with situations in which a SME is an optimal choice. This seems to indicate the firms chose the SME structure because of business challenges that warranted the input and skill sets of a SME, but whose impact would not be evidenced in the performance metrics used.

The decision to install a SME in the TMT is clearly a complex one. And, despite claims of a high turnover rate in the SME position in the USA, information gathered in this study did seem to provide cross-national support for the general assertion that short



tenures are the norm (Welch, 2004). The average tenure of a SME in this data set was 2.6 years for the USA, 3.1 years for the UK and 3.2 years for Germany. In comparison, the average CEO tenure in this data set ranged from 6.4 years in the USA, 5.5 years in the UK to 6.2 in Germany. The shorter tenures for the SME might be a reflection of the career ambition of an executive moving on in order to move up the corporate ladder, rather than the result of being dismissed due to unfulfilled expectations or the inherent risk of a job tied to consumer fickleness (Welch, 2004). It is not clear whether the average tenure for a SME diverges significantly from the average tenure for other senior executive positions, such as COO or CFO.

Limitations and further research

There are several limitations to this research. First, this research relied on secondary archival data. Future research that emphasizes primary survey and field data to address the qualitative foundations of management structure decision making would provide more understanding of the mechanisms behind marketing executive selection. Structural contingency theory is based on managerial recognition of the importance of specific contingency factors that must be addressed so that a firm can achieve "fit" and optimize performance. The perceived saliency, or lack of it, for any particular contingency factor drives this "fit" process, and thus the selection of a marketing executive to satisfy "fit" requirements. Because contingency theory has not found the empirical support in longitudinal studies which some of its most ardent supporters expected (Donaldson, 2001), particularly in its application to managerial structures (Nath



and Mahajan, 2008, Zhang, 2006), qualitative research to close this gap and identify the contingencies driving managerial structural change is needed. Field research focused on how executives perceive the contingency factors influencing informational complexity and market uncertainties within the marketing domain might be a fruitful pursuit to give more substance to structural contingency approaches.

Second, the sample used in this research was limited to large manufacturing firms. Future studies should include smaller firms and firms operating in a greater variety of industry segments, particularly those firms which make relatively greater use of marketing resources (e.g. consumer products). The use of large manufacturing firms was dictated by the interest in bridging to prior research. Although the sample used in this study did validate and extended results for manufacturing firms in the USA context, it would have been more informative if the sample had included more data points from non-USA manufacturing firms.

Third, the sample sizes for the country level analyses should be larger. A priori estimates of adequate sample size were made using standard calculations (Hedecker et al., 1999) and effect size estimates from the related executive literature (Nath and Mahajan, 2008; Zhang, 2006) indicated that the sample sizes would be sufficient, even with population averaged estimation methods. However, the effect sizes achieved for some of the variables in the proposed model were extremely small, particularly in the country samples. In order to effectively test the generalizability of the effects found in the USA sample, larger country level samples are required. It might be that, in general, structural contingency effects are very small and therefore require large sample sizes representing longer time frames in order to identify stable patterns.



Fourth, contingency theory emphasizes a deterministic, and primarily linear, relationship between a structural accommodation and a contingency factor. However, U-shaped relationships may very well exist between contingency factors and structural accommodations (Donaldson, 2001). It would be easy to conjecture that low levels of internationalization do not present enough complexity or uncertainty to warrant a SME, but as internationalization increases to a point where a SME would provide essential support at the executive level, they would, but beyond which the marketing domain complexities become so great they must necessarily be dealt with at divisional levels. In any case, the investigation of quadratic variable interactions would be instructive for future research efforts.

Fifth, the inability of this study to show a consistent positive impact by the presence of a senior marketing executive in the top executive team on firm performance, particularly revenue (and margin) based performance. Previous authors have wrestled with the difficulty of identifying compelling metrics with which to assess the impact of executive level marketing resources on firm performance (Slotegraaf and Dickson, 2004; Srinivasans and Hanssens, 2009). It is possible that the models presented are too expansive (containing too many moderators) and need to be reduced in complexity. Future research could focus on reduced models with just a few variables in one factor area at a time. This might allow for a more effective unpacking and teasing out of variables with significant effects. The tendency in contingency models is to attempt to represent the organizational context, leading to complex and unwieldy models.

Sixth, endogeneity is a potential problem with this data set. An augmented regression test did not indicate that endogeneity was a problem. Therefore, the choice of



having a SME in the TMT was treated as exogenous in the firm performance analyses based on standard treatment (Nath and Mahajan, 2008). However, in order to tease out the rather fickle relationships, future research could address the problem of potential endogeneity by using more sophisticated techniques such as propensity score matching or instrumental variables (Germann et al., 2015).

Seventh, the application of an event analysis method could be a fruitful approach for future research. This approach would specifically analyze the firm performance impact of a status change to the presence, or absence, of a SME as a member of the TMT. Further, it would also allow a possible assessment of the impact of consistency in commitment to this particular executive structure.

Conclusion

The importance of having the marketing domain represented in top management circles is probably in a state of functional transition today, but not in a crisis as some seem to want to argue (Day, 1992; Varadarajan, 1992; Welch, 2004). Yes, there does seem to be a slow trend away from the organizational structural choice of having a SME in the top management team. And, yes, this might indicate an eroding of marketing's influence in the strategic planning and other processes of today's firms. But, in the subset of large manufacturing firms, such as those analyzed, there is clearly an association between the presence of a SME in top management when there is greater international orientation in structure and sales, when there is a corporate branding strategy and when there is greater market turbulence. These contextual contingencies might be driving the choice of a SME to improve the firm's structural fit, but the contingency theory argument



that the structural choice of a SME represents an organization attempting to optimize its "fit" to the contextual contingencies facing it in order to optimize performance was not supported. There is a striking lack of association between the presence of a SME and a positive impact on firm performance. Perhaps the performance metrics must be directly linked to SME activities and areas of responsibility in order to identify an association with performance. For example, other possible performance metrics might include market share, brand equity or margins might be effective in establishing a performance link.

I hope that the findings in this dissertation help contribute to a better understanding of the structural choices and performance impact of corporate elites.



www.manaraa.com

REFERENCES

- Aaker, David A., Erich Joachimsthaler. 2000. The brand relationship spectrum:
 The key to the brand architecture challenge. *California Management Review*. 42 (2): 8-23.
- Ambos, Bjorn and Bodo B. Schlegelmilch. 2007. Innovation and control in the multinational firm: A comparison of political and contingency approaches. *Strategic Management Journal*. 28 (1): 473-486.
- Anderson, Eugene W., Claes Fornell and Sanal K. Mazvancheryl. 2004.
 Customer satisfaction and shareholder value. *Journal of Marketing*. 68 (October): 172-185.
- Barney, Jay B. 1991. Firm resources and sustained competitive advantage. *Journal of Management*. 17 (1): 99-120.
- Bartlett, Christopher A. 1986. Building and managing the transnational: The new organizational challenge. In Porter, M. E.: *Competition in Global Industries*. Boston, MA: Harvard Business School Press.
- Bartlett, Christopher A. and Sumantra Ghoshal. 1989. *Managing across borders: The transnational solution*. Boston, MA: Harvard Business School Press.



- Barwise, Patrick and Alan Styler. 2002. *Marketing expenditure trends*. London Business School/Havas Marketing Report, London, UK
- Bitektine, Alex, Patrick Haack. 2015. The "macro" and "micro" of legitimacy:
 Toward a multilevel theory of the legitimacy process. *Academy of Management Review*. 40 (1): 49-75.
- Boyd, D. Eric, Rajesh K. Chandy and Marcus Cunha. 2010. When do chief marketing officers affect firm value? A customer power explanation. *Journal of Marketing Research*. 47 (6): 1162-1176.
- Brettel, Malte, Andreas Engelen, Florian Heineman and Papachong
 Vadhanasindhu. 2008. Antecedents of market orientation: A cross-cultural comparison. *Journal of International Marketing*. 16 (2): 84-119.
- Brush, Thomas H., Philip Bromiley and Margaretha Hendrickx. 2000. The free cash flow hypothesis for sales growth and firm performance. *Strategic Management Journal*. 21 (4): 455-473.
- Buckley, Paul J. and Mark C. Casson. 1976. *The future of the multinational enterprise*. New York, NY: Holmes and Meier.
- Burns, Tom and George M. Stalker. 1961. *The management of innovation*. London, UK: Tavistock.
- Chan, Su Han, John D. Martin and John W. Kensinger. 1990. Corporate research and development expenditures and share value. *Journal of Financial Economics*. 26 (2): 255-276.



- Chandler, Alfred D. 1962. Strategy and structure: Chapters in the History of the American industrial enterprise. Cambridge, MA: MIT Press.
- Child, John. 1972. Organizational structure, environment and performance: The role of strategic choice. *Sociology*. 6 (1): 1-22.
- Child, John. 1997. Strategic choice in the analysis of action, structure, organizations and environment: Retrospect and prospect. *Organization Studies*. 18 (1): 43-76.
- Chung, Kee H. and Stephen W. Pruitt. 1994. A simple approximation of tobin's q. *Financial Management*. 23 (3): 70-74.
- Cohen, Wesley M. and Daniel A. Levinthal. 1990. Absorptive capacity: A new perspective on learning and innovation. *Administrative Science Quarterly*. 35 (1): 128-152.
- Cyert, Richard M. and James G. March. 1963. *A behavioral theory of the firm*. Englewood Cliffs, NJ: Prentice-Hall.
- Daft, Richard L. and Karl E. Weick. 1984. Toward a model of organizations as interpretation systems. *Academy of Management Review*. 9 (2):284-295.
- Dammann, Jens C. 2003. The future of codetermination after centros: Will German corporate law move closer to the U.S. model? *Fordham Journal of Corporate & Financial Law.* 8 (2): 607-687.

Day, George S. 1986. Tough questions for developing strategies. *Journal of Business Strategy*. 6 (3): 60-69.



- -----. 1992. Marketing's contribution to the strategy dialogue. *Journal of the Academy of Marketing Science*. 20 (Fall): 323-329.
- -----. 1994. The capabilities of market-driven organizations. *Journal of Marketing*. 58 (October): 37-52.
- DiMaggio, Paul J. and Walter W. Powell. 1983. The iron cage revisited:Institutional isomorphism and collective rationality in organizational fields. *American Sociological Review*. 48 (2): 147-160.
- Donaldson, Lex. 2001. *The contingency theory of organizations*. Thousand Oaks, CA: Sage Publications.
- Doz, Yves L. and C. K. Prahalad. 1987. *The multinational mission, balancing global integration with local responsiveness*. New York, NY: Free Press.
- Drucker, Peter F. 1985. *Innovation and entrepreneurship*. New York, NY: Harper & Row Publishers.
- Dwyer, Robert F. and John F. Tanner. 2008. *Business marketing: Connecting strategy, relationships, and learning.* (4th Ed.), McGraw-Hill/Irwin.
- Eisenhardt, Kathleen M. and L. J. Bourgeois III. 1988. Politics of strategic decision making in high-velocity environments: Toward a midrange theory. *Academy of Management Journal*. 31 (December): 737-770.
- Engelen, Andreas and Malte Brettel. 2011. A cross-cultural perspective of marketing departments' influence tactics. *Journal of International Marketing*. 19 (2): 73-94.



- Finkelstein, Sydney and Brian K. Boyd. 1998. How much does the CEO matter? The role of managerial discretion in the setting of CEO compensation. *Academy of Management Journal*. 41 (2): 179-190.
- -----and Donald C. Hambrick. 1996. *Strategic leadership: Top executives and their effects on organizations*. New York, NY: West Publishing Company.
- Fitzmaurice, Garrett, Marie Davidian, Geert Verbeke and Geert Molenbergh. 2009. Longitudinal Data Analysis. Boca Raton, FL: Chapman and Hall/CRC Press.
- Fligstein, Neil. 1987. The intraorganizational power struggle: The rise of finance personnel to top leadership in large corporations. *American Sociological Review*. 52 (1): 44-58.
- Ford, David I., Lars-Erik Gadde, Hakan Hakansson and Ivan Snehota. 2003. Managing Business Relationships. New York, NY: John Wiley & Sons, Ltd.
- Fritz, Wolfgang. 1996. Market orientation and corporate success: Findings from Germany. The European Journal of Marketing, 30 (8): 59-74.

Froeschl, Friedrich. 1997. Vitalisierung bei Siemens-Nixdorf.
Booz/Allen/Hamilton (Eds.), Unternehmensvitalisierung. Stuttgart:
Schaefer-Poeschel, pp. 188-206.

Germann, Frank, Peter Ebbes and Rajdeep Grewal. 2015. The chief marketing officer matters! *Journal of Marketing*. Vol. 79, 1-22.



www.manaraa.com

- Glazer, Rashi and Allen M. Weiss. 1993. Marketing in turbulent environments: Decision processes and the time-sensitivity of information. *Journal of Marketing Research*. 30 (November): 509-521.
- Gordon, Shelley S., Wayne H. Stewart, Robert Sweo, William A. Luker. 2000.
 Convergence versus strategic reorientation: The antecedents of fast-paced organizational change. *Journal of Management*. 26 (5): 911-945.
- Grant, Robert M. 1987. Multinationality and performance among British manufacturing companies. *Journal of International Business Studies*. 18 (1): 79-89.
- -----. 1996. Prospering in dynamically-competitive environments: Organizational capability as knowledge integration. *Organization Science*. 7 (4): 339-360.
- Greve, Henrich R. 2000. Marketing niche entry decisions: Competition, learning, and strategy in Tokyo banking. *Academy of Management Journal*. 43 (5): 816-836.
- Hair, Joseph F. Jr., Rolph E. Anderson, Ronald L. Tatham and William C. Black.

1998. *Multivariate Data Analysis* (5th ed.). Upper Saddle River, NJ: Pearson Prentice-Hall.

Hakansson, Hakan, Jan Johanson and Bjorn Wootz. 1976. Influence tactics in buyer-seller processes. *Industrial Marketing Management*. 5 (6): 319-332.



- Hambrick, Donald C. 1983. High profit strategies in mature capital goods industries: A contingency approach. *Academy of Management Journal*. 26 (4): 687-707.
- ----- and Albert A. Cannella. 2004. CEOs who have COOs: Contingency analysis of an unexplored structural form. *Strategic Management Journal*. 25 (October): 959-979.
- -----, Sydney Finkelstein and Ann C. Mooney. 2005. Executive job demands: New insights for explaining strategic decisions and leader behaviors. *Academy of Management Review*. 20 (3): 472-491.
- -----, Theresa Seung Cho and Ming-Jer Chen. 1996. The influence of tope management team heterogeneity on firms' competitive moves. *Administrative Science Quarterly*. 41 (4): 659-684.
- Hardin, James. W. and Joseph M. Hilbe. 2003. *Generalized Estimating Equations*. Boca Raton, FL: CRC Press, LLC.
- Harmon, Harry A., Craig A. Conrad and Gene Brown. 1997. Industrial buyer behavior: Toward an understanding. *The Journal of Marketing Management*. 7 (1): 101-114.
- Haunschild, Pamela R. 1993. Interorganizational imitation: The impact of interlocks on corporate acquisition activity. *Administrative Science Quarterly.* 38 (4): 564-592.



- ----- and Anne S. Miner. 1997. Modes of interorganization imitation: The effects of outcome salience and uncertainty. *Administrative Science Quarterly*. 42 (3): 472-500.
- Haveman, Heather A. 1993. Follow the leader: Mimetic isomorphism and entry into new markets. *Administrative Science Quarterly*. 38 (4): 593-627.
- ------ and Hayagreeva Rao. 1997. Structuring a theory of moral sentiments: Institutional and organizational coevolution in the thrift industry. *American Journal of Sociology*. 102 (6): 1606-1652.
- Homburg, Christian, Ove Jensen and Alexander Hahn. 2012. How to organize pricing? Vertical delegation and horizontal dispersion of pricing authority. *Journal of Marketing*. 76 (5): 49-69.
- Homburg, Christian, John P. Workman and Harley Krohmer. 1999. Marketing's influence within the firm. *Journal of Marketing*. 63 (2):1-17.
- Hopkins, David S. and Earl L. Bailey. 1971. The chief marketing executive. *Conference Board Report*. New York, NY: The Conference Board, Inc.
- ------ and -----. 1984. Organizing corporate marketing. *Conference Board Report*. New York, NY: The Conference Board, Inc.
- Hou, Kewei and David T. Robinson. 2006. Industry concentration and average stock returns. *The Journal of Finance*. 61 (4): 1927-1956.



- Hu, Frank, Jack Goldberg, Donald Hedeker, Brian Flay and Mary Pentz. 1998.
 Comparison of population-averaged and subject-specific approaches for analyzing repeated binary outcomes. *American Journal of Epidemiology*. 147 (7): 694-703.
- Hymer, Stephen H. 1966. *The international operations of national firms*. Cambridge, MA: MIT Press.
- Jelinek, Mariann. 1977. Technology, organizations, and contingency. Academy of Management Review. 2 (1): 17-26.
- Joshi, Amit M., Dominique M. Hanssens. 2009. Movie advertising and the stock market valuation of studios: A case of "great expectations?". *Marketing Science*. 28 (2): 239-250.
- ----- and -----. 2010. The direct and indirect effects of advertising spending on firm value. *Journal of Marketing*. 74 (January): 20-33.
- Keller, Kevin L. 2003. Designing and implementing brand strategies. *Strategic brand management*. Upper Saddle River, NJ: Pearson Education, Inc.
- Keith, Robert J. 1960. The marketing revolution. *Journal of Marketing*. 24 (Janauary): 25-38.
- Kerin, Roger, Vijay Mahajan and P. Rajan Varadarajan. 1990. Contemporary Perspectives on Strategic Market Planning. Boston, MA: Allyn and Bacon.



- Kile, Charles O., Mary E. Phillips. 2009. Using industry classification codes to sample high-technology firms: Analysis and recommendations. *Journal of Accounting, Auditing & Finance*. 24 (1): 25-58.
- Kotler, Philip. 1984. Design: A powerful but neglected strategic tool. Journal of Business Strategy. 5 (2): 16-22.
- Krause, Patrick and Bernd Britzelmaier. 2011. Corporate governance and corporate performance: A German perspective. *International Journal of Management*. 13 (3): 327-340.
- Krasnikov, Alexander and Satish Jayachandran. 2008. The relative impact of marketing, research-and-development, and operations capabilities on firm performance. *Journal of Marketing*. 72 (July): 1-11.
- Kumar, V. and Denish Shah. 2009. Expanding the role of marketing: From cutomer equity to market capitalization. *Journal of Marketing*. 73 (November): 119-136.
- Laforet, Sylvie and John Saunders. 1994. Managing brand portfolios: How the leaders do it? *Journal of Advertising Research.* 34 (September): 64-76.
- Lane, Christel. 2003. Changes in corporate governance of German corporations:
 Convergence to the Anglo-american model? *Competition & Change*. 7 (2-3): 79-100.



- Lawrence, Paul R. and Jay W. Lorsch. 1967. Organizations and environment: Managing differentiation and integration. Boston, MA: Graduate School of Business Administration, Harvard University.
- Lee, Jim. 2009. Does size matter in firm performance? Evidence from US public firms. *International Journal of the Economics of Business*. 16 (2): 189-203.
- Lee, Ho-Uk and Jong-Hun Park. 2006. Top team diversity, internationalization and the mediating effect of international alliances. *British Journal of Management*. 17(3), 195-213.
- Li, Cheng. 2013. Little's test of missing completely at random. *The Stata Journal*. 13(4): 795-809.
- Little, R. 1988. A test of missing completely at random for multivariate data with missing values. *Journal of the American Statistical Association*. 83: 1198-1202.
- Malone, Thomas W. and Kevin Crowston. 1994. The interdisciplinary study of coordination. *ACM Computing Surveys*. 26 (1): 87-119.
- March, James G. and Herbert A. Simon. 1993. Organizations revisited. *Industrial* & Corporate Change. 2 (3): 299-316.
- Martinez, Jon I. and Carlos J. Jarillo. 1991. Coordination demands of international strategies. *Journal of International Business Studies*. 22 (3): 429-444.



- McAlister, Leigh, Raji Srinivasan and MinChung Kim. 2007. Advertising, research and development, and systematic risk of the firm. *Journal of Marketing*. 71 (1): 35-48.
- McGahan, Anita M. and Michael E. Porter. 1997. How much does industry matter, really? *Strategic Management Journal*. 18 (summer special issue): 15-30.
- McGovern, Gail J., David Court, John A. Quelch and Blair Crawford. 2004. Bringing customers into the boardroom. *Harvard Business Review*. 82 (11): 70-80.
- Menon, Anil, Sundar G. Bharadwaj, Phan Tej Adidam and Steven W. Edison. 1999. Antecedents and consequences of marketing strategy making: A model and a test. *Journal of Marketing*. 63 (2): 18-40.
- Meyer, John W. and Brian Rowan. 1977. Institutional organizations: formal structure as myth and ceremony. *American Journal of Sociology*. 83 (2): 340-363.
- Milne, Richard. 2007. Germany's two-tier governance system comes under fire. *Financial Times*, May 8: 1-5.
- Mizik, Natalie and Robert Jacobson. 2009. Valuing branded businesses. *Journal of Marketing*. 73 (November): 153-174.
- Moore, Karl J. and Julian M. Birkinshaw.1998. Managing knowledge in global firms. *Academy of Management Journal*. 12 (4): 81-92.



www.manaraa.com

- Morgan, Neil A. and Nigel F. Piercy. 1999. Competitive advantage, quality strategy and the role of marketing. *British Journal of Management*. 7, 231-245.
- Moorman, Christine and Roland T. Rust. 1999. The role of marketing, *Journal of Marketing*. 63 (special issue): 180-197.
- Murphy, John M. 1987. *Branding: A key marketing tool*. London, UK: Macmillan Publishing.
- -----. 1989. Good brand husbandry techniques. In: J. Murphy (Ed.), *Brand valuation: Establishing true and fair view*. London, UK: The Interbrand Group.
- Nath, Pravin. (2006) *Antecedents of the chief marketing officer's presence and influence in top management teams*. (Doctoral dissertation). Retrieved from ProQuest Dissertations and Theses. (Accession Order No. 3244914)
- ----- and Vijay Mahajan. 2008. Chief marketing officers: A study of their presence in firms' top management teams. *Journal of Marketing*. 72 (January): 65-81.
- ----- and -----. 2010. Marketing in the c-suite: A study of chief marketing officer power in firms' top management teams. *Journal of Marketing*. 75 (January): 60-77.

Olins, Wally. 1989. Corporate Identity. London, UK: Thames and Hudson.



www.manaraa.com

- Pan, W. (2001). Akaike's information criterion in generalized estimating equations. *Biometrics*. 57, 120-125.
- Pan, Yue, George Zinkhan and Shibin Sheng. 2007. The subjective wellbeing of nations: A role for nations? *Journal of Macromarketing*. 27 (4): 360-369.
- Pauwels, Koen H., Dominique Hanssens, Jorge Silva-Risso, and Shuba Srinivasan. 2004. New products, sales promotions, and firm value: The case of the auto industry. *Journal of Marketing*. 68 (October): 142-156.
- Piercy, Nigel. 1986. The role and function of the chief marketing executive and the marketing department. *Journal of Marketing Management*. 2 (3): 265-289.

Porter, Michael E. 1985. Competitive Strategy. New York, NY: Free Press.

- -----.1986. *Competition in global industries*. Cambridge, MA: Harvard Business School Press.
- Ramaswami, Sridhar N. 1996. Marketing controls and dysfunctional employee behaviors: A test of traditional and contingency theory postulates. *Journal of Marketing*. 60 (April): 105-120.
- Rangan, V. Kasturi, Rowland T. Moriarty and Gordon S. Swartz. 1992.
 Segmenting customers in mature industrial markets. *Journal of Marketing*. 56 (October): 72-82.


- Rao, Vithala R., Manoj K. Agarwal and Denise Dahlhoff. 2004. How is manifest branding strategy related to the intangible value of a corporation? *Journal* of Marketing. 68 (October): 126-141.
- Reed, Gary, Vicky Story and Jim Saker. 2004. Information technology: Changing the face of automotive retailing? *International Journal of Retail & Distribution.* 32 (1): 19-32.
- Ruekert, Robert W., Orville C. Walker and Kennth J. Roering. 1985. The organization of marketing activities: A contingency theory of structure and performance. *Journal of Marketing*. 49 (Winter): 13-25.
- Scott, W. Richard. 1987. The adolescence of institutional theory. *Administrative Science Quarterly*. 32 (4): 493-511.
- -----. 2001. *Institutions and Organizations*. Thousand Oaks, CA: Sage Publications.
- Segal, Madhave N. 1989. Implications of single vs. multiple buying sources. Industrial Marketing Management. 18 (3): 163-178.
- Simon, Herbert A. 1979. Rational decision making in business organizations. *American Economic Review*. 69 (4): 493-513.
- Simonin, Bernard L. 1999. Transfer of marketing know-how in international strategic alliances: An empirical investigation of the role and antecedents of knowledge ambiguity. *Journal of International Business Studies*. 30 (3): 463-490.



- Srinivasan, Shuba and Dominique M. Hanssens. 2009. Marketing and firm value: Metrics, methods, findings, and future directions. *Journal of Marketing Research.* 46 (June): 293-312.
- Srinivasan, Raji, Gary L. Lilien and Shrihari Sridhar. 2011. Should firms spend more on research and development and advertising during recessions? *Journal of Marketing*. 75 (May): 49-65.
- Srivastava, Rajendra K., Tasadduq A. Shervani and Liam Fahey. 1998. Market based assets and shareholder value: A framework for analysis. *Journal of Marketing*. 62 (January): 2-18.
- -----, ----- and -----. 1999. Marketing, business processes, and shareholder value: An organizationally embedded view of marketing activities and the discipline of marketing. *Journal of Marketing*. 63 (Special Issue): 168-179.
- Suchman, Mark C. 1995. Managing legitimacy: Strategic and institutional approaches. *Academy of Management Journal*. 20 (3): 571-610.
- Sujan, Harish, Barton A. Weitz, and Mita Sujan. 1988. Increasing sales productivity by getting salespeople to work smarter. *Journal of Personal Selling & Sales Management*. 8 (2): 9-20.
- Sullivan, Daniel. 1994. Measuring the degree of internationalization of the firm. Journal of International Business Studies. 25(2), 325-342.



- Tallman, Stephen and Jiatao Li, 1996. Effects of international diversity and product diversity on the performance of multinational firms. *Academy of Management Journal*. 39 (1): 179-196.
- Teerenstra, Steven, Bing Lu, John Preisser, Leo van Achtenberg, George Borm, 2010. Sample size considerations for GEE analyses of three-level cluster randomized trials. *Biometrics*. 66(4):1230-1237.
- Tost, L. 2011. An integrative model of legitimacy judgments. *Academy of Management Review.* 36: 686–710.
- Tjur, T. 2009. Coefficients of determination in logistic regression models—A new proposal: The coefficient of discrimination. *The American Statistician*. 63: 366-372.
- Ulijn, Jan, Arie Nagel, Wee Lliang Tang. 2001. The impact of national, corporate and professional cultures on innovation: German and Dutch firms compared. *Journal of Enterprising Culture*. 9 (1): 21-52.
- Varadarajan, P. Rajan. 1992. Marketing's contribution to strategy: The view from a different looking glass. *Journal of the Academy of Marketing Science*. 20 (Fall): 335-343.
- Varadarajan, P. Rajan, Satish Jayachandran, and J. Chris White. 2001. Strategic interdependence in organizations: Deconglomeration and marketing strategy. *Journal of Marketing*. 65 (1): 15-29.



- Villanueva, Julian and Dominique M. Hanssens. 2006. Customer equity:
 Management and research opportunities. *Foundations and Trends in Marketing*. 1 (1): 1-95.
- Vorhies, Douglas W., Robert E. Morgan and Chad W. Autry. 2009. Productmarket strategy and the marketing capabilities of the firm: Impact on market effectiveness and cash flow performance. *Strategic Management Journal*. 30 (12): 1310-1334.
- Wang, Fang, Xiao-Ping Zhang and Ming Ouyang. 2009. Does advertising create sustained firm value? The capitalization of brand intangible. *Journal of the Academy of Marketing Science*. 37 (2): 130-143.
- Webster, Frederick E. 1981. Top management's concerns about marketing: Issues for the 1980's. *Journal of Marketing*. 45 (Summer): 9-16.
- -----, Alan J. Malter and Shankar Ganesan. 2005. The decline and dispersion of marketing competence. *MIT Sloan Management Review*. 46 (4): 35-43.
- Weigelt, Carmen and M. B. Sarkar. 2009. Learning from supply-side agents: The impact of technology solution providers' experiential diversity on clients' innovation adoption. *Academy of Management Journal*. 52 (1): 37-60.
- Weitz, Barton. 1981. Effectiveness in sales interactions: A contingency framework. *Journal of Marketing*. 45 (Winter): 850-103.
- Weinzimmer, Lawrence G., Edward U. Bond, Mark B. Houston and Paul C. Nystrom. 2003. Relating marketing expertise on the top management team



and strategic market aggressiveness to financial performance and shareholder value. *Journal of Strategic Marketing*. 11 (June): 133-159.

- Welch, Greg. 2004. CMO tenure: Slowing down the revolving door. Spencer Stuart Report, (accessed May 2nd, 2014),[available at: http://content.spencerstuart.com/sswebsite/pdf/lib/CMO_brochureU1.pdf].
- Wernerfelt, Birger. 1984. A resource-based view of the firm. *Strategic Management Journal*. 5 (2): 171-180.
- Wind, Yoram and Vijay Mahajan. 1981. Designing product and business portfolios. *Harvard Business Review*. 59 (1): 155-165.
- Woolridge, Jeffrey. 2013. Introductory Econometrics: A modern approach, 5th Edition. Mason, OH: South-Western.
- Workman, J. P., Homburg, C., and Gruner, K. 1998. Marketing Organization: An integrative framework of dimensions and determinants. *Journal of Marketing.* 62 (7): 21-41.
- Wurff, Richard van der, Piet Bakker, and Robert G. Picard. 2008. Economic Growth and Advertising Expenditures in Different Media in Different Countries. *Journal of Media Economics*. 21: 28-52.
- Zeithaml, Valerie A., A. Parasuraman and Leonard L. Berry. 1985. Problems and strategies in services marketing. *Journal of Marketing*. 49 (Spring): 33-46.

Xu, Dean and Oded Shenkar. 1994. Institutional distance and the multinational enterprise. *Academy of Management Review*. 27 (4): 608-618.



- -----, P. Rajan Varadarajan and Carle P. Zeithaml. 1988. The contingency approach: Its foundations and relevance to theory building and research in marketing. *European Journal of Marketing*. 22 (7): 37-64.
- Zeger, Scott and Kung-Yee Liang. 1986. Longitudinal data analysis for discrete and continuous outcomes. *Biometrics*. 42 (1): 121-30.
- Zorn, Dirk M. 2004. Here a chief, there a chief: The rise of the CFO in the American firm. *The American Sociological Review*. 69 (June): 345-364.
- Zuckerman, Gregory and Kris Hudson. 2007. Will RadioShack lead investors to a letdown? *The Wall Street Journal*. April 25: C1-C2.



APPENDIX A

DESCRIPTION OF DATA

Table A.1: Distribution of observations by industry segment

		Data Distribution by Industry Segment							
	-	Tot	al	US.	A	U	K	Germ	any
Two digit	-	Firm		Firm		Firm		Firm	
SIC code	Industry Description	Years	%	Years	%	Years	%	Years	%
25	Furniture and fixtures	161	(2.2)	150	(3.0)	0	(0.0)	11	(0.8)
26	Paper and allied products	271	(3.8)	187	(3.8)	29	(3.5)	55	(4.1)
28	Chemicals and allied products	1,428	(20.1)	1,075	(21.8)	154	(18.4)	199	(14.9)
30	Rubber and miscellaneous plastics products	290	(4.1)	207	(4.2)	39	(4.7)	44	(3.3)
33	Primary metal industries	325	(4.6)	189	(3.8)	59	(7.1)	77	(5.7)
34	Fabricated metal products, except machinery and transportation equipment	318	(4.5)	205	(4.2)	64	(7.6)	49	(3.7)
35	Industrial and commercial machinery and computer equipment	1183	(16.6)	746	(15.1)	38	(4.5)	399	(29.8)
36	Electronic and other electrical equipment and components, except computer equipment	974	(13.7)	722	(14.6)	83	(9.9)	169	(12.6)
38	Measuring, analyzing, and controlling instruments; photographic, medical and optical goods; watches and clocks	588	(8.3)	413	(8.4)	76	(9.1)	99	(7.4)
73	Business services	1,574	(22.1)	1,042	(21.1)	295	(35.2)	237	(17.7)
Total		7,112	100	4,936	100	837	100	1,339	100



APPENDIX B

T-TEST

Table B.1 presents the results of the assessment of the mean differences on firm performance measures between these two groups. The USA data set indicates that firm performance, as measured by the mean values for Tobin's q and market to book (assets), was better for firms which had a SME as a member of the TMT for the entire period of interest than the firms which did not during the same period. The UK data indicates a marginal positive firm performance effect of having a SME in the top management on return on sales. The Germany data actually indicates lower firm performance on market to book (assets), sales growth and return on assets for firms with a SME as a member of the TMT.

The distribution of both groups across two-digit industry segments were similar, but deviated substantially in segments 28, 36 and 73 (see Table B.2). Part of the performance differences seen in the mean variance analysis may be due to differences between the two groups in their distribution across industry segments (McGahan and Porter, 1997).



	SME No SME						
	mean	sd	mean	sd	T-test	df	sig
USA							
Tobin's q	0.72	0.06	0.21	0.02	-7.61	734	***
Market to Book(assets)	0.65	0.06	0.22	0.02	-6.54	727	***
Market to Book(equity)	1.02	0.14	1.19	1.15	0.15	1710	ns
Sales Growth	0.77	0.53	0.42	0.21	-0.61	680	ns
Return on Assets	-0.00	0.01	-0.01	0.01	-0.49	1695	ns
Return on Sales	0.45	0.05	0.13	0.29	-1.09	595	ns
UK							
Tobin's q	1.26	0.66	1.85	0.46	0.73	139	ns
Market to Book(assets)	1.29	0.66	1.87	0.46	0.71	140	ns
Market to Book(equity)	2.34	0.83	5.67	1.88	1.62	464	ns
Sales Growth	0.04	0.03	0.04	0.02	-0.03	96	ns
Return on Assets	0.02	0.01	0.01	0.01	-0.79	129	ns
Return on Sales	0.03	0.01	0.01	0.01	-1.92	103	*
Germany							
Tobin's q	0.27	0.07	0.44	0.12	1.23	329	ns
Market to Book(assets)	0.39	0.12	9.29	2.29	3.88	642	***
Market to Book(equity)	0.45	0.14	2.67	2.54	0.87	643	ns
Sales Growth	0.02	0.01	0.05	0.01	1.67	680	*
Return on Assets	0.00	0.01	0.53	0.18	2.89	614	***
Return on Sales	0.01	0.01	-0.00	0.01	-1.00	616	ns

Table B.1: T-test of firm performance measures

*p<.1, **p<.05, ***p<.01, ns= not significant



www.manaraa.com

SIC 2	SME pro	esent	SME not present		
Segment	Firms %		Number	%	
25	3	3.9	4	1.6	
26	4	5.2	12	4.7	
28	9	11.7	57	22.4	
30	4	5.2	7	2.7	
33	2	2.6	16	6.3	
34	3	3.9	17	6.7	
35	9	11.7	40	15.7	
36	17	22.1	31	12.2	
38	7	9.1	22	8.6	
73	19	24.7	48	18.8	
Total	77	100	255	100	

Table B.2: Industry segment distribution of T-test data



www.manaraa.com

APPENDIX C MISSING DATA ANALYSIS

Missing data can pose problems for statistical inference (Rubin, 1988). The missing completely at random (MCAR) assumption is typically not achievable, and not necessary for efficient estimates. However, missing at random (MAR) data patterns are assumed for most estimation procedures, including GEE and GLS procedures.

The data set used in this research was complete for over two thirds of the observations. Table C.1 lists the number of missing values for each of the variables used in the models and the percentage of the total observations which were missing.

The MCAR assumption was tested using the correlation technique (Hair, 1998). Dummy dichotomous variables were created for each variable in the model with more than 1% of their data missing. Then a value of 0 was assigned for each missing value, and 1 for each present value. Table C.2 presents the resulting correlation table. As an example, it can be seen that there is very little correlation between the missing information in the variable MTB(equity) and *Internationalization* (r = 0.02), but it is high between *ROA* and *ROS* (r = 0.88). This indicates that the missing data for both *ROA* and *ROS* were related to the data sources used. A high missing data correlation between a covariate and a response variable was seen between *Turbulence* and *Sales Growth* (r = 0.97). Other high correlations were not of concern since they were between response



146

variables or with controls. This technique is coarse, so a more formal missing

data test was also conducted.

Table C.	1: Miss	ing data	per	variable
----------	---------	----------	-----	----------

	Variable	#Missing	%Missing
	Dependent Variables		
1	Tobin's q	91	1.3
2	MTB(equity)	395	5.6
3	MTB(assets)	399	5.6
4	Sales growth	777	10.9
5	ROA	119	1.7
6	ROS	112	1.6
7	SME presence	0	0
	Institutional Variables		
8	Isomorphic pressure	0	0
9	Marketing acceptance	0	0
10	Unitary governance	0	0
	Structural Variables		
11	Industry turbulence	752	10.6
12	Market internationalization	336	4.7
	Strategic Variables		
13	Corporate brand	16	0.2
14	House of brands	16	0.2
15	Business to business	32	0.4
16	Business to customer	32	0.4
17	Service product	0	0
	Control Variables		
18	Firm size	240	3.4
19	R&D intensity	1361	19.1
20	CEO tenure	219	3.1
21	COO presence	176	2.5
22	CEO change	180	2.5
23	Year	0	0



Little's MCAR Chi-square test was used to formally test whether the missing data patterns within the data set might be an issue to the analytic methods used, and to assess the degree of relatedness of missing data between the covariate and dependent variables (Li, 2013). The test gave strong evidence that the pattern of missing data in the response variables were not MCAR (X^2 distance=4518, p<0.00). However, the covariate dependent missing data (CDM) pattern between the covariate and response variables indicated that the dependence was not significant (X^2 =775, p<0.07). So, although the data is not MCAR, it does pass the CDM test at the 0.05 level. A moderate CDM level supports the veracity of estimation results achieved with both GEE and GLS methods.



Table C.2:	Missing	data	corre	lation
------------	---------	------	-------	--------

	%												
Variable	Missing	1	2	3	4	5	6	7	8	9	10	11	12
1 Tobin's q	1.3	1											
2 MTB(equity)	5.6	0.14	1.00										
3 MTB(assets)	5.6	-0.16	-0.99	1.00									
4 Sales growth	10.9	0.00	-0.09	0.09	1.00								
5 ROA	1.7	-0.24	-0.17	0.20	0.11	1.00							
6 ROS	1.6	-0.11	-0.13	0.13	0.12	0.88	1.00						
7 Turbulence	10.6	0.00	-0.05	0.05	0.97	0.06	0.05	1.00					
8 Internationalization	4.7	0.01	0.02	-0.02	0.03	0.06	0.07	0.02	1.00				
9 Firm size	3.4	-0.01	-0.20	0.19	0.05	0.16	0.17	0.00	0.05	1.00			
10 R&D intensity	19.1	0.05	-0.11	0.11	0.05	0.16	0.19	0.03	0.12	0.10	1.00		
11 CEO tenure	3.1	0.01	-0.07	0.07	0.04	0.01	0.02	0.05	0.00	-0.02	0.01	1.00	
12 COO presence	2.5	0.00	-0.12	0.11	0.03	0.06	0.05	0.03	0.06	0.00	0.00	0.56	1.00
13 CEO change	2.5	0.00	-0.09	0.09	0.03	0.03	0.04	0.02	0.01	0.05	0.03	0.56	0.63

المنسارات

APPENDIX D

NATH & MAHAJAN DATA AND REPLICATE DATA SETS

Following Nath and Mahajan (2008), a replicate data set was constructed in order to retest their findings and to extend their original five-year time horizon. This replicate data set and analyses are presented here, along with comparisons and contrasts with their original published findings.

Replicated Data Set

In general, the replicated data shows strong similarities in the summary statistics, variable values and firm distribution across industry segments when compared with the original information published by Nath and Mahajan (2008). But, they are not identical.

Table D.1 presents a comparison of the two data sets based on the distribution of firms in the data across industry segments, categorized using the two-digit Standard Industrial Code format. The dispersion of firms across the SIC segments in the respective data sets is between 0.3-3.0 percentage points of each other. Further, there are 167 individual firms in the base year (2002) in the Nath and Mahajan (2008) data set and 166 firms in the same year of the replicate data set. This indicates a very similar composition and distribution across industries for the two data sets.



		Numbe	r of firms	Percer	ntage (%)
SIC					
industry		N&M	Replicate	N&M	Replicate
code	Description of industry segment	data	data	data	data
25	Furniture and fixtures	4	5	2.4	3
26	Paper and allied products	3	3	1.8	1.8
28	Chemicals and allied products	21	27	12.6	16.3
30	Rubber and miscellaneous plastics products	7	3	4.2	1.8
33	Primary metal industries	3	2	1.8	1.2
	Fabricated metal products, except machinery and				
34	transportation equipment	6	5	3.6	3
35	Industrial and commercial machinery and computer equipment	37	34	22.1	20.5
	Electronic and other electrical equipment and components,				
36	except computer equipment	29	28	17.4	16.9
	Measuring, analyzing, and controlling instruments;				
38	photographic, medical and optical goods; watches and clocks	19	16	11.4	9.6
73	Business services	38	43	22.7	25.9
Total		167	166	100	100

 Table D.1: Comparison of Nath & Mahajan (2008) data and replicate data across industry segments

Tables D.2 and D.3 present the descriptive statistics and correlations of both the replicate data and the published information for Nath and Mahajan's (2008) data respectively. The mean values, standard deviations and correlations of the replicate data are very similar to those of the published values.¹³ None of the correlations in the replicate data exceeded 0.5 and the variance inflation factors for the indicator and control variables were low. The VIF average was 1.18 for the replicate set of variables and none of the individual VIF values exceeded 4. This result matches the reported results of Nath and Mahajan (2008).

¹³ It is important to note here that the variables "Total diversification", "TMT marketing experience" and "TMT general management experience" that were present in Nath and Mahajan's (2008) article were not replicated because they were not statistically significant and were not of theoretical interest for the additional comparative analyses done in this research. Therefore, they don't appear in any of the tables or figures.



Variable 9 Mean s.d. 3 5 8 10 11 12 6 1 CMO presence 0.35 0.48 1.00 2 Innovation_{t-1}^a 0.08 0.11*** 1.00 0.02 3 Differentiation_{t-1}^a 0.01 0.04 0.06 -0.05 1.00 4 Corportate branding^b 0.49 0.50 0.14*** 0.31*** -0.17*** 1.00 5 Outsider CEOt 0.12 0.39 0.01 -0.03 0.19*** 0.03 1.00 6 Market Concentration_{t-1}^c 0.11*** -0.16*** -0.15*** 0.16*** 0.01 0.09 0.05 1.00 7 Log(number of employees)_t 8.68 1.35 -0.01 -0.08 0.02 -0.05 -0.17*** -0.12*** 1.00 8 Year 2002.1 1.38 0.04 -0.02 -0.03 0.02 -0.01 0.05 0.00 1.00 9 CEO tenuret 6.71 7.52 0.07 -0.07 -0.11** -0.06 -0.19*** -0.10** -0.11*** -0.03 1.00 10 COO presence_t 0.31 0.46 0.04 -0.02 -0.05 -0.01 0.00 -0.04 0.06 -0.02 0.19*** 1.00 11 Prior performance_{t-1}^f 0.21*** -0.04 -0.04 0.24 0.01 -0.04 0.01 0.03 0.04 -0.09** 0.02 0.05 1.00 12 Business type ratio^b 0.63 0.48 0.00 -0.20*** 0.23*** -0.36*** 0.07 0.04 0.20*** -0.02 0.04 0.03 -0.05 1.00 0.02 0.11*** -0.08** 0.08** 13 Tobin's q 2.10 1.69 0.20 0.08 0.04 -0.06 0.03 0.10*** 0.20*** 0.01

Table D.2: Descriptive statistics and Pearson correlation coefficients for replicate data set

Notes: Data set constructed following N&M (2008) instructions; pooled across 168 firms over four years, 770 firm years.

*p<.1,**p<.05, ***p<.01

a = ratios of R&D(innovation) and advertising (differentiation) to revenue, less industry median ratio.

b= assumed to be constant over period.

c=Herfindahl-Hirschman industry concentration index (HHI).

d= sales growth, less industry median values.



Variable Mean s.d. 2 3 4 5 6 7 8 9 10 11 12 1 1 CMO presence 0.42 0.50 1.00 2 Innovation_{t-1}^a -0.03 0.09 0.20*** 1.00 3 Differentiation_{t-1}^a 0.01 0.03 0.16*** -0.05 1.00 4 Corportate branding^b 0.50 0.26*** 0.31*** -0.09** 1.00 0.53 5 Outsider CEO_t 0.32 0.47 0.14*** 0.01 0.12*** 0.05 1.00 6 Market Concentration_{t-1}^c 0.04 0.02 0.00 0.20*** 0.03 -0.06 0.15*** 1.00 7 Log(number of employees)_t 8.65 1.37 -0.06 -0.15*** 0.07* -0.32*** -0.05 0.08** 1.00 8 Year 2002.50 1.12 -0.02 0.08* 0.04 0.00 0.00 0.11** 0.01 1.00 -0.08** 9 CEO tenure_t 6.72 7.71 -0.07* -0.12*** 0.11*** -0.05 -0.08** -0.10** 0.02 1.00 10 COO presence_t 0.45 -0.08** -0.09** 0.08* 0.29 -0.12** 0.05 -0.03 0.13*** 1.00 0.06 0.03 11 Prior performance_{t-1}^t 0.06 0.35 0.04 0.04 0.12** -0.07** 0.00 -0.05 -0.08** -0.13*** -0.02 0.07* 1.00 12 Business type ratio^b -0.07* -0.24*** 0.43*** -0.42*** 0.08** 0.26*** 0.00 0.43 0.49 0.06 -0.18*** -0.18*** 0.02 1.00

Table D.3: Descriptive statistics and Pearson correlation coefficients for Nath and Mahajan (2008) data set

Notes:Nath and Mahajan (2008) published data; pooled across 167 firms over four years, 668 firm years.

*p<.1,**p<.05, ***p<.01

a = ratios of R&D(innovation) and advertising (differentiation) to revenue, less industry median ratio.

b= assumed to be constant over period.

c=Herfindahl-Hirschman industry concentration index (HHI).

d= sales growth, less industry median values.

Another important sample characteristic to note is the percentage of firms with CMOs in the data. Nath and Mahajan (2008) reported that the average incidence of firms with a senior marketing executive in their data was 41.4%, and for marketing executives having the CMO title it was 19.6%.¹⁴ The replicate data set presented an average incidence for senior marketing executives of 37.8%, and 10.7% for executives with the CMO title (Figure D.1). In the replicate data set, the percentage of senior marketing executives with the CMO title increased over the time frame of interest, but senior marketing executive positions in aggregate actually decreased slightly. The identification of a CMO and TMT members, although defined in both prior work and in this research, is open to some



Figure D.1: The rate of the presence of a senior marketing executive in the top management team (Nath and Mahajan use the CMO identification) comparison between the replicate data set and Nath and Mahajan's (2008) published results

¹⁴ Nath and Mahajan (2008) use the label "CMO" to mean a senior marketing executive in the top management team. However, they also acknowledge that there is a subgroup of senior marketing executives which carry the "Chief Marketing Officer" title. However, their analyses use "CMO" is the analyses and discussion to mean "SME," and is followed here for continuity.



interpretation when the researcher is attempting to identify these constructs in financial and company reports. This might be partial explanation for the difference.

Because the replicated data set is comparable in overall composition to the original data set described by Nath and Mahajan (2008), it was concluded that the replicate data set can reasonably be used for analytic comparisons. Table D.4 describes the variables, as defined by Nath and Mahajan (2008), which were used for the comparative analyses. The data were analyzed using the generalized estimating equation (GEE) approach (Zeger and Liang, 1986) following Nath and Mahajan (2008).

Variable	Definition
CMO presence	A marketing executive in the top management team as
	identified by a firm in its annual filings with the Securities
	and Exchange Commision.
Innovation	The ratio of R&D expenditure to sales.
Differentiation	The ratio of advertising expenditures to sales.
Corporate branding	The corporate brand is dominant in endorsement of firm
	products or services (Agarwal and Dahlhoff, 2004).
Outsider CEO	A newly appointed CEO with less than one year with the
	firm.
Market concentration	The Herfindahl-Hirschmann index at the two digit SIC
	level.

Table D.4: Explanation of the Nath and Mahajan (2008) variables used

In general, the results for the antecedent model using the replicated data set (USA, years 2000-2004) did not support prior published findings (see Table D.5). Nath and Mahajan (2008) had predicted that the presence of a CMO in the TMT would be positively associated with higher levels of R&D expenditures, advertising expenditures, a



corporate branding strategy, a CEO from outside the firm, and would be negatively associated with a highly concentrated market segment. They found support for all of these hypotheses, except in the case of high market concentration. The replicated data only supported the hypothesized positive association between CMO presence in the TMT and higher R&D expenditures.

Table D.5: H	ypotheses from	Nath and	Mahajan	(2008) w	ith predic	ted and	actual
results compa	ared to the actua	al results o	of the repl	icated da	ata set		

			Results			
Hypothesis	Description	Predicted	N&M (2008)	Replication		
Hypothesis 1	A firm's level of innovation is positively related to the	Positive	Yes	Yes		
	likelihood of CMO presence in its TMT.					
Hypothesis 2	A firm's level of differentiation is positively related to	Positive	Yes	No		
	the likelihood of CMO presence in its TMT.					
Hypothesis 3	The likelihood of CMO presence in the TMT is higher	Positive	Yes	No		
	in firms that have a corporate branding strategy than in					
	firms that pursue other branding strategies.					
Hypothesis 4	Not included in tested model					
Hypothesis 5	Not included in tested model					
Hypothesis 6	Not included in tested model					
Hypothesis 7	The likelihood of CMO presence in the TMT is higher	Positive	Yes	No		
	in firms with an outsider CEO than in firms with an					
	insider CEO.					
Hypothesis 8	The degreee of market concentration in a firm's	Negative	No	No		
	primary industry is negatively related to the likelihood					
	of CMO presence in its TMT.					
Hypotheses 9a-d	Firm performance is improved by CMO presence in					
	the TMT for firms that:					
	a: have relatively high levels of innovation.	Positive	No	No		
	b: have relatively high levels of differentiation.	Positive	No	No		
	c: have a corporate branding strategy.	Positive	No	No		
	d: have an outsider CEO.	Positive	No	No		
	e: are in industries that are relatively less concentrated.	Positive	No	No		

Table D.6, presents the detailed results of the logistic generalized estimating equation analyses of the antecedent model over the same time frame (2000-2004) and compares them to the results published by Nath and Mahajan (2008). As mentioned,



Nath and Mahajan (2008) found significant support for their hypothesized positive

relationships between the presence of a CMO in the TMT with the variables Innovation,

							Supp Hypo	ort for theses
	Model	1 ^a	Model	2^{b}	Model 3	3 ^c	A†	B†
Constant	66.18	125.1	33.46	126.87	-1.57 ***	0.43	-	-
Innovation	3.18 ***	1.03	3.64 ***	0.99	3.21 ***	1.02	Yes	Yes
Differentiation	1.12	2.75	3.05	2.87	1.39	2.79	No	Yes
Corporate branding	0.25	0.30	0.16	0.31	0.26	0.30	No	Yes
Outsider CEO	0.02	0.15	0.01	0.15	0.02	0.15	No	Yes
Market Concentration	3.92 *	2.18	-	-	3.84 *	2.18	No	No
Firm size	0.04	0.10	0.1	0.11	0.04	0.10	-	-
Year	-0.03	0.06	-0.02	0.06	-	-	-	-
CEO tenure	0.03 *	0.01	0.03 **	0.01	0.03 *	0.01	-	-
COO presence	0.11	0.17	0.08	0.1	0.18	0.16	-	-
Prior performance	-0.49 *	0.29	-0.61 **	0.29	-0.55 *	0.28	-	-
Customer ratio	0.13	0.33	0.23	0.34	0.13	0.33	-	-
SIC2	-	-	included		-	-	-	-
Wald X^2	19.7 **		19.37 **		19.02 **			

Table D.6: Logistic regression comparison with CMO as the dependent variable using replicated data from 2000-2004

*p<.1, **p<.05, ***p<.01

a= pooled logistic GEE regression over 2000-2004 period; N=636.

b= same as Model 1, but including industry segment and without market concentration. N=636 c= same as Model 1, but without year. N=636.

†A= replicated data set; B= reported in Nath & Mahajan (2008)

Differentiation, Corporate Branding and *Outsider CEO*, but not for their hypothesized negative relationship with *Market Concentration* (see †B). Models 1, 2, and 3 represent the same modeling presented by Nath and Mahajan (2008), but contrary to the prior published results, only the hypothesized positive relationship between *Innovation* and the presence of a CMO was supported across the three models. The relationship between *Market Concentration* and the presence of a CMO was actually positive and significant,



contrary to their prediction, in the replicate data (p<.1). None of the other relationships hypothesized by Nath and Mahajan (2008) were significantly supported in the replicated data. Also, it is interesting to note that the control variable, *Prior Performance*, was negatively and significantly related to the presence of a CMO in the replicate data. This was not the case in the original data and might indicate that firms that are experiencing difficulty with their revenues are more apt to choose to have a senior marketing executive in the TMT as a response to these difficulties. As discussed earlier, the differences in the results might also be because the replicate data are not identical to the data analyzed by Nath and Mahajan (2008). However, given the large effect sizes and high level of significance previously reported, it was expected that these relationships were robust enough to overcome these small differences.

In order to test the generalizability of Nath and Mahajan's (2008) hypotheses, the same analyses were applied to two additional time periods, first, the nine-year period from 2000 to 2007 in which economic and market conditions were similar to the period from 2000 through 2004, and second, the recessionary/post-recessionary three-year period from 2008 to 2010 in which the economic and market conditions were arguably very different.¹⁵

Table D.7 presents the results from 2000-2007. Here the antecedent relationship between the variable *Innovation* and the presence of a CMO in the TMT remains, as it is

¹⁵ The National Bureau of Economic Research (NBER) identified the beginning of the recession in December of 2007 and lasting through June of 2009. For simplicity, this research identifies the recession time frame as starting in January, 2008 and continuing through January, 2009 in the USA, with a recovery period extending through 2010. The exact recession beginning and end dates identified by the NBER for the USA, by Office for National Statistics for the UK, and by the Statistisches Bundesamt for Germany, are slightly different, but fall reasonably within the 2008 through 2009 period. All three economies were in a recovery phase by 2010.



in the 2000-2004 period, positive and significant, supporting Nath and Mahajan's (2008) results. However, the variable *Differentiation*, which previously did not indicate a significant association, now indicates partial support for an antecedent relationship to the

							Suppo	rt for
							Hypot	heses
	Model	1 ^a	Model	2 ^b	Model	3 ^c	A†	B†
	coef	se	coef	se	coef	se		
Constant	114.94	77.85	104.08	79.02	-1.28 ***	0.36	-	-
Innovation	2.71 ***	0.86	2.20 ***	0.80	2.64 ***	0.86	Yes	Yes
Differentiation	3.10	2.27	4.63 *	2.57	3.50	2.24	Partial	Yes
Corporate branding	0.39	0.26	0.30	0.25	0.40	0.26	No	Yes
Outsider CEO	-0.04	0.14	-0.04	0.15	-0.04	0.14	No	Yes
Firm size	0.03	0.08	0.08	0.09	0.02	0.09	-	-
Year	-0.06	0.04	-0.05	0.04	-	-	-	-
CEO tenure	0.02	0.01	0.01	0.01	0.02	0.01	-	-
COO presence	-0.09	0.13	-0.09	0.14	-0.08	0.14	-	-
Prior performance	-0.05	0.16	0.04	0.16	-0.01	0.15	-	-
Customer ratio	-0.02	0.28	0.01	0.28	-0.02	0.28	-	-
SIC2	-	-	included		-	-	-	-
Wald X^2	20.96		29.37		18.87			

Table D.7: Logistic regression comparison with CMO as the dependent variable using replicated data from 2000-2007

*p<.1, **p<.05, ***p<.01

a= pooled logistic GEE regression over 2000-2007 period. N=1071.

b= same as Model 1, but including industry segment. N=1071

c = same as Model 1, but without year. N = 1071.

†A= replicated data set; B= reported in Nath & Mahajan (2008)

presence of a CMO. When selecting only the recessionary years (2008-2010), none of the hypothesized relationships are significant (Table D.8). When looking at the entire data time frame from 2000 through 2010, we see (Table D.9) that none of the hypothesized relationships tested are significant, except for a rather weakly significant association with *Corporate branding* that appears (the previous positive and significant association with



Innovation disappears). Leaving the impact of the economically turbulent three years

from 2008 through 2010 aside, the results from 2000 onward are stable. They show that

							Supp Hypo	ort for theses
	Mode	el 1 ^a	Mode	el 2 ^b	Model	3 ^c	A†	B†
	coef	se	coef	se	coef	se		
Constant	14.86	174.95	-46.38	185.38	-1.15	0.52	-	-
Innovation	0.02	0.03	0.01	0.04	0.02	0.03	No	Yes
Differentiation	-0.13	0.13	0.19	0.15	0.13	0.13	No	Yes
Corporate branding	0.27	0.37	0.20	0.41	0.28	0.37	No	Yes
Outsider CEO	0.39	0.36	0.42	0.38	0.38	0.36	No	Yes
Firm size	0.13	0.14	0.11	0.15	0.11	0.14	-	-
Year	-0.01	0.08	0.02	0.09	-	-	-	-
CEO tenure	0.01	0.02	0.01	0.02	0.01	0.02	-	-
COO presence	0.36	0.30	0.44	0.28	0.35	0.26	-	-
Prior performance	-0.02 *	0.01	-0.03 *	0.02	-0.02 *	0.01	-	-
Customer ratio	-0.26	0.41	-0.37	0.44	-0.26	0.42	-	-
SIC2	-	-	included		-	-	-	-
Wald X^2	10.17		13.79		10.17			

 Table D.8: Logistic regression comparison with CMO as the dependent variable

 from 2008-2010

*p<.1, **p<.05, ***p<.01

a= pooled logistic GEE regression over 2008-2010. N= 350.

b= same as Model 1, but controlling for industry effects at two digit SIC level. N= 350.

c = same as Model 1, but without year. N = 350.

† A= USA data 2008-2010; B= reported in Nath & Mahajan (2008)

only research and development expenditure is a significant predictor for the presence of a

CMO in the TMT. This result contrasts with the prior published results.

Nath and Mahajan (2008) also looked at the possible impact of the presence of a

CMO on firm performance, as measured by Tobin's q and sales growth. They

hypothesized that firm performance is improved by the presence of a CMO in firms with

1) relatively high levels of innovation, 2) differentiation, 3) an outsider CEO, 4) a

corporate branding strategy, and 5) in industry segments with lower market



concentration. They reported no main effects and no significant interaction effects

between the presence of a CMO and firm performance (Table D.10).

							Suppo Hypot	ort for heses
	Mode	11 ^a	Mode	l 2 ^b	Model 3	3 ^c	A†	B†
	coef	se	coef	se	coef	se		
Constant	102.68 *	57.65	93.21 *	57.38	-1.15 ***	0.32	-	-
Innovation	0.02	0.03	0.02	0.03	0.02	0.03	No	Yes
Differentiation	-0.04	0.09	0.06	0.13	-0.06	0.09	No	Yes
Corporate branding	0.40 *	0.23	0.34	0.24	0.41 *	0.23	Partial	Yes
Outsider CEO	0.07	0.13	0.01	0.13	0.07	0.13	No	Yes
Firm size	0.01	0.08	0.04	0.08	-0.01	0.08	-	-
Year	-0.05 *	0.03	-0.05 *	0.03	-	-	-	-
CEO tenure	0.01	0.01	0.01	0.01	0.01	0.01	-	-
COO presence	-0.04	0.12	-0.09	0.12	-0.03	0.12	-	-
Prior performance	-0.01	0.01	-0.02	0.02	-0.01	0.01	-	-
Customer ratio	-0.10	0.27	-0.05	0.26	-0.08	0.26	-	-
SIC2	-	-	included		-	-	-	-
Wald X^2	13.82		22.48		11.1			

 Table D.9: Logistic regression comparison with CMO as the dependent variable

 from 2000-2010

*p<.1, **p<.05, ***p<.01

a= pooled logistic GEE regression over 2000-2010. N= 1348.

b= same as Model 1, but controlling for industry effects at two digit SIC level. N= 1348.

c = same as Model 1, but without year. N = 1348.

† A= USA data 2000-2010; B= reported in Nath & Mahajan (2008)

The same replicate data set was used to assess the firm performance analyses done by Nath and Mahajan (2008) in the same way as was completed for the hypothesized antecedents. Table D.11 presents a summary of Nath and Mahajan's (2008) hypothesized and reported direct and moderated effects of the presence of a CMO on firm performance (tobin's q and sales growth). Table D.12 presents the GLS regression results for the dependent variables *Tobin's q* and *Sales Growth* for the USA data over the 2000-2004, time frame. No main effect was found for the presence of a CMO on either



Tobin's q or *Sales Growth*. Each variable hypothesized to have an association with the presence of a CMO from the antecedent model was introduced separately as an interaction term for each firm performance variable. Only one interaction was significant; the interaction

 Table D.10: Summary of hypothesized direct and interaction effects of the presence of a CMO on firm performance and actual results

					Dependent	Variable
			†Predicted	†Reported	Tobin's q	Sales Growth
Direct effect		CMO presence		no effect	no effect	no effect
Interactions	a.	Innovation x CMO	positive	no effect	negative ***	no effect
	b.	Differentiation x CMO	positive	no effect	no effect	no effect
	c.	Corporate branding x CMO	positive	no effect	no effect	no effect
	d.	Outsider CEO x CMO	positive	no effect	no effect	no effect
	e.	Market concentration x CMO	positive	no effect	no effect	no effect

*p<.1, **p<.05, ***p<.01

†Nath & Mahajan (2008)

between the presence of a CMO and *Innovation* with *Tobin's q* (Table D.11) was significant and negative, rather than positive, as predicted. The replicated data set results confirmed the prior published results of Nath and Mahajan (2008). Support was not found for the hypotheses that there is a positive and significant relationship between firm performance and the interaction of a CMO and, 1) innovation, 2) differentiation, 3) corporate branding, 4) an outsider CEO, and 5) a concentrated market segment.



Table D.11: GLS regression results with firm performance as the dependent variable 2000-2004

	De	pender	t Variable		De	pender	nt Variable			
	Tobin's	q	Tobin's	q	Sales Gro	wth	Sales Gr	owth	Suppo	ort for
	Model	1 ^a	Model	1 ^b	Model	2^{a}	Model	2 ^b	Hypot	theses
	coef	se	coef	se	coef	se	coef	se	A†	B^{\dagger}
Constant	0.57 ***	0.14	0.62 ***	0.14	0.04	0.03	0.03	0.03	-	-
Innovation	1.47 ***	0.49	2.46 ***	0.67	0.39 ***	0.13	0.29 ***	• 0.11	-	-
Differentiation	2.45 ***	0.95	2.42 ***	0.94	0.34	0.24	0.36	0.24	-	-
Corporate branding	-0.09	0.08	-0.10	0.08	-0.03	0.02	-0.03	0.02	-	-
Outsider CEO	0.12	0.14	0.12	0.14	0.01	0.02	0.01	0.02	-	-
Market Concentration	0.59	0.67	0.44	0.67	0.14	0.15	0.15	0.15	-	-
Firm size	0.02	0.04	0.02	0.03	0.01	0.01	0.01	0.01	-	-
CEO tenure	0.01	0.01	0.01	0.01	0.01 *	0.01	0.00 *	0.00	-	-
COO presence	0.01	0.08	0.01	0.08	0.02	0.02	0.02	0.02	-	-
Prior performance	0.53 ***	0.05	0.53 ***	0.05	0.27 ***	0.06	0.27 ***	0.06	-	-
ROA	0.37	0.23	0.36	0.23	-0.15	0.05	-0.04	0.05	-	-
Sales growth	1.34 ***	0.29	1.34 ***	0.29	-	-	-	-	-	-
CMO presence	-0.04	0.08	-0.02	0.08	-0.02	0.02	-0.02	0.02	No	No
Innovation x CMO	-	-	-2.27 ***	0.82	-	-	0.21	0.21	No	No

*p<.1, **p<.05, ***p<.01

1a= with CMO direct effect; N=757, groups=166, Wald $X^2(12)$ =495.73, Prob > X^2 =.0001

1b= with direct and interaction term; N=653, groups=166, Wald $X^2(12)=511.68$, Prob > $X^2=.0001$

2a= with CMO direct effect, N=505, groups=166, Wald $X^{2}(11)=47.67$, Prob > $X^{2}=.0001$

2b= with dirct and interaction term; N=505, groups=166, Wald $X^{2}(11)=47.67$, Prob > $X^{2}=.0001$

† A= USA data 2000-2004; B= reported in Nath & Mahajan (2008)



APPENDIX E

COUNTRY LEVEL COMPARISONS

The country level firm performance comparisons for the 2000-2010 time period are presented here. The GLS results are presented first for the USA, Germany and UK, and then the GEE results. The summary support for the proposed hypotheses follow. Finally, the performance results for each country over the recessionary period from 2008-2010 are presented.



			Industry A	Adjusted		Industry Adjusted						
			Market Based	Performa	ance			Rev	enue Based Pe	rforma	nce	
	Tobin's	q	MTB(asse	ts) ^a	MTB(equ	ity) ^a	Sales Grov	vth ^a	ROA ^a		ROS	a
	coef	se	coef	se	coef	se	coef	se	coef	se	coef	se
Constant	0.25 **	0.12	0.18	0.16	2.68	5.02	-2.64	3.4	-0.05 *	0.03	-0.25	0.19
SME presence	0.04	0.06	0.04	0.11	2.83	2.22	-0.00	2.06	-0.05	0.04	-0.21	0.48
Institutional Interactions												
Isomorphic pressure x SME	0.19	0.15	0.13	0.14	-7.56 *	4.46	0.33	1.09	0.06	0.04	0.60	0.83
Marketing acceptance x SME	-0.03	0.04	-0.06	0.04	-2.47	1.52	-0.04	1.03	0.01	0.01	-0.16	0.27
Board membership x SME	-0.33 **	0.20	-0.37 **	0.14	0.26	1.53	-1.22	0.96	0.12 ***	0.04	0.22	0.28
Structural Interactions												
Industry turbulence x SME	-0.01	0.02	-0.02	0.02	0.71	0.49	-4.87	4.07	0.01	0.01	0.04	0.12
Internationalization x SME	-0.02	0.04	-0.02	0.04	2.52 **	1.20	0.47	0.85	-0.00	0.01	0.02	0.08
Strategic Interactions												
Corporate brand x SME	-0.10	0.11	0.13	0.11	3.36	3.21	-2.82	2.18	-0.02	0.02	0.07	0.28
House of brands x SME	-0.19	0.15	-0.04	0.16	4.98	3.47	-0.19	0.92	-0.02	0.04	0.06	0.24
Business to customer x SME	0.05	0.11	0.11	0.21	6.38	4.35	2.30	3.06	0.07	0.04	0.08	0.10
Business to business x SME	0.04	0.12	-0.06	0.11	1.01	2.85	3.81	2.69	0.04	0.02	0.19	0.10
Service product x SME	-0.15	0.14	-0.06	0.14	-7.52	5.63	1.49	1.32	0.01	0.03	-0.08	0.23
Year ^c	-0.02 **	0.01	-0.01	0.01	-0.18	0.23	0.36	0.24	0.01 **	0.00	0.02	0.02
Wald X^2	239.63 ***		225.10 ***		93.46 ***		94.52 ***		64.60 ***		225.8 ***	
R sq	0.44		0.51		0.01		0.028		0.01		0.162	
Observations	3296		3142		3140		2917		3292		3291	

Table E.1: GLS random effects analysis of SME impact on firm performance for the USA from 2000-2010

*p<.1, **p<.05, ***p<.01

a= Tobin's q, MTB(assets), MTB(equity), Sales Growth, ROA and ROS, R&D intensity, and Prior performance are centered at the country

and two digit SIC level to control for industry segment effects.

b=Prior performance is the lagged form of the dependent variable.

c= Year is controlled for as a dummy variable.



			Industry A	Adjusted					Industry Adj	usted		
			Market Based	Perform	ance			Rev	venue Based Pe	erforma	nce	
	Tobin's	q	MTB(asse	ts) ^a	MTB(e	equity) ^a	Sales Grov	vth ^a	ROA ^a		ROS	a
	coef	se	coef	se	coef	se	coef	se	coef	se	coef	se
Constant	-0.87	0.54	-2.10	10.50	-24.5	29.35	0.18	0.1	1.80	1.90	0.08	0.09
SME presence	0.76	0.85	3.47	5.64	-16.14	14.10	-0.25	0.28	-0.49	0.56	-0.12	0.08
Institutional Interactions												
Isomorphic pressure x SME	0.47	0.32	-7.47	7.64	-7.27	7.81	0.06	0.08	-0.09	0.24	-0.02	0.05
Marketing acceptance x SME	0.25	0.23	-1.6	4.30	-2.90	2.74	-0.06	0.08	-0.11	0.14	-0.02	0.03
Board membership x SME	-0.06	0.18	-0.31	0.36	0.05	1.72	0.04	0.03	-0.09	0.11	0.03	0.02
Structural Interactions												
Industry turbulence x SME	0.08 *	0.05	-0.32	0.36	5.81	4.88	0.00	0.01	-0.04	0.05	-0.00	0.00
Internationalization x SME	-0.03	0.07	-1.98	1.42	-1.15	1.69	0.02 *	0.01	-0.30	0.32	0.00	0.01
Strategic Interactions												
Corporate brand x SME	-0.15	0.23	-4.47	3.30	-6.30	5.39	-0.05	0.04	0.63	0.62	-0.02	0.03
House of brands x SME	0.29	0.28	1.24	1.74	9.20	8.02	0.68 ***	0.14	0.92	0.99	-0.02	0.03
Business to customer x SME	0.15	0.36	5.51	4.20	17.46	15.64	0.03	0.04	0.73	0.85	0.03	0.03
Business to business x SME	-0.24	0.28	4.03	2.92	14.72	13.59	0.01	0.05	-0.86	0.89	0.02	0.02
Service product x SME	0.44	0.33	3.02	2.44	-0.21	3.34	0.06	0.05	0.36	0.39	0.07 **	0.03
Year ^c	0.03 **	0.01	0.25	0.32	1.97	1.99	-0.01	0.00	-0.04	0.04	-0.00	0.00
Wald X^2	244.9 ***		6.2x10 ⁵ ***		19.41		184.63 ***		1.9x10 ⁶ ***		172.9 ***	
R sq	0.65		0.74		0.03		0.10		0.55		0.04	
Observations	806		848		848		782		840		841	

Table E.2: GLS random effects analysis of SME impact on firm performance for Germany from 2000-2010

*p<.1, **p<.05, ***p<.01

a= Tobin's q, MTB(assets), MTB(equity), Sales Growth, ROA and ROS, R&D intensity, and Prior performance are centered at the country

and two digit SIC level to control for industry segment effects.

b=Prior performance is the lagged form of the dependent variable.

c= Year is controlled for as a dummy variable.



		Industry A					Industry Adj	usted				
			Market Based	Perform	ance			Rev	enue Based Pe	erforma	nce	
	Tobin's	s q	MTB(asse	ts) ^a	MTB(equity) ^a	Sales Grov	vth ^a	ROA ^a		ROS	a
	coef	se	coef	se	coef	se	coef	se	coef	se	coef	se
Constant	-0.05	0.33	-0.24	0.51	2.79	41.06	-0.02	0.09	0.09	0.14	0.02	0.08
SME presence	0.36	0.66	0.27	0.52	-14.31	12.56	0.08	0.08	-0.06	0.09	0.16 **	0.07
Institutional Interactions												
Isomorphic pressure x SME	-0.32	0.69	-0.69	0.98	-8.42	8.71	-0.19	0.13	-0.16	0.14	-0.06	0.07
Marketing acceptance x SME	0.02	0.16	-0.03	0.19	4.30	4.67	-0.03	0.03	-0.02	0.03	-0.01	0.02
Structural Interactions												
Industry turbulence x SME	-0.03	0.09	-0.50	0.15	-0.85	1.85	0.02	0.01	0.01	0.03	-0.01	0.01
Internationalization x SME	-0.03	0.15	-0.08	0.27	7.17	5.37	0.12	0.04	-0.02	0.05	0.01	0.02
Strategic Interactions												
Corporate brand x SME	0.75 **	0.39	0.86 *	0.76	16.86	17.12	0.03	0.08	0.05	0.06	-0.14	0.13
House of brands x SME	0.39	0.46	0.22	0.41	13.35	11.87	0.17	0.09	0.05	0.05	-0.10	0.15
Business to customer x SME	-0.41	0.42	-0.40	0.56	16.06	13.52	-0.31	0.08	0.11	0.10	-0.01	0.06
Business to business x SME	-1.25 **	0.65	-0.92 *	0.69	2.20	21.20	-0.33	0.17	0.12	0.11	0.02	0.16
Service product x SME	0.26	0.48	-0.02	0.72	5.91	16.22	0.26	0.19	0.11	0.07	0.09	0.06
Year ^c	0.03 **	0.01	-0.03	0.03	0.84	1.51	0.01	0.01	0.00	0.00	0.00	0.00
Wald X^2	163.99 ^d ***		127.37 ^d ***		16.58 ^d		73.49 ^d ***		63.88 ^d ***		108.06 ^d ***	
R sq	0.58		0.30		0.06		0.15		0.45		0.29	
Observations	471		477		482		436		477		490	

Table E.3: GLS random effects analysis of SME impact on firm performance for the UK from 2000-2010

*p<.1, **p<.05, ***p<.01

Note: There were no observed instances of a SME being a board member, so the "Board membership x SME" variable was dropped from the UK analyses.

a= Tobin's q, MTB(assets), MTB(equity), Sales Growth, ROA and ROS, R&D intensity, and Prior performance are centered at the country

and two digit SIC level to control for industry segment effects.

b=Prior performance is the lagged form of the dependent variable.

c= Year is controlled for as a dummy variable.

d= Bootstrap of 50 replications.



			Industry Adju	isted					Industry Adju	isted		
		Ν	/larket Based Pe	rformanc	ce			Rev	venue Based Pe	rforman	ce	
	Tobin's q	[MTB(asset	s) ^a	MTB(equit	y) ^a	Sales Grow	rth ^a	ROA ^a		ROS	1
	coef	se	coef	se	coef	se	coef	se	coef	se	coef	se
Constant	0.78 ***	0.18	0.54 ***	0.14	2.83	4.95	-3.04	3.8	-0.06 **	0.02	-0.27	0.19
SME presence	0.27 *	0.14	0.21	0.13	2.57	2.13	-0.66	3.01	-0.01	0.02	-0.18	0.61
Institutional Interactions												
Isomorphic pressure x SME	0.08	0.13	0.05	0.13	-6.77 *	3.99	0.79	2.76	0.03	0.03	0.52	0.87
Marketing acceptance x SME	-0.13 ***	0.05	-0.14 ***	0.04	-2.35 *	1.40	-0.24	1.09	-0.00	0.01	-0.18	0.29
Board membership x SME	-0.39 *	0.21	-0.43 **	0.21	0.39	1.43	-0.54	0.96	0.04 **	0.02	0.25	0.27
Structural Interactions												
Industry turbulence x SME	0.00	0.02	-0.01	0.03	0.82	0.63	-4.67	4.01	0.00	0.01	0.04	0.11
Internationalization x SME	-0.06	0.04	-0.06	0.04	2.17 **	1.05	0.57	0.93	-0.01	0.01	0.03	0.09
Strategic Interactions												
Corporate brand x SME	-0.10	0.14	-0.13	0.12	3.42	2.99	-3.40	2.52	-0.00	0.02	-0.11	0.31
House of brands x SME	-0.33 **	0.14	-0.32 **	0.13	3.65	3.10	-0.03	1.46	-0.01	0.03	0.01	0.19
Business to customer x SME	-0.05	0.19	0.13	0.17	5.15	3.74	3.85	3.97	0.05	0.05	0.12	0.40
Business to business x SME	0.06	0.15	0.10	0.13	0.61	2.52	4.93	3.14	-0.02	0.02	0.22	0.42
Service product x SME	-0.20	0.21	-0.16	0.18	0.17	0.02	2.29	1.68	0.01	0.02	-0.02	0.26
Year ^c	-0.06 ***	0.01	-0.04 ***	0.01	-0.18	0.23	0.53	0.36	0.00	0.00	0.02	0.02
Wold V^2	226.02 ***		104 72 ***		2002 14 ***		$1.1_{\rm w}10^5$ ***		12/ / 2 ***		106 9 ***	
observations	230.03		2001		2002.14		2060		2009		490.0	
Observations	3232		3091		3085		2808		3228		3230	

Table E.4: GEE random effects analysis of SME impact on firm performance for the USA from 2000-2010

*p<.1, **p<.05, ***p<.01

a= Tobin's q, MTB(assets), MTB(equity), Sales Growth, ROA and ROS, R&D intensity, and Prior performance are centered at the country

and two digit SIC level to control for industry segment effects.

b=Prior performance is the lagged form of the dependent variable.

c= Year is controlled for as a dummy variable.



			Industry Adj	usted					Industry Adju	isted		
		М	arket Based Pe	rformance	2			Re	venue Based Pe	rformar	ice	
	Tobin's	q	MTB(asse	(s) ^a	MTB(e	quity) ^a	Sales Grow	/th ^a	ROA ^a		ROS ^a	1
	coef	se	coef	se	coef	se	coef	se	coef	se	coef	se
Constant	0.89	0.62	-1.98	11.88	-29.47	33.35	0.33 **	0.2	1.06	1.12	0.04	0.09
SME presence	0.29	1.08	-5.04	6.55	-17.93	15.02	-0.33	0.32	-0.37	0.44	-0.10	0.09
Institutional Interactions												
Isomorphic pressure x SME	0.73 *	0.42	-6.67	6.94	-9.42	9.32	0.07	0.09	-0.04	0.14	-0.03	0.04
Marketing acceptance x SME	0.06	0.28	-2.05	3.11	-3.56	3.16	-0.07	0.10	-0.05	0.08	-0.02	0.03
Board membership x SME			-0.07	0.48	0.50	1.95	0.03	0.04	-0.02	0.05	0.03	0.02
Structural Interactions												
Industry turbulence x SME	0.07	0.06	0.18	0.32	6.25	5.13	0.00	0.01	-0.02	0.03	-0.00	0.00
Internationalization x SME	0.00	0.09	-1.61	1.17	-1.44	1.35	0.03 **	0.01	-0.15	0.16	0.00	0.01
Strategic Interactions												
Corporate brand x SME	-0.13	0.27	-4.54	3.11	5.77	5.02	0.05	0.06	-0.33	0.34	0.03	0.02
House of brands x SME	0.69	0.33	1.20	1.64	8.71	7.62	0.93 **	0.02	0.13	0.23	0.00	0.02
Business to customer x SME	-0.44	0.48	8.68	5.77	18.08	16.37	0.01	0.08	0.79	0.83	0.00	0.01
Business to business x SME	-0.74	0.40	4.11	2.76	15.18	13.70	-0.04	0.06	0.42	0.46	-0.02	0.02
Service product x SME	0.48	0.57	3.42	2.81	-0.98	4.33	0.18	0.06	0.32	0.33	0.05 **	0.02
Year ^c	-0.02	0.02	0.29	0.34	2.21	2.16	-0.01	0.01	-0.02	0.02	-0.00	0.00
Wald X^2	26.66		8.4x10 ⁵ ***		19.05		219.96 ***		8.4x10 ⁶ ***		1219.8 ***	
Observations	750		790		790		714		781		782	

Table E.5: GEE random effects analysis of the SME impact on firm performance for Germany from 2000-2010

*p<.1, **p<.05, ***p<.01

a= Tobin's q, MTB(assets), MTB(equity), Sales Growth, ROA and ROS, R&D intensity, and Prior performance are centered at the country

and two digit SIC level to control for industry segment effects.

b=Prior performance is the lagged form of the dependent variable.

c= Year is controlled for as a dummy variable.



			Industry Adju	isted					Industry Adju	isted		
		Ν	larket Based Per	formance	2			Re	venue Based Pe	rforma	nce	
	Tobin's q		MTB(asset	s) ^a	MTB(equ	ity) ^a	Sales Grow	vth ^a	ROA ^a		ROS	ı
	coef	se	coef	se	coef	se	coef	se	coef	se	coef	se
Constant	0.48	0.93	-0.25	0.44	-3.40	12.17	0.06	0.06	0.06	0.04	0.04	0.03
SME presence	-0.12	0.74	0.48	0.51	-13.47	8.91	0.04	0.08	0.03	0.06	0.07	0.05
Institutional Interactions												
Isomorphic pressure x SME	-0.19	1.07	-0.67	0.67	-28.95 *	15.09	-0.23	0.14	0.07	0.08	-0.05	0.05
Marketing acceptance x SME	0.04	0.13	0.12	0.17	6.74	7.05	-0.04	0.03	0.02	0.02	0.01	0.02
Structural Interactions												
Industry turbulence x SME	-0.13	0.17	-0.09	0.10	-2.11	1.70	0.03	0.01	-0.02	0.01	-0.01	0.01
Internationalization x SME	0.05	0.57	-0.27	0.21	7.79	4.97	0.05	0.03	-0.03	0.02	-0.01	0.01
Strategic Interactions												
Corporate brand x SME	1.15 ***	0.25	1.33 ***	0.34	27.10	19.67	0.18	0.05	0.03	0.07	-0.01	0.02
House of brands x SME	1.50 ***	0.43	1.13 ***	0.48	40.34 **	20.38	0.12	0.12	-0.00	0.06	-0.02	0.02
Business to customer x SME	-0.92	0.29	-0.70 *	0.32	7.04	9.72	-0.23	0.08	0.02	0.05	0.00	0.02
Business to business x SME	-1.54	0.59	-1.27 **	0.66	-3.19	11.39	-0.16	0.07	-0.14	0.09	-0.05	0.03
Service product x SME	0.16	0.47	0.10	0.47	22.75	16.87	0.19	0.06	0.06	0.06	0.04	0.03
Year ^c	-0.33	0.21	0.01	0.02	1.60	2.31	0.02 **	0.01	0.00	0.00	0.00	0.01
Wald X^2	1.1x10 ^{5d} ***		4.7x10 ^{4d} ***		195.99		3.4x10 ^{3d} ***		5.3x10 ^{4d} ***		1.2x10 ^{4d} ***	
Observations	288		339		340		274		477		302	

Table E.6: GEE random effects analysis of the SME impact on firm performance for the UK from 2000-2010

*p<.1, **p<.05, ***p<.01

Note: There were no observed instances of a SME being a board member, so the "Board membership x SME" variable was dropped from the UK analyses.

a= Tobin's q, MTB(assets), MTB(equity), Sales Growth, ROA and ROS, R&D intensity, and Prior performance are centered at the country

and two digit SIC level to control for industry segment effects.

b=Prior performance is the lagged form of the dependent variable.

c= Year is controlled for as a dummy variable.

d= Bootstrap of 50 replications.



			Tobin's o	1		MTB(asse	ts)	I	MTB(equit	y)
Hypothesis	Predicted	USA	Germany	UK	USA	Germany	UK	USA	Germany	UK
Hypothesis 8	Positive	Yes (*)	No	No	No	No	No	No	No	Yes (**)
Hypothesis 9	Positive	No	Yes (*)	No	No	No	No	No	No	No
Hypothesis 10	Positive	No	No	No	No	No	No	No	No	No
Hypothesis 11	Positive	No	No	No	No	No	No	No	No	No
Hypothesis 12	Positive	No	No	No	No	No	No	No	No	No
Hypothesis 13	Positive	No	No	No	No	No	No	No	No	No
Hypothesis 14	Positive	No	No	Yes (***)	No	No	Yes (***)	No	No	No
Hypothesis 15	Positive	No	No	No	No	No	No	No	No	No
Hypothesis 16	Positive	No	No	No	No	No	No	No	No	No

Table E.7: Summary of findings of SME impact on market based firm performance metrics using GEE

*p<.1; **p<.05; ***p<.01

171

Note: all independent variables testing hypotheses were lagged one year.

Variables for hypotheses 3 & 4 were centered by two digit SIC code.

The year was controlled by using a dummy variable with 2000 as the reference year.
	_	Sales Growth				ROA		ROS			
Hypothesis	Predicted	USA	Germany	UK	USA	Germany	UK	USA	Germany	UK	
Hypothesis 8	Positive	No	No	No	No	No	No	No	No	No	
Hypothesis 9	Positive	No	No	No	No	No	No	No	No	No	
Hypothesis 10	Positive	No	No	No	No	No	No	No	No	No	
Hypothesis 11	Positive	No	No	No	Yes (**)	No	No	No	No	No	
Hypothesis 12	Positive	No	No	No	No	No	No	No	No	No	
Hypothesis 13	Positive	No	Yes (**)	No	No	No	No	No	No	No	
Hypothesis 14	Positive	No	No	No	No	No	No	No	No	No	
Hypothesis 15	Positive	No	No	No	No	No	No	No	No	No	
Hypothesis 16	Positive	No	No	No	No	No	No	No	Yes (**)	No	

Table E.8: Summary of findings of SME impact on accounting based firm performance metrics using GEE

*p<.1; **p<.05; ***p<.01

Note: all independent variables testing hypotheses were lagged one year.

Variables for hypotheses 3 & 4 were centered by two digit SIC code.

The year was controlled by using a dummy variable with 2000 as the reference year.



	Industry Adjusted							Industry Adjusted						
	Market Based Performance							Revenue Based Performance						
	Tobin's q		MTB(assets) ^a		MTB(equity) ^a		Sales Growth ^a		ROA ^a		ROS ^a			
	coef	se	coef	se	coef	se	coef	se	coef	se	coef	se		
Constant	-0.29	0.29	0.54	0.29	-39.77	30.47	9.26	22.84	-0.35	0.08	-3.27	3.98		
SME presence	0.09	0.09	0.90	0.10	2.39	1.88	-1.49	3.71	-0.01	0.04	-0.91	1.66		
Institutional Interactions														
Isomorphic pressure x SME	0.05	0.12	0.04	0.13	-5.71	4.38	2.41	2.69	-0.06	0.06	1.77	2.53		
Marketing acceptance x SME	-0.11	0.07	-0.11	0.07	-3.28 *	1.89	-1.49	2.04	-0.08 ***	0.02	0.26	0.94		
Board membership x SME	-0.30 **	0.15	-0.34 *	0.19	-0.28	1.83	-6.24	7.42	0.10 *	0.06	-0.37	2.21		
Structural Interactions														
Industry turbulence x SME	-0.07	0.05	-0.10 *	0.06	-1.02	1.37	-4.19	3.15	-0.01	0.02	-0.55	0.89		
Internationalization x SME	0.01	0.03	0.02	0.03	2.39	1.95	-0.79	1.12	0.01	0.01	0.19	0.34		
Strategic Interactions														
Corporate brand x SME	0.10	0.08	0.09	0.09	1.27	2.44	-3.11	4.79	0.01	0.04	0.35	1.13		
House of brands x SME	-0.11	0.16	-0.07	0.10	5.74	3.57	-0.86	2.62	0.02	0.06	-0.53	0.68		
Business to customer x SME	-0.00	0.19	0.05	0.20	0.06	3.56	2.49	6.46	0.06	0.09	-0.06	0.77		
Business to business x SME	-0.08	0.09	-0.08	0.10	2.29	2.57	4.91	4.61	0.00	0.04	0.38	1.16		
Service product x SME	-0.05	0.09	-0.04	0.10	-0.90	1.74	4.29	3.19	-0.04	0.04	-0.41	1.09		
Year ^c	0.02	0.03	-0.06	0.03	3.81	2.97	-0.76	2.01	0.03 ***	0.01	0.32	0.42		
Wald X^2	2501.7 ***		3271.40 ***		845.69 ***		21.94		82.94 ***		92.6 ***			
R sq	0.66		0.63		0.02		0.02		0.06		0.12			
Observations	835		804		804		830		835		833			

Table E.9: GLS random effects analysis of USA SME impact on firm performance 2008-2010

*p<.1, **p<.05, ***p<.01

Note: There were no observed instances of a SME being a board member, so the "Board membership x SME" variable was dropped from the UK analyses.

a= Tobin's q, MTB(assets), MTB(equity), Sales Growth, ROA and ROS, R&D intensity, and Prior performance are centered at the country

and two digit SIC level to control for industry segment effects.

b=Prior performance is the lagged form of the dependent variable.

c= Year is controlled for as a dummy variable.



			Industry Adju	isted		Industry Adjusted						
		Revenue Based Performance										
	Tobin's q		MTB(assets) ^a		MTB(equity) ^a		Sales Growth ^a		ROA ^a		ROS ^a	
	coef	se	coef	se	coef	se	coef	se	coef	se	coef	se
Constant	-0.34	0.29	0.58 *	0.30	-37.78	28.32	9.20	22.57	-0.04	0.10	-1.72	4.61
SME presence	0.06	0.09	0.07	0.10	2.45	1.90	-1.58	3.65	0.02	0.02	-1.62	2.29
Institutional Interactions												
Isomorphic pressure x SME	0.05	0.11	0.03	0.12	-5.11	3.78	2.32	2.58	-0.01	0.03	1.61	2.76
Marketing acceptance x SME	-0.12 *	0.07	-0.13 *	0.07	-3.46 *	2.04	-1.61	2.11	-0.09	0.03	0.04	0.83
Board membership x SME	-0.24	0.17	-0.27	0.21	-0.49	1.78	-6.09	7.43	-0.01	0.03	0.83	2.56
Structural Interactions												
Industry turbulence x SME	-0.06	0.06	-0.08	0.07	-0.81	1.25	-4.23	3.18	-0.01	0.02	-0.70	0.90
Internationalization x SME	0.01	0.02	0.02	0.02	2.29	1.85	-0.73	1.10	-0.01	0.01	0.49	0.47
Strategic Interactions												
Corporate brand x SME	0.06	0.07	0.03	0.07	1.15	2.31	-3.00	4.81	0.01	0.01	-0.30	1.59
House of brands x SME	-0.11	0.14	-0.07	0.14	5.65	3.55	-0.60	2.64	-0.01	0.03	-0.41	0.82
Business to customer x SME	-0.03	0.18	0.04	0.18	0.37	3.50	2.66	6.76	-0.01	0.04	0.23	1.28
Business to business x SME	-0.02	0.07	0.00	0.07	2.02	2.51	4.95	4.63	0.00	0.01	1.51	1.82
Service product x SME	-0.04	0.08	-0.03	0.09	-0.99	1.76	4.52	3.27	-0.03 **	0.02	-0.06	1.41
Year ^c	0.03	0.03	-0.06 **	0.03	3.61 *	2.78	-0.76	1.98	0.01	0.01	0.16	0.46
Wald X^2	3612.3 ***		4703.10 ***		179.23		143.90 ***		1128.49 ***		399.65 ***	
Observations	816		786		786		811		816		815	

Table E.10: GEE random effects analysis of USA SME impact on firm performance 2008-2010

*p<.1, **p<.05, ***p<.01

Note: There were no observed instances of a SME being a board member, so the "Board membership x SME" variable was dropped from the UK analyses.

a= Tobin's q, MTB(assets), MTB(equity), Sales Growth, ROA and ROS, R&D intensity, and Prior performance are centered at the country

and two digit SIC level to control for industry segment effects.

b=Prior performance is the lagged form of the dependent variable.

c= Year is controlled for as a dummy variable.

